# Quantum Information in Geometric Quantum Mechanics

Research Doctorate (PhD) in Fundamental and Applied Physics

Università degli Studi di Napoli Federico II

Georg Friedrich Volkert Ciclo XXIII

Advisor: Professor Giuseppe Marmo

#### Abstract

A fundamental starting point in quantum information theory is the consideration of the von Neumann entropy and its generalization to relative quantum entropies. A particular feature of quantum relative entropies is their relation via their Hessian to monotonic Riemannian metrics on the dense set of invertible mixed quantum states (Lesniewski and Ruskai 1999). These metrics are also known as quantum Fisher information metrics and provide a direct link to quantum estimation theory (Helstrom 1969). Quantum Fisher information metrics which are extendable to pure states coincide all with the Fubini Study metric of the projective Hilbert space of complex rays.

This theses outlines possible advantages of an inverse approach to quantum information theory, by starting with the Fubini Study metric rather then with the von Neumann entropy. This is done in a first step by associating to the Fubini Study metric a covariant and a contra-variant structure on the punctured Hilbert space as being available in the geometric formulation of quantum mechanics. While the contra-variant structure leads to a quantum version of the Cramér-Rao inequality for general 1-dimensional submanifolds of pure states, the covariant structure provides alternative entanglement monotones by identifying an inner product on the pullback tensor fields on local unitary group orbits of quantum states. It is shown in the case of two qubits that these monotones yield a more efficient estimation of entanglement than standard measures from the literature as those associated with the linearization of the von Neumann entropy.

Per Daniela

# Preface

This work focuses on mathematical foundations of quantum information from a geometric point of view. The main motivation for this project originated essentially from two recent observations made by Marmo and collaborators on the Fubini Study metric as used in covariant form in the geometric formulation of quantum mechanics. First, this metric links to alternative quantum entanglement measures [1-6], and second, it links to the so-called *quantum* Fisher information measure [7] as used in quantum estimation problems. The two concepts of entanglement measures and Fisher information are both well known as fundamental concepts in quantum information theory. However, the connection between the two concepts in strict geometrical terms and it's resulting implications hasn't been discussed yet and clearly indicates a lack on a deeper understanding on some of the most fundamental concepts of quantum information theory. The present work aims to fill this gap by reviewing the above stated two sets of observations in detail and by discussing what happens when merging these two together. As a result, a new application on the quantum experimental bounds of weak entanglement quantification within the currently emerging research field of entanglement estimation theory [8] is presented.

At this point I would like to thank all persons particularly involved in the process of this work.

I thank Professor Giuseppe Marmo for his motivating exceptional ideas and his kind and very special teaching over the last four years. He gave me the opportunity to get a great insight into one of the most profounds mathematical foundations of quantum mechanics.

In this regard I'm very grateful to Professor Franco Ventriglia and Doctor Paolo Facchi for fundamental insights resulting from their discussions together with Professor Marmo held in January of this year in Zaragoza leading to an essential part of this thesis in section 3.2.

Moreover, I like to thank my co-advisor Doctor Paolo Aniello for many

stimulating discussions and important suggestions on the literature during all work stages of our publications. In this regard I thank Doctor Jesús Clemente-Gallardo from the University of Zaragoza, in particular, for the constructive collaboration and for motivating me to deepen my knowledge in scientific computing.

A special thank I would like to give to Professor Detlef Dürr from the University of Munich for discussions and suggestions on the reconsideration of the definition on 'maximal entanglement' in the literature.

Concerning the bureaucratic universe, I'm particularly grateful to Mr Guido Celentanto for always being very kind and helpful during all phases of the PhD program.

Following persons I would like to give my very special thanks: Angela and Aldo Ibello for being accepted as part of their family and for the extraordinary insights into the excellent Napoletean cuisine and culture. Doctor Victor Andrés Ferretti for being best friend. Last but not least, my parents Giovanna Valli and Wolfgang Volkert for their love, patience and financial support.

Georg F. Volkert

 $a |\text{Napoli}\rangle + b |\text{München}\rangle$ November 2011

# Contents

1	Qua	antum information – Getting started 7					
	1.1	Quantum information processing					
	1.2	The role of entanglement					
	1.3	Quant	tum error quantification	10			
	quantum information to geometric QM	12					
	1.5	An in	verse approach: This thesis	14			
<b>2</b>	Geometric Quantum Mechanics						
	2.1	Covar	iant structures	18			
		2.1.1	From Hermitian operators to real-valued functions	18			
		2.1.2	The Fubini-Study metric seen from the Hilbert space $% f(x)=0$ .	20			
		2.1.3	The pullback on general submanifolds	22			
		2.1.4	An example	24			
		2.1.5	The pull-back on homogenous spaces	28			
		2.1.6	Example: The pullback in a Weyl-system	30			
	2.2	Contr	avariant structures	33			
		2.2.1	From Hilbert spaces to Hilbert manifolds	33			
		2.2.2	Projectable tensor fields	37			
		2.2.3	Non-commutative $C^{\ast}\mbox{-algebras}$ of Kähler functions	37			
3	Info	ormati	on Inequalities	39			
	3.1	Quant	tum Entropy Inequalities	39			
		3.1.1	Operator convex functions	40			
		3.1.2	Entropy in Geometric QM	42			
		3.1.3	Wehrl-Inequality	43			
		3.1.4	Rény-Wehrl-Inequalities	45			
	3.2	Geom	etric Inequalities	46			
		3.2.1	Inner products on tensor spaces	46			
		3.2.2	Tensor field contractions	48			

		3.2.3	The Robertson-Schrödinger inequality	51	
		3.2.4	Quantum Cramér Rao Inequality	53	
4	Ent	anglen	nent Monotones	56	
	4.1	Separa	ability and Lagrangian entanglement	57	
		4.1.1	Segre embeddings seen from the Hilbert space	59	
		4.1.2	Symplectic orbits seen from the Hilbert space	60	
	4.2	Inner <sup>-</sup>	products on tensor fields and intermediate entanglement	61	
		4.2.1	Inner products on higher order tensor fields	63	
	4.3	Entan	glement monotones on two qubits	66	
	4.4	Towards a generalized algorithm		69	
	4.5	Invariant operator valued tensor fields (IOVTs) and mixed			
		states entanglement			
		4.5.1	The basic construction	73	
		4.5.2	Purity, concurrence and covariance measures	75	
<b>5</b>	Ent	anglen	nent Estimation	77	
5	<b>Ent</b> 5.1	<b>anglen</b> Quant	ment Estimation	<b>77</b> 80	
5	<b>Ent</b> 5.1	anglen Quant 5.1.1	nent Estimation         um state estimation         Schmidt coefficient estimation	<b>77</b> 80 81	
5	<b>Ent</b> 5.1	<b>anglen</b> Quant 5.1.1 5.1.2	nent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation	77 80 81 82	
5	<b>Ent</b> 5.1	anglen Quant 5.1.1 5.1.2 5.1.3	nent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation	77 80 81 82 83	
5	<b>Ent</b> 5.1 5.2	anglen Quant 5.1.1 5.1.2 5.1.3 Estima	nent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         ation of inner products on tensor fields	77 80 81 82 83 85	
5	<b>Ent</b> 5.1 5.2	anglen Quant 5.1.1 5.1.2 5.1.3 Estima 5.2.1	hent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         ation of inner products on tensor fields         Discussion for monotones from the symmetric part	<ul> <li>77</li> <li>80</li> <li>81</li> <li>82</li> <li>83</li> <li>85</li> <li>87</li> </ul>	
5	<b>Ent</b> 5.1 5.2	anglen Quant 5.1.1 5.1.2 5.1.3 Estima 5.2.1 5.2.2	hent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         ation of inner products on tensor fields         Discussion for monotones from the symmetric part         Discussion for monotones from the anti-symmetric part	77 80 81 82 83 85 85 87 89	
5	Ent 5.1 5.2 5.3	anglen Quant 5.1.1 5.1.2 5.1.3 Estima 5.2.1 5.2.2 Conclu	hent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         ation of inner products on tensor fields         Discussion for monotones from the symmetric part         Discussion for monotones from the anti-symmetric part	77 80 81 82 83 85 85 87 89 90	
5 A	Ent 5.1 5.2 5.3 Wey	anglen Quant 5.1.1 5.1.2 5.1.3 Estima 5.2.1 5.2.2 Conclu	hent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         Purity estimation         ation of inner products on tensor fields         Discussion for monotones from the symmetric part         Discussion for monotones from the anti-symmetric part         usions and outlook	<ul> <li>77</li> <li>80</li> <li>81</li> <li>82</li> <li>83</li> <li>85</li> <li>87</li> <li>89</li> <li>90</li> <li>95</li> </ul>	
5 A	Ent 5.1 5.2 5.3 Wey A.1	anglen Quant 5.1.1 5.1.2 5.1.3 Estima 5.2.1 5.2.2 Conclu yl Syst Introd	hent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         Purity estimation         ation of inner products on tensor fields         Discussion for monotones from the symmetric part         usions and outlook         ems         ucing canonical coordinates	<ul> <li>77</li> <li>80</li> <li>81</li> <li>82</li> <li>83</li> <li>85</li> <li>87</li> <li>89</li> <li>90</li> <li>95</li> <li>96</li> </ul>	
5 A	Ent 5.1 5.2 5.3 Wey A.1	anglen Quant 5.1.1 5.1.2 5.1.3 Estima 5.2.1 5.2.2 Conclu yl Syst Introd A.1.1	hent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         Purity estimation         ation of inner products on tensor fields         Discussion for monotones from the symmetric part         Discussion for monotones from the anti-symmetric part         usions and outlook         ems         ucing canonical coordinates         Linear entropy estimation	<ul> <li>77</li> <li>80</li> <li>81</li> <li>82</li> <li>83</li> <li>85</li> <li>87</li> <li>89</li> <li>90</li> <li>95</li> <li>96</li> <li>96</li> </ul>	
5 A	Ent 5.1 5.2 5.3 Wey A.1	anglen Quant 5.1.1 5.1.2 5.1.3 Estima 5.2.1 5.2.2 Conclu <b>vl Syst</b> Introd A.1.1 A.1.2	hent Estimation         um state estimation         Schmidt coefficient estimation         Linear entropy estimation         Purity estimation         Purity estimation         ation of inner products on tensor fields         Discussion for monotones from the symmetric part         Discussion for monotones from the anti-symmetric part         usions and outlook         ems         ucing canonical coordinates         Complex coordinates	<ul> <li>77</li> <li>80</li> <li>81</li> <li>82</li> <li>83</li> <li>85</li> <li>87</li> <li>89</li> <li>90</li> <li>95</li> <li>96</li> <li>96</li> <li>98</li> </ul>	

# 4

	A.3	The Wigner-Weyl correspondence		
в	Ten	sor fie	lds from geometrized C*-algebras	103
	B.1	Contra	a-variant tensors on the dual vector space $\mathcal{A}^*$	105
	B.2	Covariant tensors on the vector space $\mathcal{A}$		
	B.3	Kählerian manifolds from states		
		B.3.1	Pure states	114
		B.3.2	Mixed states	114
		B.3.3	Maximal rank states	115
	B.4	Covari	ant tensors on $u^*(n)$ - Construction in the Bloch repre-	
		sentation		
	B.5	Tensor	rs from $\mathcal{G}$ -orbits - A relation to IOVTs $\ldots \ldots \ldots$	119
		B.5.1	Vector representation induced orbits	120
		B.5.2	Adjoint representation induced orbits	124
		B.5.3	A relation to IOVTs on $\mathcal{G}$ : The case $\omega = \mathbb{1} \dots \dots$	127
С	Ten	sors fr	om generalized momentum maps	133

# **1** Quantum information – Getting started

In this first chapter we will give an introduction to the topic of the underlying work. For this purpose we will start with a brief overview on the basic ideas behind the concept of quantum information and its future emerging technological applications. The engineering problems arising here will then directly lead to the central motivation and content of this work.

#### **1.1** Quantum information processing

In 1982, Feynman made following observation [9]. A computer simulation of the time evolution of a given quantum system requires a computational time *exponentially growing with the dimension* of the quantum system. Feynman concluded that, vice versa, given a computer based on the laws of quantum mechanics, one should in principle solve computational problems (including the problem of simulating the time evolution of a big quantum system) in a notable faster time.

The mathematical ingredient being responsible for the speed up may be seen physically based on the particular way how a quantum system with a certain number of degrees of freedom is decomposed into its subsystems as follows. While in the classical case one uses a Cartesian product

$$\mathbb{R}^{2k}\times\mathbb{R}^{2k}..\times\mathbb{R}^{2k}=\mathbb{R}^{2kN}$$

to identify a decomposition into subsystems, one has to take into account for the quantum case a tensor product

$$\mathbb{C}^k \otimes \mathbb{C}^k .. \otimes \mathbb{C}^k \cong \mathbb{R}^{(2k)^N}$$

Unitary operations defining a quantum dynamics may therefore act on a complex Hilbert space of dimension  $k^N$ .

The smallest quantum information unit is given by a so-called *qubit* being

represented by a normalized vector in a 2-level complex Hilbert space isomorphic to  $\mathbb{C}^2$  [10]. An initial state vector representing a circuit with 32 qubits for instance, allows therefore to consider unitary transformations inducing a *parallel quantum information processing* on a complex Hilbert space of more than four billion (!) dimensions.

Actually, the idea of quantum computation became of general interest a decade latter after Feynman's observation due to the work of Shor, proposing a quantum algorithm implying an exponential speed up for factorizing integers into prime numbers [11, 12].

Of course, solving 'classical' problems like prime number factorization on a quantum computer requires the extraction of classical information (that is a sequence of classical bits) from the final quantum state vector as output result of the corresponding quantum algorithm. This extraction is achieved by a measurement inducing 'a collapse' defined by a (non unitary) projection of the final quantum state vector

$$|\psi_{final}\rangle = U |\psi_{initial}\rangle = \sum_{j \in \{0,1\}^N}^{2^N} a_j |j\rangle$$
(1.1)

to a 1-dimensional subspace spanned by an eigenvector

$$|1011001...\rangle := |1\rangle \otimes |0\rangle \otimes |1\rangle \otimes |1\rangle \otimes |0\rangle \otimes |0\rangle \otimes |1\rangle \otimes ..$$

associated to an Hermitian operator on the composite Hilbert space

$$\mathcal{H}_{sys} \cong (\mathbb{C}^2)^{\otimes N}$$

representing the quantum register of N qubits.

#### 1.2 The role of entanglement

The final quantum state vector in the unitary process (1.1) will be in general *entagled* before the measurement occurs. Actually, any unitary quantum information process like (1.1) may be seen decomposed into a sequence of elementary unitary transformations provided by logical quantum gates inducing entanglement on each quantum state vector containing the information of the intermediate result related to the computational task. This may be illustrated in the simplest case of a CNOT (or 'controlled NOT') gate represented by a matrix

$$U_{CNOT} \equiv \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 1 \\ 0 & 0 & 1 & 0 \end{pmatrix}$$
(1.2)

acting on a quantum register consisting of two qubits according to

$$U_{CNOT} |x\rangle \otimes |y\rangle = |x\rangle \otimes |x \oplus y\rangle \tag{1.3}$$

with  $x \oplus y := x + y \mod 2$  (see e.g. [10]). Such an operation is *non-local* as the second qubit state changes in dependence of the first qubit state. The resulting state may therefore become entangled if the first qubit is in a superposition state, say

$$|x\rangle \equiv \sqrt{\lambda} |0\rangle + \sqrt{1-\lambda} |1\rangle, \quad \lambda \in [0,1].$$

As a matter of fact, for  $|y\rangle \equiv |0\rangle$  one finds

$$\begin{aligned} |x\rangle \otimes |x \oplus y\rangle &= \sqrt{\lambda} |0\rangle \otimes |0 \oplus 0\rangle + \sqrt{1 - \lambda} |1\rangle \otimes |0 \oplus 1\rangle \\ &= \sqrt{\lambda} |0\rangle \otimes |0\rangle + \sqrt{1 - \lambda} |1\rangle \otimes |1\rangle \\ &= \sqrt{\lambda} |00\rangle + \sqrt{1 - \lambda} |11\rangle \end{aligned}$$
(1.4)

which recovers a *Bell state* [13], that is, a maximal entangled state for  $\lambda \equiv 1/2$ . Indeed, any quantum algorithm may be seen decomposed into a finite sequence of universal quantum gates including the CNOT gate (1.3) generating entangled states as intermediate results in the corresponding quantum computational steps. Entanglement provides therefore a fundamental physical resource for realizing quantum computers.

#### **1.3** Quantum error quantification

Unfortunately, there is a serious obstruction on the road to the concrete technological realizations of such quantum information processing devices. The main quantum engineering challenge arises here when taking precisely into account the interaction with the macroscopic environment. The latter implies in particular for quantum systems the destruction of the entanglement required for an error-free quantum information processing on the microscopic scale. Actually, this kind of destruction is a direct consequence of *entanglement with the environment* on the macroscopic scale (also known as 'decoherence'). An opposition between quantum information devices with optimal functionality and decoherence may therefore be captured by the notion of entanglement on different scales according to the following scheme.

Microscopic scale entanglement as resource for quantum error corrections [14]

vs.

Macroscopic scale entanglement as origin of errors on the microscopic scale.

This scheme makes clear that any serious quantum engineering approach to the realization and testing of quantum information devices requires a quantitative understanding on the relation between error tolerances and entanglement. Thus the elementary question which arises is how to quantify, both theoretically and experimentally, the error occurring in open quantum systems.

First of all, when taking into account an interaction with the environment, the unitary quantum information process becomes crucially modified to a non-unitary process, mathematically precisely defined by the notion of a (trace preserving) *positive map*  $\Phi$  [15,16]. The latter encodes the unitarily evolved dynamics U defining an interaction with the macroscopic environment down to the microscopic scale illustrated by the following diagram

$$\begin{array}{cccc} \mathcal{H}_{sys} \otimes \mathcal{H}_{env} & \stackrel{U}{\longrightarrow} & \mathcal{H}_{sys} \otimes \mathcal{H}_{env} \\ \text{``partial trace''} & & & & & & \\ D(\mathcal{H}_{sys}) & \stackrel{\Phi}{\longrightarrow} & D(\mathcal{H}_{sys}), \end{array}$$

where  $\mathcal{H}_{(.)}$  denote Hilbert spaces associated with a subsystem ('sys' for the the microscopic system representing for instance a quantum computer register, and 'env' for the environment) and  $D(\mathcal{H}_{sys})$  denotes the corresponding partial traced density state description of the microscopic subsystem.

To realize in this setting a quantum computer according to DiVincenzo [17] it becomes necessary to impose the constraint for any quantum state vector  $|\psi\rangle \in \mathcal{H}_{sys}$  to differ only by a factor within an interval  $[0, \epsilon]$  from a mixed state  $\rho_{\epsilon} \in D(\mathcal{H}_{sys})$  after an elementary time unit according to

$$\langle \psi | \rho | \psi \rangle \ge 1 - \epsilon.$$
 (1.5)

That is, a non-unitary process

$$\left|\psi\right\rangle\left\langle\psi\right|\mapsto\rho_{\epsilon}\equiv\left(1-\epsilon\right)\left|\psi\right\rangle\left\langle\psi\right|+\epsilon\left|\phi\right\rangle\left\langle\phi\right|\tag{1.6}$$

would be tolerable for performing quantum information processes with corresponding quantum error correction codes [14] if  $\epsilon$  is small enough. The identification of the exact value of  $\epsilon$  is therefore considered as one of the most crucial questions in the field of quantum computer engineering [17].

At this point we observe the following. Any measure on the space of quantum states  $S : D(\mathcal{H}) \to [0, 1]$  being monotonic under positive maps would be able to give a quantification on the error. The dependence of a given measure S on the error parameter  $\epsilon$  may be induced here from the parametrization of a family of quantum states

$$\Phi_{\epsilon}(|\psi\rangle \langle \psi|) \equiv \rho_{\epsilon}, \qquad (1.7)$$

each related to a fiducial state  $\rho_0 \equiv |\psi\rangle \langle \psi|$  by virtue of a positive map  $\Phi_{\epsilon}$  as illustrated in (1.6).

Actually, a complementary task in proving the functionality of quantum information processing devices could be given by the *quantification of entan-glement*. Indeed, the quantification of errors *and* entanglement may turn out to appear both captured by one single measure known as the *von Neunman entropy*.

#### 1.4 From quantum information to geometric QM

The traditional point of view in quantum information theory considers the von Neumann entropy

$$S_{vN}(\rho) := -\operatorname{Tr}(\rho \log \rho) \tag{1.8}$$

at the first place. It's fundamental role may be summarized by two remarkable facts: The von Neumann entropy provides

• the minimum amount of quantum information units needed for encoding a state  $\rho$  (representing a typical sequence of eigenstates related to letters of an alphabet) without loosing information after decoding [18], and • a unique entanglement measure for pure states, when applied to reduced density states [19].

At this point we may observe that the quantum relative entropy

$$S_{vN}(\rho,\sigma) \equiv S_{vN}(\rho) - \operatorname{Tr}(\rho \log \sigma), \qquad (1.9)$$

has the interpretation of a 'distance' between quantum states which contracts under positive maps [16],

$$S(\Phi(\rho), \Phi(\sigma)) \leqslant S(\rho, \sigma). \tag{1.10}$$

Actually, the quantum relative entropy does *not* provide a *metrical* distance due to its lack of symmetry under permutation

$$S_{vN}(\rho,\sigma) \neq S_{vN}(\sigma,\rho). \tag{1.11}$$

However, it possible to associate a metrical distance with the Hessian

$$-\partial_{\alpha}\partial_{\beta}S_{vN}(\rho + \alpha A, \rho + \beta B)|_{\alpha = \beta = 0} := M_{\rho}^{vN}(A, B)$$
(1.12)

defining a Riemmanian metric on the dense set of invertible mixed quantum state operators. Actually, there is a *family* of Riemmanian metrics constructed in this way [20], whenever taking into account *alternative* relative quantum entropies

$$S_h(\rho, \sigma) := \operatorname{Tr}(\rho \, h(\Delta_{\sigma, \rho})) \tag{1.13}$$

each defined in terms of an operator convex function<sup>1</sup> h, where  $\Delta_{\sigma,\rho}$  denotes an operator being decomposed into a left and right action

$$\Delta_{\sigma,\rho}(A) := L_{\sigma} R_{\rho}^{-1}(A) = \sigma A \rho^{-1}$$
(1.14)

<sup>&</sup>lt;sup>1</sup>We'll introduce this notion in more detail in section 3.1.1.

defined by the quantum states  $\sigma$  and  $\rho$  respectively.

To any given alternative quantum relative entropy  $S_h$  specified by a operator convex function h, one finds a quantum Fisher information metric

$$-\partial_{\alpha}\partial_{\beta}S_h(\rho + \alpha A, \rho + \beta B)|_{\alpha = \beta = 0} := M_{\rho}^h(A, B)$$

in accordance to the work of Petz, Gibilisco and al. (see e.g. [21] and references therein).

At this point we note the following. All quantum information metrics with an extension to pure states coincide with the *Fubini Study metric* when restricted to pure states. As being defined on the projective space of complex rays, it takes into account a fundamental structure in *the geometric formulation of quantum mechanics* [22–45].

#### 1.5 An inverse approach: This thesis

The geometric formulation of quantum mechanics [22–45] has its historical origin in the pioneering works of Strocchi (1956), Cantoni (1975), Cirelli et al (1983). This approach is on the opposite to the *geometric quantization* program [46] by taking into account dequantization at the first place, as emphasized and worked out in recent developments by Ashtekar, Schilling, Brody, Hughston, Marmo et al. (1997-2010).

There are several sound reasons for considering a dequantization program in the geometric formulation of quantum mechanics. The initial motivation comes automatically by accepting physical states to be fundamentally realized as elements in the projective space of complex rays rather than in a ordinary Hilbert space. This space has a fundamental geometric structure provided by the Fubini-Study metric which decomposes along a realification into a real Riemannian and an imaginary symplectic structure. Such a geometric setting makes therefore available geometric methods as used in general relativity and classical mechanics, suggesting a powerful framework for approaching both the conceptual and mathematical foundations of quantum mechanics.

Actually, the specific relation of the Fubiny-Study metric to the quantum Fisher information, as outlined in the previous section, strongly suggests a particular impact on the mathematical foundations of quantum information. Indeed, we may ask more concretely for both conceptual and methodological implications of an inverse approach to quantum information theory, by setting the Fubiny-Study metric rather the von Neumann entropy at the first place.

A first step in this direction has been considered recently by a quantum entanglement characterization in finite level quantum systems [1–3] in terms of the pullback of the Fubini Study metric on local unitary group orbits of quantum states. Based on subsequent works [4–6], it is the aim of the present thesis to establish three further steps implying the following.

- 1. Information inequalities including the Cramér-Rao inequality from contravariant structures associated to the Fubini Study metric.
- 2. A (re-) construction of quantum entanglement monotones and related quantum entropies from the pull-back of the Fubini Study metric.
- 3. An efficient quantum estimation of weakly entangled qubits on the basis of the Cramér-Rao inequality in point 1 provided by an estimation of the monotones in point 2.

The last point 3 tackles a current raised problem on the experimental bounds of entanglement quantification [8,47] and outlines therefore one of the possible advantages of an 'inverse approach', as proposed here, to quantum information.

The basic structure of the underlying work is illustrated according to the following diagram below. We'll start with the geometric formulation of quantum mechanics in section 2. This section will be focussed on the identification

of covariant and contravariant structures related to the Fubini Study metric as seen from the Hilbert space in the subsections 2.1 and 2.2 respectively. Thereafter, we will consider alternative entanglement monotones arising from covariant structures in section 4 and quantum information inequalities arising from contravariant structures in section 3. In the last section 5 we'll bring both aspects together by illustrating an application to entanglement estimation.



# 2 Geometric Quantum Mechanics

The standard formulation of quantum mechanics associates to any quantum system a complex Hilbert space  $\mathcal{H}$ . However, on the basis of Born's probabilistic interpretation it becomes appropriated to define the physical states not with vectors  $\psi$  in  $\mathcal{H}$  but with the equivalence classes

$$[\psi] = [c\psi], \quad c \in \mathbb{C}_0 \tag{2.1}$$

which are elements of the *projective* Hilbert space  $\mathcal{R}(\mathcal{H})$ .<sup>2</sup> The projective Hilbert space carries a natural geometric structure given by the Fubini Study metric measuring the distance between two complex rays. The physically relevant distance between two quantum states should thus be considered in terms of the Fubini Study metric rather then in terms of a Hermitian scalar product. This observation highlights one of the motivations for considering a geometric formulation of quantum mechanics as suggested in several papers [22–45]. Here we shall review some of the basic ideas following the specific argumentaion line close to the reviews done in [2,5] with particular emphasize on the following notation:

To keep formulas both visible and computable with the familiar Dirac's 'ket'-notation it will be convinient to translate the physical relevant geometric information carried by the Fubini Study metric on  $\mathcal{R}(\mathcal{H})$  on the level of the (punctured) Hilbert space  $\mathcal{H}_0 \equiv \mathcal{H} - \{0\}$ . This can be done both in terms of a covariant and a contravariant tensor field whenever  $\mathcal{H}_0$  becomes identified with a differentiable manifold.

<sup>&</sup>lt;sup>2</sup>At this point one may remark also alternative interpretations of quantum mechanics like the one by DeBroglie and Bohm, taking into account the quantum current density dependent quotient  $j^{\psi}/|\psi|^2$  at the first place [48, 49]. The latter structure is invariant under  $\mathbb{C}^*$ -transformations on  $\psi \in \mathcal{H}_0$  and provides therefore an alternative motivation for considering  $\mathcal{R}(\mathcal{H})$  instead of  $\mathcal{H}$  as the appropriated space of quantum states. Actually, a way of evading the measurement problem in such a setting may be seen based on the notion of *conditional* quantum states parametrized by the particle position configurations of the measurement device [50]. Such states may be modeled by a finite (macroscopic high) dimensional submanifold  $Q_{apparatus} \subset \mathcal{R}(\mathcal{H})$ .

One of the most appealing aspects in geometric quantum mechanics will be the translation of operator  $C^*$ -algebras to non-commutative  $C^*$ -algebras of functions. This translation makes use of a contravariant structure being related to the Fubini Study metric. In contrast, we will consider covariant structures when performing a pullback to general submanifolds in  $\mathcal{H}$ . It will therefore be convenient to distinguish covariant from contavariant structures by discussing them in separated subsections, as provided here in 2.1 and 2.2 respectively.

In what follows, our statements should be considered to be always mathematically well defined whenever the Hilbert space we intend to identify with a manifold is finite dimensional. Indeed, the basic ideas coming along the geometric approach in the finite dimensional case are fundamental for approaching the infinite dimensional case. The additional technicalities which may be required in the latter case will be discussed here by means of specific examples rather than by focusing on general claims. Readers interested in the mathematical foundations of infinite dimensional manifolds are invited to consult [51–53].

#### 2.1 Covariant structures

Before considering the Fubiny Study metric in covariant form, we shall start with covariant tensor fields of lowest order.

#### 2.1.1 From Hermitian operators to real-valued functions

Given a Hermitian operator  $A \in u^*(\mathcal{H})$  defined on a Hilbert space  $\mathcal{H}$ , we shall find a real symmetric function

$$f_A(\psi) := \langle \psi | A\psi \rangle, \quad \psi \in \mathcal{H}$$
(2.2)

on  $\mathcal{H}$ . These functions decompose into functions

$$f_{P_j}(\psi) = \langle \psi | P_j \psi \rangle, \quad \psi \in \mathcal{H}$$
 (2.3)

associated with a family of projectors  $P_j := |e_j\rangle \langle e_j|$  constructed from an orthonormal basis  $\{|e_j\rangle\}_{j\in I}$  on  $\mathcal{H}$ . The decomposition of the function induced by the spectral decomposition of the operator

$$A = \sum_{j} \lambda_{j} P_{j} \tag{2.4}$$

yields a quadratic function

$$f_A(\psi) = \sum_j \lambda_j f_{P_j}(\psi) = \sum_j \lambda_j |z^j|^2(\psi).$$
(2.5)

with the coordinate functions

$$\langle e_j | \psi \rangle := z^j(\psi) \,.$$
 (2.6)

In this regard we may recover expectation values of an operator A as values of a function

$$e_A(\psi) := \frac{f_A(\psi)}{\langle \psi | \psi \rangle} \tag{2.7}$$

on the punctured Hilbert space  $\mathcal{H}_0 := \mathcal{H} - \{0\}$ . By virtue of the map<sup>3</sup>

$$\mu : \mathcal{H}_0 \to u^*(\mathcal{H}), \quad |\psi\rangle \mapsto \rho_{\psi} := \frac{|\psi\rangle \langle \psi|}{\langle \psi |\psi\rangle}$$
(2.8)

<sup>&</sup>lt;sup>3</sup>This map provides an instance of a so called *momentum map* as being known from Hamiltonian mechanics [54]. To be specific, for an action of a Lie group  $\mathcal{G}$  on a symplectic manifold  $(M, \omega)$  one defines a momentum map as a map  $\mu$  from the symplectic manifold to the dual of the Lie algebra of  $\mathcal{G}$ , such that all  $\mathbb{R}$ -valued pairings between  $\mu(v), v \in M$ and elements a of the Lie algebra generate a 1-form being the contraction of  $\omega$  with a vector field associated with a by  $v \mapsto \frac{d}{dt} \exp(at)|_{t=0}v$ . In the geometric formulation of quantum mechanics we may specialize this situation with  $M \equiv \mathcal{H}_0$  and  $\mathcal{G} \equiv U(\mathcal{H})$  [45]. The symplectic structure  $\omega$  on  $\mathcal{H}_0$  will be introduced later in section 2.2.

we note that

$$e_A(\psi) = \rho_{\psi}(A), \quad \rho_{\psi} \in D^1(\mathcal{H})$$
(2.9)

identifies a pull-back function from the set  $D^1(\mathcal{H})$  of normalized rank-1 projectors which are in 1-to-1 correspondence with pure physical states in  $\mathcal{R}(\mathcal{H})$ . Hence,  $e_A$  is the result of the pull-back of a function form  $\mathcal{R}(\mathcal{H})$  to  $\mathcal{H}_0$ .

#### 2.1.2 The Fubini-Study metric seen from the Hilbert space

The map  $\mu$  in (2.8) relates to the commutative diagram

$$\begin{array}{ccc} \mathcal{H}_0 & \stackrel{\mu}{\longrightarrow} & u^*(\mathcal{H}) \\ \pi \\ \downarrow & & \iota \\ \mathcal{R}(\mathcal{H}) & \stackrel{\cong}{\longrightarrow} & D^1(\mathcal{H}), \end{array}$$

providing a fundamental tool for pulling back, in a both computable and – with the familiar 'ket'-notation of Dirac – visible way, any covariant structure defined on  $D^1(\mathcal{H}) \cong \mathcal{R}(\mathcal{H})$  to the 'initial' punctured Hilbert space  $\mathcal{H}_0$ .

In the following we will be particularly interested in the pullback of a covariant structure related to the Fubini Study metric on  $\mathcal{R}(\mathcal{H})$ . For this purpose, we consider for a given Hermitian operator A, the operator-valued differential dA in respect to a real parametrization<sup>4</sup> of  $u^*(\mathcal{H})$ , and define the (0, 2)-tensor field

$$\operatorname{Tr}(dA \otimes dA).$$
 (2.10)

The differential calculus on a submanifold  $\mathcal{M} \subset u^*(\mathcal{H})$ , may then be inherited from the 'ambient space'  $u^*(\mathcal{H})$  together with this covariant structure. In

<sup>&</sup>lt;sup>4</sup>Such a parametrization is available, for instance in terms of the Bloch representation  $A = \sum \lambda_j \sigma^j$  with  $\lambda_j := \text{Tr}(A\sigma^j)$ . An explicit computation of the resulting tensor field is illustrated for  $\mathcal{H} \cong \mathbb{C}^2$  in the appendix section B.4.

particular for  $\mathcal{M} \cong D^1(\mathcal{H}) \cong \mathcal{R}(\mathcal{H})$  we find<sup>5</sup>

$$\operatorname{Tr}(d\rho_{\psi} \otimes d\rho_{\psi}) = \frac{\langle d\psi \otimes d\psi \rangle}{\langle \psi | \psi \rangle} - \frac{\langle \psi | d\psi \rangle}{\langle \psi | \psi \rangle} \otimes \frac{\langle d\psi | \psi \rangle}{\langle \psi | \psi \rangle} := \kappa_{\mathcal{H}_0}, \qquad (2.11)$$

as  $\mu$ -map (2.8) induced pull-back tensor field on the associated punctured Hilbert space  $\mathcal{H}_0$  [1]. We note here that  $|d\psi\rangle$  defines a  $\mathcal{H}$ -vector-valued 1form which provides a 'classical' 1-form according to  $\langle e_j | d\psi \rangle \equiv dz^j$ . In the latter coordinates one may identify a degenerate covariant tensor field

$$\kappa_{\mathcal{H}_0} = \frac{d\bar{z}^j \otimes dz^j}{\sum_j |z^j|^2} - \frac{z^j d\bar{z}^j \otimes \bar{z}^k dz^k}{(\sum_j |z^j|^2)^2}$$
(2.12)

on the punctured Hilbert space  $\mathcal{H}_0 \cong \mathbb{C}^{n+1} - \{0\}$ . Due to the misleading convention often appearing in the literature we shall give a warning at this point: This is *not* the Fubini Study metric from the associated projective space  $\mathbb{C}P^n$ . Indeed, the above covariant tensor-field  $\kappa_{\mathcal{H}_0}$  defines a pull-back of the Fubini-Study metric tensor field from the space of complexe rays  $\mathcal{R}(\mathcal{H}) \cong$  $\mathbb{C}P^n$  to  $\mathcal{H}_0 \cong \mathbb{C}^{n+1} - \{0\}$ .

The degenerate structure  $\kappa_{\mathcal{H}_0}$  (2.11) decomposes in this regard into a real symmetric and an imaginary anti-symmetric part

$$\kappa_{\mathcal{H}_0} := \eta_{\mathcal{H}_0} + i\omega_{\mathcal{H}_0},\tag{2.13}$$

relating to a corresponding pullback of a Riemannian and a symplectic structures from the associated complex projective space to  $\mathcal{H}_0$  respectively. Of course, such a decomposition may either be induced by polar coordinates  $z^j = p^j e^{iW^j}$  or by cartesian coordinates  $z^j \equiv x^j + iy^j$ .

 $<sup>{}^{5}</sup>A$  detailed derivation of formula (2.11) can be found in the appendix C in (C.8).

#### 2.1.3 The pullback on general submanifolds

For a given embedding of a general finite dimensional manifold  ${\cal M}$  in  ${\cal H}_0$ 

$$\iota: \mathcal{M} \hookrightarrow \mathcal{H}_0, \quad \lambda \mapsto \psi(\cdot, \lambda) \tag{2.14}$$

we will find an induced pullback of the Fubini-Study metric on  $\mathcal{M}$  in terms of the pullback of the degenerate covariant structure given by  $\kappa_{\mathcal{H}_0}$  in (2.11).

Let us review an explicit derivation of the pullback when the Hilbert space under consideration is identified with a space of square integrable functions on some configuration space  $\mathbb{R}^n$  [7]. As a first step we consider here a polar coordinate decomposition

$$\psi(x,\lambda) \equiv p(x,\lambda)^{1/2} e^{iW(x,\lambda)}$$
(2.15)

and define for any given tensor field  $T(x, \lambda)$  of order r (including functions for order r = 0) the generalized expectation value integral

$$\mathbb{E}_p(T) := \int_{\mathbb{R}^n} p(x,\lambda) T(x,\lambda) dx, \qquad (2.16)$$

which 'traces out' the x-dependence of the tensor field T. Hence, for a given embedding (2.14) we shall find the pull-back structures

$$\iota^* \left\langle \psi \middle| \psi \right\rangle = \mathbb{E}_p(1) \tag{2.17}$$

$$\iota^* \langle \psi | d\psi \rangle = \mathbb{E}_p(d \ln \psi) \tag{2.18}$$

$$\iota^* \langle d\psi | \psi \rangle = \mathbb{E}_p(d \ln \psi^*) \tag{2.19}$$

$$\iota^* \langle d\psi \otimes d\psi \rangle = \mathbb{E}_p(d\ln\psi^* \otimes d\ln\psi), \qquad (2.20)$$

by using

$$\frac{d\psi}{\psi} = d\ln\psi. \tag{2.21}$$

The latter splits within the polar-decomposition (2.15) into a sum

$$d\ln\psi = d(\ln p^{1/2}e^{iW}) = d(\ln p^{1/2} + \ln(e^{iW})) = \frac{1}{2}d\ln p + idW.$$
(2.22)

By taking into account the normalization condition  $\langle \psi | \psi \rangle = 1$  one finds

$$d\mathbb{E}_p(1) = \mathbb{E}_p(dp) = \mathbb{E}_p(d\ln p) = 0$$
(2.23)

and  $\langle d\psi | \psi \rangle = - \langle \psi | d\psi \rangle$ ,

$$\mathbb{E}_p(d\ln\psi^*) = -\mathbb{E}_p(d\ln\psi). \tag{2.24}$$

From (2.16)-(2.24) we conclude the identification of a pull-back tensor field

$$\iota^* \kappa_{\mathcal{H}_0} \equiv \kappa_{\mathcal{M}} = \eta_{\mathcal{M}} + i\omega_{\mathcal{M}}, \qquad (2.25)$$

on the submanifold  $\mathcal{M}$  which is decomposed into a symmetric tensor field

$$\eta_{\mathcal{M}} := \mathbb{E}_p((d\ln p)^{\otimes 2}) + \mathbb{E}_p(dW^{\otimes 2}) - \mathbb{E}_p(dW)^2$$
(2.26)

and an antisymmetric tensor field

$$\omega_{\mathcal{M}} := \mathbb{E}_p(d\ln p \wedge dW). \tag{2.27}$$

While the latter is related to the geometric phase, we shall take into account in the symmetric part a further decomposition

$$\eta_{\mathcal{M}} \equiv \mathcal{F} + \operatorname{Cov}(dW), \qquad (2.28)$$

which identifies the classical Fisher Information metric

$$\mathcal{F} := \mathbb{E}_p((d\ln p)^{\otimes 2}) \tag{2.29}$$

and a phase-covariance matrix tensor field

$$\operatorname{Cov}(dW) := \mathbb{E}_p(dW^{\otimes 2}) - \mathbb{E}_p(dW)^2.$$
(2.30)

For the parts of the pull-back tensor field containing the phase W in differential form, we may therefore identify for pure states according to [7] the non-classical counterpart of the Fisher classical within the quantum information metric. As a matter of fact, the quantum information metric collapses to the *classical Fisher information metric* for

$$dW = 0. (2.31)$$

#### 2.1.4 An example

As a possible application we may show in the following the existence of a non-trivial Ricci curvature of a submanifold of quantum states. Consider for this purpose a family of two-modes coherent state vectors

$$|\psi_{q,p}\rangle := |q,p\rangle \in L^2(\mathbb{R}) \otimes L^2(\mathbb{R}) \cong L^2(\mathbb{R}^2)$$
(2.32)

parametrized by four-dimensional phase space vectors

$$(q,p) := (q_1, q_2, p_1, p_2) \in T^*(\mathbb{R} \times \mathbb{R}),$$
 (2.33)

which induce a parametrization of position representation wave functions

$$\langle x_1, x_2 | q, p \rangle = \psi_{q,p}(x_1, x_2) := \frac{1}{\sqrt{2\pi\sigma^2}} e^{\frac{-(q_1+x_1)^2 - (q_2+x_2)^2}{2\sigma^2}} e^{ip_1x_1 + ip_1x_2}$$
(2.34)

with an overall width  $\sigma \in \mathbb{R}_+$ . At this point we may consider the latter as an additional variable parameter, and describe a 5-dimensional submanifold of quantum state vectors

$$\psi_{q,p,\sigma}(x_1, x_2) := \psi(x, \lambda) \equiv \frac{1}{\sqrt{2\pi\lambda_5^2}} e^{\frac{-(\lambda_1 + x_1)^2 - (\lambda_2 + x_2)^2}{2\lambda_5}} e^{i\lambda_3 x_1 + i\lambda_4 x_2}$$
(2.35)

parametrized by

$$(q_1, q_2, p_1, p_2, \sigma) \equiv (\lambda_1, \lambda_2, \lambda_3, \lambda_4, \lambda_5) \equiv \lambda \in \mathbb{R}^4 \times \mathbb{R}_+.$$
(2.36)

Using the polar decomposition  $\psi(x,\lambda) \equiv p(x,\lambda)^{1/2} e^{W(x,\lambda)}$  as described in (2.15) we identify the modulo of the wave function

$$p(x,\lambda) = \frac{1}{2\pi\lambda_5^2} e^{-\frac{(x_1+\lambda_1)^2 + (x_2+\lambda_2)^2}{\lambda_5^2}}$$
(2.37)

and the phase

$$W(x,\lambda) = x_1\lambda_3 + x_2\lambda_4. \tag{2.38}$$

Using the formulas (2.26)-(2.30) we find a degenerate symmetric tensor coefficient matrix

being related to the classical Fischer information metric on a 3-dimensional submanifold, and a phase-covariance matrix tensor

The symmetric part of the pull-back  $\kappa_{\mathcal{M}}$  of (2.11) implies therefore the sum of the two tensor fields, which yield the 'quantum' Riemannian structure

$$\eta_{\mathcal{M}} = \begin{pmatrix} \frac{1}{\lambda_5^2} & 0 & 0 & 0 \\ 0 & \frac{1}{\lambda_5^2} & 0 & 0 & 0 \\ 0 & 0 & \frac{1}{4} \left( 2\lambda_1^2 + \lambda_5^2 \right) - \frac{\lambda_1^2}{4} & \frac{\lambda_1\lambda_2}{4} & 0 \\ 0 & 0 & \frac{\lambda_1\lambda_2}{4} & \frac{1}{4} \left( 2\lambda_2^2 + \lambda_5^2 \right) - \frac{\lambda_2^2}{4} & 0 \\ 0 & 0 & 0 & 0 & \frac{2}{\lambda_5^2} \end{pmatrix}.$$
(2.41)

Based on this Riemannian metric defined on a 5-dimensional submanifold of coherent states, we may compute the Ricci scalar curvature

$$R = (\eta_{jk})^{-1} R_{jk} = -\frac{5\lambda_1^4 + 5(2\lambda_2^2 + 3\lambda_5^2)\lambda_1^2 + 5\lambda_2^4 + 12\lambda_5^4 + 15\lambda_2^2\lambda_5^2}{2(\lambda_1^2 + \lambda_2^2 + \lambda_5^2)^2} \quad (2.42)$$

from the result of the associated Riemannian curvature tensor  $R_{ijl}^k$  as illustrated on the next page.

$-\frac{\lambda_1 \left(\lambda_1^2 + \lambda_2^2\right)}{\lambda_5 \left(\lambda_1^2 + \lambda_2^2 + \lambda_5^2\right)^2}$	$-\frac{\lambda_5(\lambda_1^2+\lambda_2^2+\lambda_5^2)^2}{(\lambda_1^2+\lambda_2^2+\lambda_5^2)^2}$	0 0	$-\frac{\left(\lambda_{1}^{2}+\lambda_{2}^{2}+2\lambda_{5}^{2}\right)\left(2\lambda_{5}^{2}+3\left(\lambda_{1}^{2}+\lambda_{2}^{2}\right)\right)}{\lambda_{5}^{2}\left(\lambda_{1}^{2}+\lambda_{2}^{2}+\lambda_{5}^{2}\right)^{2}},$
0 0	$0\\\lambda_1\lambda_2\left(\lambda_1^2+\lambda_2^2\right)$	$8 \left(\lambda_{1}^{2} + \lambda_{2}^{2} + \lambda_{5}^{2}\right) \\ \frac{\lambda_{2}^{4} + \lambda_{1}^{2}\lambda_{2}^{2} - 2\lambda_{5}^{4}}{\lambda_{5}^{4}}$	$8(\lambda_1^{\pm}+\lambda_2^{\pm}+\lambda_5^{\pm})$ 0
0 0	$\frac{\lambda_1^4 + \lambda_2^2 \lambda_1^2 - 2\lambda_5^4}{\lambda_1^4 + \lambda_2^2 \lambda_1^2 - 2\lambda_5^4}$	$\frac{8\left(\lambda_1^2 + \lambda_2^2 + \lambda_5^2\right)}{\lambda_1 \lambda_2 \left(\lambda_1^2 + \lambda_2^2\right)}$	$8(\lambda_1^2 + \lambda_2^2 + \lambda_5^2)$
$\frac{\lambda_1\lambda_2\left(\lambda_1^2+\lambda_2^2+3\lambda_5^2\right)}{2\lambda_5^2\left(\lambda_1^2+\lambda_2^2+\lambda_5^2\right)^2}\\2\lambda_4^3\left(\lambda_1^2+\lambda_2^2+\lambda_5^2\right)+\lambda_4^2+2\lambda_5^4+\left(4\lambda_1^2+\lambda_2^2\right)\lambda_5^2$	$-\frac{2\lambda_5^2(\lambda_1^2+\lambda_2^2+\lambda_5^2)^2}{0}$	0 0	$-\frac{\lambda_2 \left(\lambda_1^2+\lambda_2^2\right)}{\lambda_5 \left(\lambda_1^2+\lambda_2^2+\lambda_5^2\right)^2}$
$ \begin{pmatrix} -\frac{\lambda_1^4 + (3\lambda_2^2 + \lambda_5^2)\lambda_1^2 + 2(\lambda_2^2 + \lambda_5^2)^2}{2\lambda_5^2(\lambda_1^2 + \lambda_2^2 + \lambda_5^2)^2} \\ \lambda_1\lambda_2(\lambda_1^2 + \lambda_2^2 + 3\lambda_5^2) \end{pmatrix} $	$\frac{2\lambda_5^2(\lambda_1^2+\lambda_5^2+\lambda_5^2)^2}{0}$	0 0	$\sqrt{-\frac{\lambda_1\left(\lambda_1^2+\lambda_2^2\right)}{\lambda_5\left(\lambda_1^2+\lambda_2^2+\lambda_5^2\right)^2}}$
	$(B) - (B^k_{-}) - (B^k_{-})$	(11) - (11) - (11) - (11)	

#### 2.1.5 The pull-back on homogenous spaces

Hilbert spaces being relevant in the description of a quantum mechanical system may be decomposed in terms of irreducible unitary representations of a Lie Group  $\mathcal{G}$ . This implies that any covariant structure defined on a Hilbert space will admit a pull-back to a homogenous space manifold of quantum states in dependence of the chosen Lie group action. In particular for the Fubini Study metric related degenerate covariant structure  $\kappa_{\mathcal{H}_0}$  in (2.11) one finds the following statement [1,3,55].

**Theorem 2.1.** Let  $\{\theta_j\}_{j\in J}$  be a basis of left-invariant 1-forms on a Lie group  $\mathcal{G}$ , and let  $\{X_j\}_{j\in J}$  be a dual basis of left-invariant vector fields, and let iR be the Lie algebra representation associated to the unitary representation  $U: \mathcal{G} \to U(\mathcal{H})$ , inducing by means of any fiducial state vector  $|\psi\rangle \in S(\mathcal{H})$  a map

$$\iota_{\mathcal{G}} : \mathcal{G} \to \mathcal{H},$$
$$\iota_{\mathcal{G}}(g) := U(g) |\psi\rangle.$$

Then

$$\iota_{\mathcal{G}}^* \kappa_{\mathcal{H}_0} = Cov_{\rho_{\psi}}((R(X_j)R(X_k))\theta^j \otimes \theta^k := \kappa_{\mathcal{G}}^{\rho_{\psi}}$$
(2.43)

for  $\rho_{\psi} := |\psi\rangle \langle \psi| \in D^1(\mathcal{H}).$ 

In conclusion, the pull-back of  $\kappa_{\mathcal{H}_0}$  on a Lie group endowed with a unitary Hilbert space representation gives rise to a *covariance tensor field* 

$$(\rho_{\psi}(R(X_j)R(X_k)) - \rho_{\psi}(R(X_j))\rho_{\psi}(R(X_k))\theta^j \otimes \theta^k.$$
(2.44)

which reduces in particular for 1-dimensional representations to a *variance* tensor field

$$Var_{\rho_{\psi}}(R(X))\theta \otimes \theta.$$
 (2.45)

We may identify this (degenerate) pull-back tensor field construction with the pull-back of a non-degenerate pull-back tensor field which lives on a homogenous space  $\mathcal{G}/\mathcal{G}_0$ . The latter admits a smooth embedding via the unitary action of the Lie Group as orbit manifold  $\Gamma$  in the Hilbert space and establishes therefore a pull-back of the Hermitian structure both on the orbit  $\Gamma$  and the homogenous space  $\mathcal{G}/\mathcal{G}_0$ . Hence, the computation of the pull-back on the orbit, reduces to the the computation of the pull-back on the Lie group, as indicated here in the commutative diagram below.

$$\begin{array}{ccc} \mathcal{G} & \stackrel{\iota_{\mathcal{G}}}{\longrightarrow} & \mathcal{H} \\ \pi & & & \iota \\ \mathcal{G}/\mathcal{G}_0 & \stackrel{\cong}{\longrightarrow} & \Gamma. \end{array}$$

The embedding of the Lie group and its corresponding orbit is related to the co-adjoint action map on all group elements modulo U(1)-representations  $U(h) = e^{i\phi(h)}$ 

$$\iota_{\mathcal{G}}^{U(1)}: \mathcal{G}/U(1) \to \mathcal{R}(\mathcal{H}), \quad g \mapsto U(g)\rho U(g)^{\dagger}, \quad \rho \in \mathcal{R}(\mathcal{H}).$$
(2.46)

Let us underline again: The structure (2.44) is defined on the Lie group via a pull-back tensor field from the Hilbert space even though it contains the complete information of the (non-degenerate) tensor field on the corresponding co-adjoint orbit  $\Gamma$  which is embedded in the projective Hilbert space. The additional U(1)- degeneracy is here captured in a corresponding enlarged isotropy group  $\mathcal{G}_0^{U(1)}$  according the commutative diagram below.

$$\begin{array}{ccc} \mathcal{G} & \stackrel{\iota_{\mathcal{G}}}{\longrightarrow} & S(\mathcal{H}) \\ & & & \\ U(1) \downarrow & & & \\ \mathcal{G}/U(1) & \stackrel{\iota_{\mathcal{G}}^{U(1)}}{\longrightarrow} & \mathcal{R}(\mathcal{H}) \\ & & & \\ \pi \downarrow & & & \iota \uparrow \\ & & & \\ \mathcal{G}/\mathcal{G}_{0}^{U(1)} & \stackrel{\cong}{\longrightarrow} & \Gamma \end{array}$$

This approach provides therefore in an 'algorithmic' procedure to find a geometric description of coherent state manifolds, as defined in [56–58]. Indeed, the associated orbits in our approach turn out to be more general as those give by coherent states, whenever we allow to take into account also reducible representations, as it typically occurs in composite Hilbert spaces. We'll illustrate this later on in section 4.1.

#### 2.1.6 Example: The pullback in a Weyl-system

Let  $\{X_j\}_{j\in J}$  be a basis on  $V \cong \mathbb{R}^{2n}$  represented within a Weyl system according to the definition in section A.1 by a set of Hermitian operators  $\{R(X_j)\}_{j\in J}$ on  $\mathcal{H} \cong L^2(U)$ , where  $U \cong \mathbb{R}^n$  defines a Lagrangian subspace of V. Given a pure state  $\rho_{\psi} = |\psi\rangle \langle \psi|$  associated to a normalized vector  $\psi \in S(\mathcal{H})$ , we identify

$$Cov_{\rho_{\psi}}((R(X_j)R(X_k))dv^j \otimes dv^k$$
(2.47)

as the pull-back tensor field of the covariant tensor field  $\kappa_{\mathcal{H}_0}$  as defined in (2.11) from  $\mathcal{H}_0$  to V. This pullback is induced by a map

$$V \to \mathcal{H}, \quad v \mapsto W(v) |\psi\rangle$$
 (2.48)

specified by the Weyl-system map  $W: V \to U(\mathcal{H})$  with  $W(v) = e^{iR(v)}$  (see appendix A) and the choice of a fiducial state vector  $\psi \in S(\mathcal{H})$ . This follows from a generalization of theorem 2.1 to infinite dimensional Hilbert spaces, once we impose the additional condition that the fiducial vector  $\psi$  is smooth and in the domain of the Hermitian operators  $R(X_j)$  [55]. While the anti-
symmetric coefficients recover the symplectic structure<sup>6</sup>

$$\rho_{\psi}([R(X_j)R(X_k)]) = \frac{i}{4a^2}\omega_{[jk]}, \qquad (2.49)$$

due to

$$[R(v)R(v')] = \frac{i}{2a^2}\omega(v,v') \text{ and } \operatorname{Tr}(\rho_{\psi}) = 1, \qquad (2.50)$$

independently from the state  $\rho_{\psi}$ , its symmetric tensor coefficients are identical to the coefficients of a symmetric covariance matrix  $\sigma$  in dependence of the state:

$$\rho_{\psi}([R(X_j)R(X_k)]_+) - \rho_{\psi}(R(X_j))\rho_{\psi}(R(X_k)) := \sigma_{jk}(\rho_{\psi}).$$
(2.51)

In conclusion, the Weyl-system-induced pull-back of  $\kappa_{\mathcal{H}_0}$  in (2.11) establishes by its tensor coefficient matrix a  $n \times n$  covariance matrix

$$\{\operatorname{Cov}_{\rho_{\psi}}(R(X_j), R(X_k)\}_{j,k}, \tag{2.52}$$

being decomposed into a quantum state-dependent real symmetric part and a quantum state-independent imaginary anti-symmetric part

$$\sigma(\rho_{\psi}) + \frac{i}{4a^2}\omega. \tag{2.53}$$

The anti-symmetric part  $\omega$  stays invariant by definition under symplectic transformations. In contrast, we shall encounter in the symmetric part  $\sigma(\rho_{\psi})$ a non-trivial transformation. In particular, by the virtue of *Williamson's the*orem [60], there exists to any real symmetric matrix  $\sigma \in M_{2n}(\mathbb{R})$  a symplectic

 $<sup>^6\</sup>mathrm{We}$  adopt the convention as used in [59] for the commutation relations involving

 $a := \begin{cases} 2^{-1/2} \text{ for canonical position and momentum} \\ 1 \text{ for optical position and momentum.} \end{cases}$ 

transformation matrix  $S\in Sp(2n,\mathbb{R})$  such that

$$S\sigma S^T = \bigoplus_{k=1}^n d_k \mathbb{1}_2, \quad d_k \in \operatorname{spec}(i\omega \cdot \sigma)$$
 (2.54)

where  $d_k$  denote the symplectic eigenvalues of the matrix product  $i\omega \cdot \sigma$ .

# 2.2 Contravariant structures

So far we considered covariant structures derived as pullback tensor fields of a degenerate covariant tensor field  $\kappa_{\mathcal{H}_0}$ . The latter has been related to a pullback of the Fubini Study metric from the projective Hilbert space to the punctured Hilbert space.

At this point we note, it is not possible to associate to a degeneratate covariant structure  $(\kappa_{jk})$  a corresponding *contravariant* structure  $(\kappa^{jk}) =$  $(\kappa_{jk})^{-1}$ . However, it is possible to define contravariant structures on the Hilbert space which are projectable on the space of complex rays. This is done in two steps. First by considering a Hermitian structure turning a Hilbert space to a Hilbert manifold, and second, by extending this Hermitian structure in terms of dilatation and phase generating vector fields to a projectable tensor field. A benefit of such a procedure is the possibility to translate Hilbert space operator products to star products of functions on unitary orbits of pure quantum states.

#### 2.2.1 From Hilbert spaces to Hilbert manifolds

By introducing an orthonormal basis  $\{|e_j\rangle\}_{j\in J}$  on a given Hilbert space  $\mathcal{H}$ , we may define coordinate functions by setting

$$\langle e_j | \psi \rangle = z^j(\psi), \qquad (2.55)$$

which we'll write in the following simply as  $z^j$ . Correspondently, for the dual basis  $\{\langle e_j |\}$  we find coordinate functions

$$\langle \psi | e_j \rangle = \bar{z}_j(\psi^*) \tag{2.56}$$

defined on the dual space  $\mathcal{H}^*$ . By using the inner product we can identify  $\mathcal{H}$  and  $\mathcal{H}^*$ . This provides two possibilities: The scalar product  $\langle \psi | \psi \rangle$  gives rise

to a covariant Hermitian (0, 2)-metric tensor on  $\mathcal{H}$ 

$$\langle d\psi | d\psi \rangle = \sum_{j} \langle d\psi | e_j \rangle \langle e_j | d\psi \rangle = d\bar{z}_j \otimes dz^j,$$
 (2.57)

where we have used  $d \langle e_j | \psi \rangle = \langle e_j | d\psi \rangle$ , i.e., the chosen basis is not 'varied', or to a contra-variant (2,0) tensor

$$\left\langle \frac{\partial}{\partial \psi} \left| \frac{\partial}{\partial \psi} \right\rangle = \frac{\partial}{\partial \bar{z}_j} \otimes \frac{\partial}{\partial z^j}$$
(2.58)

on  $\mathcal{H}^*$ .

*Remark:* Specifically, we assume that an orthonormal basis has been selected once and it does not depend on the base point.

By introducing real coordinates, say

$$z^{j}(\psi) = x^{j}(\psi) + iy^{j}(\psi)$$

$$(2.59)$$

one finds

$$\langle d\psi | d\psi \rangle = (dx_j \otimes dx^j + dy_j \otimes dy^j) + i(dx_j \otimes dy^j - dy_j \otimes dx^j).$$
(2.60)

Thus the Hermitian tensor decomposes into an Euclidean metric (more generally a Riemannian tensor) and a symplectic form. Similarly, on  $\mathcal{H}^*$  we may consider

$$\left\langle \frac{\partial}{\partial \psi} \left| \frac{\partial}{\partial \psi} \right\rangle = \left( \frac{\partial}{\partial x_j} \otimes \frac{\partial}{\partial x^j} + \frac{\partial}{\partial y_j} \otimes \frac{\partial}{\partial y^j} \right) + i \left( \frac{\partial}{\partial y_j} \otimes \frac{\partial}{\partial x^j} - \frac{\partial}{\partial x_j} \otimes \frac{\partial}{\partial y^j} \right).$$

This tensor field, in contravariant form, may be also considered as a bidifferential operator, i.e., we may define a binary bilinear product on real smooth functions by setting

$$((f,g)) = \left(\frac{\partial f}{\partial x_j} + i\frac{\partial f}{\partial y_j}\right) \cdot \left(\frac{\partial g}{\partial x^j} - i\frac{\partial g}{\partial y^j}\right)$$
(2.61)

which decomposes into a symmetric bracket

$$(f,g) = \frac{\partial f}{\partial x_j} \frac{\partial g}{\partial x^j} + \frac{\partial f}{\partial y_j} \frac{\partial g}{\partial y^j}$$
(2.62)

and a skew-symmetric bracket

$$\{f,g\} = \frac{\partial f}{\partial y_j} \frac{\partial g}{\partial x^j} - \frac{\partial f}{\partial x_j} \frac{\partial g}{\partial y^j}.$$
(2.63)

This last bracket defines a Poisson bracket on smooth functions defined on  $\mathcal{H}$ .

Summarizing, we can replace our original Hilbert space with an Hilbert manifold, i.e. an even dimensional real manifold on which we have tensor fields in covariant form<sup>7</sup>

$$\eta_{\mathcal{H}} = dx_j \otimes dx^j + dy_j \otimes dy^j \tag{2.64}$$

$$\omega_{\mathcal{H}} = dy_j \otimes dx^j - dx_j \otimes dy^j, \qquad (2.65)$$

or tensor fields in contravariant form

$$G_{\mathcal{H}} = \frac{\partial}{\partial x_j} \otimes \frac{\partial}{\partial x^j} + \frac{\partial}{\partial y_j} \otimes \frac{\partial}{\partial y^j}$$
(2.66)

$$\Omega_{\mathcal{H}} = \frac{\partial}{\partial y_j} \otimes \frac{\partial}{\partial x^j} - \frac{\partial}{\partial x_j} \otimes \frac{\partial}{\partial y^j}$$
(2.67)

defining covariant and contravariant Riemannian and symplectic structures respectively. The contravariant tensor fields, considered as bi-differential

<sup>&</sup>lt;sup>7</sup>We shall distinguish these tensor from the symmetric and antisymmetric part of the pullback tensor  $\kappa_{\mathcal{H}_0}$ , as being related in (2.11) to the Fubini Study metric, by leaving out the index '0' within the Hilbert space notation.

operators define a symmetric product and a skew symmetric product on real smooth functions. The skew-symmetric product actually defines a Poisson bracket. In particular, for functions

$$f_A(\psi) = \langle \psi | A\psi \rangle, \quad \psi \in \mathcal{H},$$
 (2.68)

associated with Hermitian operators A, we shall end up with the relations

$$f_{[A,B]_+} \equiv G_{\mathcal{H}}(df_A, df_B). \tag{2.69}$$

$$f_{[A,B]_{-}} \equiv \Omega_{\mathcal{H}}(df_A, df_B), \qquad (2.70)$$

which replaces symmetric and anti-symmetric operator products  $[A, B]_{\pm}$  by symmetric and anti-symmetric tensor fields respectively. Hence, via these tensor fields we may identify symmetric and Poisson brackets on the set of quadratic functions according to

$$f_{[A,B]_+} = (f_A, f_B), (2.71)$$

$$f_{[A,B]_{-}} = \{f_A, f_B\}, \qquad (2.72)$$

which synthesize to a non-commutative product

$$((f_A, f_B)) = (f_A, f_B) + i\{f_A, f_B\}$$
(2.73)

of quadratic functions. In this way we may encode the original non-commutative structure on operators in terms of 'classical', i.e. Riemannian and symplectic tensor fields according to

$$((f_A, f_B)) = f_{A \cdot B}(\psi) = (G_{\mathcal{H}} + i\Omega_{\mathcal{H}})(df_A(\psi), df_B(\psi)).$$
(2.74)

#### 2.2.2 Projectable tensor fields

To take into account the geometry of the set of physical (pure) states, we need to modify G and  $\Omega$  by a conformal factor to turn them into projectable tensor fields on  $\mathcal{R}(\mathcal{H})$ . The projection is generated at the infinitesimal level by the real and imaginary parts of the action of  $\mathbb{C}_0$  on  $\mathcal{H}_0$  given by the dilation vector field  $\Delta$  and the U(1)-phase rotation generating vector field  $\Gamma := J(\Delta)$  defined by a contraction with a complex structure (1-1)-tensor field

$$J_{\mathcal{H}} = dx^j \otimes \frac{\partial}{\partial y_j} - dy^j \otimes \frac{\partial}{\partial x^j}$$
(2.75)

respectively. In this way we shall identify

$$\widetilde{G}(\psi) = \langle \psi | \psi \rangle G - (\Delta \otimes \Delta + \Gamma \otimes \Gamma)$$
(2.76)

$$\widetilde{\Omega}(\psi) = \langle \psi | \psi \rangle \,\Omega - (\Delta \otimes \Gamma - \Gamma \otimes \Delta), \tag{2.77}$$

as projectable structures [61]. They establish a Lie-Jordan algebra structure on the space of real valued functions whose Hamiltonian vector fields are also Killing vector fields for the projection  $\tilde{G}$ . In this regard one finds: A function on  $\mathcal{R}(\mathcal{H})$  defines a quantum evolution, via the associated Hamiltonian vector field, if and only if the vector field is a derivation for the Riemann-Jordan product [39,62].

## 2.2.3 Non-commutative C\*-algebras of Kähler functions

For a given (bounded) operator  $A \in \mathcal{B}(\mathcal{H})$  on a Hilbert space we consider the complex valued function  $f_A : \mathcal{H}_0 \to \mathbb{C}$  on the punctured Hilbert space defined by

$$f_A(\psi) \equiv \frac{\langle \psi | A | \psi \rangle}{\langle \psi | \psi \rangle} \tag{2.78}$$

with  $\psi \in \mathcal{H}_0 := \mathcal{H} - \{0\}$ . By introducing a contravariant tensor field  $K_{\mathcal{H}_0}$ on  $\mathcal{H}_0$  being projectable on the space of complex rays  $\mathcal{R}(\mathcal{H})$  we shall find a decomposition of the function  $f_{AB}$  on a product of two operators A and B into

$$f_{AB}(\psi) = f_A \cdot f_B|_{\psi} + K_{\mathcal{H}_0}(df_A, df_B)|_{\psi}.$$
 (2.79)

That is, a point-wise product and a contraction of the contra-variant Hermitian tensor field  $K_{\mathcal{H}_0}$  with the differential 1-forms  $df_A$  and  $df_B$  yielding a non-commutative star-product of functions

$$f_A \cdot f_B|_{\psi} + K_{\mathcal{H}_0}(df_A, df_B)|_{\psi} := f_A \star f_B|_{\psi}.$$
(2.80)

This product defines a  $C^*$ -algebra for all functions on  $\mathcal{R}(\mathcal{H})$  provided by (2.78). These functions turn out to be Kähler function as they generate in their real and imaginary part Hamiltonian vector fields which are also Killing vector fields [40]. Note that the corresponding pullback of Kähler functions on a submanifold  $\Gamma \subset \mathcal{R}(\mathcal{H})$  yields again a  $C^*$ -algebra iff  $\Gamma$  is a unitarily generated homogenous space [61]. In particular, this includes also  $\mathcal{R}(\mathcal{H})$  as an instance of a unitary orbit in the case of finite dimensional Hilbert spaces via the bijection

$$\mathcal{R}(\mathcal{H}) \leftrightarrow U(N)/U(N-1) \times U(1), \quad N = \text{Dim}(\mathcal{H}).$$
 (2.81)

For all unitary orbits  $\Gamma \subset \mathcal{R}(\mathcal{H})$  one concludes: The geometric formulation of quantum mechanics makes available a 'dequantization' relation

$$f: \mathcal{B}(\mathcal{H}) \to \mathcal{F}_K(\Gamma)$$
 (2.82)

between a  $C^*$ -algebra of operators to  $C^*$ -algebras of Kähler functions on the unitary orbits of pure quantum states.

# **3** Information Inequalities

In the following section we will consider two classes of inequalities arising in a natural way from contravariant structures of the geometric formulation of quantum mechanics. Essentially, we shall distinguish here between *entropy inequalities* in section 3.1 and *geometric inequalities* in section 3.2. The latter will fundamentally highlight a link to the *quantum Fisher information* in terms of the *quantum Cramér-Rao inequality*. Both type of inequalities may therefore be subsumed in this section under what we could call general *information inequalities*.

# 3.1 Quantum Entropy Inequalities

In the last section 2.2.3 of the previous chapter we found a dequantization relation

$$f: \mathcal{B}(\mathcal{H}) \to \mathcal{F}_K(\Gamma)$$
 (3.1)

between operators and Kähler functions as a fundamental consequence of the geometric formulation of quantum mechanics. This has been achieved by the functions defined in (2.78) according to

$$f_A(\psi) \equiv \frac{\langle \psi | A | \psi \rangle}{\langle \psi | \psi \rangle}$$

together with the non-commutative star product (2.80)

$$f_A \cdot f_B|_{\psi} + K_{\mathcal{H}_0}(df_A, df_B)|_{\psi} := f_A \star f_B|_{\psi}$$

defined by the projectable contravariant tensor field  $K_{\mathcal{H}_0}$ . At this point we may consider following problem.

Open Problem 3.1. Given a quantum entropy, or any other quantum (in-

formation) statistical measure

$$S: D(\mathcal{H}) \to \mathbb{R}_+$$
 (3.2)

on the convex subset of mixed quantum states  $D(\mathcal{H}) \subset \mathcal{B}(\mathcal{H})$ , does there exists a functional

$$\tilde{S}: \mathcal{F}_K(\Gamma) \to \mathbb{R}_+$$
 (3.3)

on  $\mathcal{F}_K(\Gamma)$  such that the following Diagram

$$D(\mathcal{H}) \xrightarrow{S=S \circ f|_{D(\mathcal{H})}} \mathbb{R}_{+}$$

$$\iota \downarrow \qquad \qquad \uparrow \tilde{s} \qquad (3.4)$$

$$\mathcal{B}(\mathcal{H}) \xrightarrow{f} \mathcal{F}_{K}(\Gamma).$$

commutes?

For a constructive approach to this problem, we'll need as first step a short digression on *operator convex functions*.

## 3.1.1 Operator convex functions

Operator convex functions (see e.g. [16] and references therein) are instances of general *operator functions* associating to any function

$$h: \mathbb{R} \to \mathbb{R},\tag{3.5}$$

a map, denoted with an abuse of notation with the same letter,

$$h: u^*(\mathcal{H}) \to u^*(\mathcal{H}) \tag{3.6}$$

on Hermitian operators  $A \in u^*(\mathcal{H})$ , such that

$$h(A) = h(U\operatorname{Diag}\{\lambda_j\}U^{\dagger}) = Uh(\operatorname{Diag}\{h(\lambda_j)\})U^{\dagger}.$$
(3.7)

Operator *convex* functions are then defined with the additional property

$$h\left(\sum_{j} p_{j} A_{j}\right) \leq \sum_{j} p_{j} h(A_{j}), \quad p_{j} \in [0, 1], \quad \sum_{j} p_{j} = 1, \quad (3.8)$$

for all  $A_j \in u^*(\mathcal{H})$ . Note that all convex operator functions are also convex functions. Given the spectral decomposition

$$A = U \sum_{j} \lambda_{j} |e_{j}\rangle \langle e_{j}| U^{\dagger}$$
(3.9)

associated to the eigenvector basis  $\{|e_j\rangle\}_{j\in J}$  one finds therefore for all  $|\psi\rangle \in \mathcal{H}$ 

$$\langle \psi | h(A) | \psi \rangle = \langle \psi' | \sum_{j} h(\lambda_{j}) | e_{j} \rangle \langle e_{j} | \psi' \rangle$$
 (3.10)

$$= \sum_{j} |\langle \psi' | e_j \rangle|^2 h(\lambda_j)$$
 (3.11)

$$\geq h(\sum_{j} |\langle \psi' | e_j \rangle|^2 \lambda_j) \tag{3.12}$$

where we set  $|\psi'\rangle := U |\psi\rangle$ . For any given resolution of the identity

$$\sum_{i} |\psi_{i}\rangle \langle \psi_{i}| = \mathrm{Id}_{\mathcal{H}}, \qquad (3.13)$$

this implies the *Peierl's inequality* [16]

$$\operatorname{Tr}(h(A)) \ge \sum_{i} h(\langle \psi_i | A | \psi_i \rangle).$$
(3.14)

This inequality reduces to an equality if the resolution of the identity is realized by the eigenvector basis  $\{|e_j\rangle\}_{j\in J}$  of A. The Peierl's inequality (3.14) may now be used for translating quantum entropy measures into the framework of geometric quantum mechanics as we will see in the next section.

#### 3.1.2 Entropy in Geometric QM

At this point we may observe that most entropy measures arise as traces

$$S_h(\rho) := \operatorname{Tr}(h(\rho)) \tag{3.15}$$

on operator convex functions h. By virtue of Peierl's inequality (3.14) one finds

$$S_{h}(\rho) = \int_{\Gamma} \langle \psi_{g} | h(\rho) | \psi_{g} \rangle \, d\Gamma \ge \int_{\Gamma} h(\langle \psi_{g} | \rho | \psi_{g} \rangle) d\Gamma \tag{3.16}$$

for a resolution of the identity<sup>8</sup>

$$\int_{\Gamma} |\psi_g\rangle \langle \psi_g| \, d\Gamma = \mathrm{Id}_{\mathcal{H}},\tag{3.17}$$

defined by a manifold  $\Gamma \cong \mathcal{G}/\mathcal{G}_0$  of coherent states

$$|\psi_g\rangle := U(g) |\psi_0\rangle, \quad U: \mathcal{G} \to \mathcal{U}(\mathcal{H})$$
 (3.18)

associated to a unitary representation  $U(g) \in \mathcal{U}(\mathcal{H})$ . With this we arrive to an inequality

$$S_h(\rho) \ge \int_{\Gamma} h(f_{\rho}(\psi_g)) d\Gamma$$
(3.19)

linking quantum entropies of the form (3.20) to the geometric formulation of quantum mechanics. The pull-back of Kähler functions  $f_{\rho}(\psi_g)$  on manifolds of coherent states coincides here with the so-called the *Husimi functions*  $Q_{\rho}(\alpha) \equiv f_{\rho}(\psi_g)$  with  $\alpha$  denoting a parametrization of the coherent states manifold  $\Gamma$ . Actually, Kähler functions are more general then Husimi functions as they may involve Kähler functions  $f_{h(\rho)}(\psi_g)$  associated with convex operator functions h. With this generalized setting we conclude the following.

<sup>&</sup>lt;sup>8</sup>We'll hide constant factors depending on the irreducible unitary representation in the measure  $d\Gamma$ . Bloch-coherent states induced by irreducible representations of SU(2), for instance, would imply an additional factor  $(2j + 1)/4\pi$  [16].

**Proposition 3.2.** For all quantum entropy measures

$$S_h: D(\mathcal{H}) \to \mathbb{R}_+, \quad S_h(\rho) := \operatorname{Tr}(h(\rho))$$
 (3.20)

defined by a trace on a convex operator function  $h : \mathbb{R}_+ \to \mathbb{R}_+$  and for all manifolds  $\Gamma \subset \mathcal{R}(\mathcal{H})$  of coherent states  $|\psi_g\rangle := U(g) |\psi_0\rangle$  there exists two functionals

$$\tilde{S}_h(f_\rho) := \int_{\Gamma} h(f_\rho(\psi_g)) d\Gamma, \qquad (3.21)$$

$$\tilde{S}(f_{h(\rho)}) := \int_{\Gamma} f_{h(\rho)}(\psi_g) d\Gamma, \qquad (3.22)$$

on Kähler functions  $f_{\rho}, f_{h(\rho)} \in \mathcal{F}_{K}(\Gamma)$  such that the diagram in (3.4),

$$D(\mathcal{H}) \xrightarrow{S_h} \mathbb{R}_+$$

$$\iota \downarrow \qquad \qquad \uparrow \tilde{S} \ge \tilde{S}_h \qquad (3.23)$$

$$\mathcal{B}(\mathcal{H}) \xrightarrow{f} \mathcal{F}_K(\Gamma).$$

commutes exactly for (3.22) and approximately for (3.21) according to the inequality

$$S_h(\rho) \ge \tilde{S}_h(f_\rho). \tag{3.24}$$

#### 3.1.3 Wehrl-Inequality

For the operator convex function  $h(x) \equiv x \ln x$  we may directly apply Proposition 3.2 to the von Neumann entropy

$$S_{vN}(\rho) := -S_h(\rho) = -\operatorname{Tr}(\rho \ln \rho), \qquad (3.25)$$

by taking into account a sign reversion. According to Proposition 3.2 we may identify the von Neumann entropy by the functional

$$S_{vN}(\rho) = -\tilde{S}(f_{\rho\ln\rho}) = -\int_{\Gamma} f_{\rho\ln\rho}(\psi_g)d\Gamma$$
(3.26)

The corresponding approximation functional (3.21) recovers the Wehrl-Entropy

$$S_W(\rho) := -\tilde{S}_h(f_\rho) = -\int_{\Gamma} f_\rho(\psi_g) \ln f_\rho(\psi_g) d\Gamma$$
(3.27)

implying by virtue of (3.24) the Wehrl-Inequality [63]

$$S_{vN}(\rho) \le S_W(f_\rho) \tag{3.28}$$

**Remark 3.3.** According to the Lieb-conjecture one finds the minimum of the Wehrl-entropy given for pure states  $\rho_{\psi} \equiv |\psi\rangle \langle \psi|$  iff  $\psi$  is a coherent state vector [64].

Here we propose another inequality being induced by the Hermitian tensor field  $K_{\mathcal{H}_0}$  within the definition of the non-commutative product of Kähler functions in (2.80). For this purpose we consider the replacement of the von Neumann entropy by the functional  $-S_{vN}(\rho) = \tilde{S}(f_{\rho \ln \rho})$  and find

$$\tilde{S}(f_{\rho \ln \rho}) = \int_{\Gamma} f_{\rho \ln \rho}(\psi_g) d\Gamma$$
(3.29)

$$\geq \int_{\Gamma} f_{\rho \ln \rho}(\psi_g) - f_{\rho}(\psi_g) f_{\ln \rho}(\psi_g) d\Gamma \qquad (3.30)$$

$$= \int_{\Gamma} K(df_{\rho}(\psi_g), df_{\ln \rho}(\psi_g)) d\Gamma, \qquad (3.31)$$

where we used (2.79) in the last equality. In conclusion we find

$$S_{vN}(\rho) \le -\int_{\Gamma} K_{\mathcal{H}_0}(df_{\rho}(\psi_g), df_{\ln\rho}(\psi_g))d\Gamma.$$
(3.32)

This last relation outlines a possible link to the inverse problem of recovering quantum relative entropies from quantum Fisher information metrics. As a matter of fact, the latter may be seen related here for pure states to the contraction with the contravariant structure  $K_{\mathcal{H}_0}$ .

### 3.1.4 Rény-Wehrl-Inequalities

Approximations of different orders to the von Neumann entropy may arise by a class of quantum entropies known as *quantum Rény entropies*. They are defined by

$$R_q(\rho) := \frac{1}{1-q} \ln(\operatorname{Tr}(\rho^q)), \quad q \in \mathbb{R}_+$$
(3.33)

and have the property

$$\lim_{q \to 1} R_q(\rho) = S_{vN}(\rho).$$
(3.34)

The operator function  $h(x) = -x^q$  is operator convex for  $q \in [0, 1]$  [16]. Thus the corresponding translation into functionals on Kähler functions on a coherent state manifold becomes

$$R_q(\rho) = \frac{1}{1-q} \ln \left( \int_{\Gamma} f_{\rho^q}(\psi_g) d\Gamma \right)$$
(3.35)

$$\leq \frac{1}{1-q} \ln \left( \int_{\Gamma} f_{\rho}^{q}(\psi_{g}) d\Gamma \right)$$
(3.36)

$$:= R_q^W(f_\rho) \tag{3.37}$$

which recovers the *Rény-Wehrl entropies* and its associated *Rény-Wehrl in*equalities (compare also [16], p. 287). A lower bound family of inequalities may be induced here by (2.79) where we find for any fixed q and all partitions  $q = \alpha + \beta$ 

$$R_q(\rho) \ge \frac{1}{1-q} \ln\left(\int_{\Gamma} K_{\mathcal{H}_0}(df_{\rho^{\alpha}}(\psi_g), df_{\rho^{\beta}}(\psi_g))d\Gamma\right).$$
(3.38)

# 3.2 Geometric Inequalities

So far we considered inequalities derived from the Peierl's inequality (3.14) which reflected a particular property of operator convex functions. This inequality gave rise to quantum entropy inequalities being very naturally described in the geometric formulation of quantum mechanics.

In the following section we shall focus on another type of inequality being itself of geometric nature. The derivation is based on a recent discussion<sup>9</sup> made by Facchi, Marmo and Ventriglia [65].

The basic idea takes into account a short digression on some basic inequalities derived from inner products on tensor spaces. After that digression, we shall focus on inequalities derived from from inner products on tensorial structures as being particularly available in geometric quantum mechanics. At that point we shall find as a consequence the Schrödinger-Robertson inequality and a quantum version of the *Cramér Rao inequality* as being centrally used in quantum estimation problems.

#### **3.2.1** Inner products on tensor spaces

Let V be a linear space over the field of real or complex numbers endowed with an inner product  $\langle \cdot | \cdot \rangle$ , inducing a positive norm

$$\|v\|^2 = \langle v | v \rangle \ge 0 \tag{3.39}$$

for all  $v \in V$ . Now, we may consider the tensor product space  $V \otimes V$  which itself is again a linear space on which the original inner product and the associated norm may be extended according to

$$\|v_1 \otimes v_2\|^2 = \langle v_1 \otimes v_2 | v_1 \otimes v_2 \rangle \equiv \langle v_1 | v_1 \rangle \langle v_2 | v_2 \rangle$$
(3.40)

<sup>&</sup>lt;sup>9</sup>At this point I would like to gratefully acknowledge once again all participants.

inducing a corresponding inequality

$$\|v_1 \otimes v_2\|^2 = \langle v_1 \otimes v_2 | v_1 \otimes v_2 \rangle \ge 0.$$
(3.41)

Such an inequality turns out to be true for all elements in  $u \in V \otimes V$  as one can check by considering a linear expansion  $u = \sum_{j,k} c^{jk} e_j \otimes e_k$  in an orthonormal tensor product basis<sup>10</sup>.

In this setting we shall identify the linear subspace  $V \wedge V$  of antisymmetric contravariant 2-tensors and find as an immediate consequence

$$\|v_1 \wedge v_2\|^2 = \langle v_1 \wedge v_2 \mid v_1 \wedge v_2 \rangle \ge 0.$$
(3.42)

This last inequality is equivalent with the Cauchy-Schwarz inequality

$$||v_1||^2 ||v_2||^2 - |\langle v_1 | v_2 \rangle|^2 \ge 0$$
(3.43)

as one can easily check when applying the anti-symmetric tensor product expansion  $v_1 \wedge v_2 := \frac{1}{2}(v_1 \otimes v_2 - v_2 \otimes v_1)$  in (3.42). As an important tool for dealing with calculations, we shall identify here the determinant of a 2 × 2 matrix

$$\|v_1 \wedge v_2\|^2 = \frac{1}{2} \det \begin{pmatrix} \langle v_1 | v_1 \rangle & \langle v_1 | v_2 \rangle \\ \langle v_2 | v_1 \rangle & \langle v_2 | v_2 \rangle \end{pmatrix} \ge 0$$
(3.44)

which admits a straight forward generalization to the determinant of a  $n \times n$  matrix

$$||v_1 \wedge v_2 \dots \wedge v_n(p)||^2 = \frac{1}{n!} \det(\{\langle v_j | v_k \rangle\}_{j,k \in J}) \ge 0$$
(3.45)

when applying the inner product on  $V^{\otimes n}$  on the linear subspace  $V^{\wedge n}$  of contravariant antisymmetric n-tensors.

$$\langle u | u \rangle = \sum_{j,k,i,l} \bar{c}^{jk} c^{il} \langle e_j \otimes e_k | e_i \otimes e_l \rangle = \sum_{j,k,i,l} \bar{c}^{jk} c^{il} \delta_{jk} \delta_{il} = \sum_{i,k} |c^{ik}|^2 \ge 0.$$

<sup>&</sup>lt;sup>10</sup>Written explicitly, we have

**Remark 3.4.** By starting on the dual vector space  $V^*$ , we shall find the same inequalities on the corresponding tensor space  $(V^*)^{\wedge n}$  of covariant antisymmetric n-tensors.

## 3.2.2 Tensor field contractions

By promoting tensors to tensor *fields*, and by promoting Euclidean or Hilbert spaces to real Riemannian or complex Hilbert manifolds M respectively, we may recover on each tangent space  $T_pM \cong V$  the inequality relations of the previous section. This can directly be seen as follows. A covariant Riemannian<sup>11</sup> tensor field  $\eta$  admits a contraction with a vector field

$$v: M \to TM, p \mapsto (p, v(p))$$
 (3.46)

full filling the inequality

$$\|v(p)\|^2 \equiv \langle v(p) | v(p) \rangle_p \equiv \eta_p(v(p), v(p)) \ge 0.$$
(3.47)

In the same lines as in the previous section we may identify extended inequalities involving higher order tensor fields

$$v_1 \otimes v_2 : p \mapsto v_1 \otimes v_2(p) \in T_p M \otimes T_p M \tag{3.48}$$

yielding

$$|v_1 \otimes v_2(p)||^2 = \langle v_1 \otimes v_2(p) | v_1 \otimes v_2(p) \rangle_p$$
  

$$\equiv \eta_p(v_1(p), v_1(p)) \eta_p(v_2(p), v_2(p)) \ge 0.$$
(3.49)

For 1-forms  $\alpha: M \to T^*M, p \mapsto (p, \alpha(p))$ , we have a corresponding contraction with a contravariant Riemannian structure

$$\|\alpha(p)\|^2 \equiv G_p(\alpha(p), \alpha(p)) \ge 0 \tag{3.50}$$

<sup>&</sup>lt;sup>11</sup>The following discussion is also valid for Hermitian tensor fields on complex manifolds.

which may be generalized to a family of inequalites

$$\|\alpha_1 \wedge \alpha_2 \dots \wedge \alpha_n(p)\|^2 = \frac{1}{n!} \det(\{G_p(\alpha_j(p), \alpha_k(p))\}_{j,k \in J}) \ge 0$$
(3.51)

defined by n-forms  $\alpha_1 \wedge \alpha_2 \dots \wedge \alpha_n$ . Hence, to any given family of k differentiable functions on M there is a non-negative valued quantity defined by

$$\|df_1 \wedge df_2 \dots \wedge df_n(p)\|^2 = \frac{1}{n!} \det(\{G_p(df_j(p), df_k(p))\}_{j,k \in J}) \ge 0.$$
(3.52)

Now, let us apply this setting within the geometric formulation of quantum mechanics as follows. Recall in this regard the fundamental relation between algebraic and geometric structures captured in (2.74) by

$$f_{A \cdot B}(\psi) = (G + i\Omega)(df_A(\psi), df_B(\psi)), \qquad (3.53)$$

which encodes via the Hermitian structure  $G + i\Omega$  the non-commutative product of two Hermitian operators A, B on a Hilbert space  $\mathcal{H}$  in terms of a contraction on 1-forms generated by real valued quadratic functions

$$f_A(\psi) \equiv \operatorname{Tr}(\rho_{\psi}A) \equiv \rho_{\psi}(A), \quad \rho_{\psi} := \frac{|\psi\rangle \langle \psi|}{\langle \psi |\psi\rangle},$$
 (3.54)

defined on  $\mathcal{H}_0$ . Given a finite set of Hermitian operators  $\{A_j\}_{j\in J}$  on  $\mathcal{H}$ , we shall therefore find a direct generalization of (3.52) according to

$$\|df_{A_1} \wedge df_{A_2} \dots \wedge df_{A_n}(\psi)\|^2 = \frac{1}{n!} \det(\{(G+i\Omega)(df_{A_j}(\psi), df_{A_k}(\psi))\}_{j,k\in J}) \ge 0.$$
(3.55)

At this point we consider (3.53) to conclude

$$\|df_{A_1} \wedge df_{A_2} \dots \wedge df_{A_n}\|^2 = \frac{1}{n!} \det(\{(f_{A_j \cdot A_k}(\psi))\}_{j,k \in J}) \ge 0,$$
(3.56)

that is

$$\frac{1}{n!} \det(\{\rho_{\psi}(A_j \cdot A_k)\}_{j,k \in J}) \ge 0.$$
(3.57)

This inequality stays valid under the affine transformation map

$$A \mapsto \widetilde{A} = A - \rho_{\psi}(A) \mathrm{Id}_{\mathcal{H}}$$
(3.58)

and becomes rewritten in terms covariances of two operators

$$\rho_{\psi}(\widetilde{A_j} \cdot \widetilde{A_k}) = \operatorname{Cov}_{\rho_{\psi}}(A_j, A_k).$$
(3.59)

The latter relation follows directly from

$$\rho_{\psi}(\widetilde{A_{j}}\widetilde{A_{k}}) = \rho_{\psi}\left((A_{j} - \rho_{\psi}(A_{j})\mathrm{Id}_{\mathcal{H}})(A_{k} - \rho_{\psi}(A_{k})\mathrm{Id}_{\mathcal{H}})\right)$$
$$= \rho_{\psi}\left(A_{j}A_{k} + \rho_{\psi}(A_{j})\rho(A_{k})\mathrm{Id}_{\mathcal{H}} - \rho_{\psi}(A_{j})A_{k} - \rho_{\psi}(A_{k})A_{j}\right)$$
$$= \rho_{\psi}(A_{j}A_{k}) - \rho_{\psi}(A_{j})\rho_{\psi}(A_{k}), \qquad (3.60)$$

where we used the normalization  $\rho_{\psi}(\mathrm{Id}_{\mathcal{H}}) = 1$ . This coincides with the contraction  $K_{\mathcal{H}_0}(df_{A_j}, df_{A_k})|_{\psi}$  on the contravariant tensor field  $K_{\mathcal{H}_0}$  defined in (2.79) as being projectable on the space of complex rays  $\mathcal{R}(\mathcal{H})$  when considering 1-forms generated by  $f_{A_j}$ .

In conclusion, the determinant of a  $n \times n$  covariance matrix associated to a pure quantum state  $\rho_{\psi}$  applied on a finite set of Hermitian operators  $\{A_j\}_{j\in J}$  full fills

$$\frac{1}{n!} \det(\{\operatorname{Cov}_{\rho_{\psi}}(A_j, A_k)\}_{j,k \in J}) \ge 0.$$
(3.61)

This is obviously true for any subset  $\{A_j\}_{j\in J'} \subset \{A_j\}_{j\in J}$ . In particular all principal minors of a  $n \times n$  covariance matrix are therefore positive, making available Sylvester's criterion to conclude

$$\{\operatorname{Cov}_{\rho_{\psi}}(A_j, A_k)\}_{j,k\in J} \ge 0.$$
(3.62)

Within the framework of geometric quantum mechanics this is equivalent with

$$\{K_{\mathcal{H}_0}(df_{A_j}(\psi), df_{A_k}(\psi))\}_{j,k\in J} \ge 0.$$
(3.63)

As we will show in the next section, this inequality implies the *Robertson-Schrödinger inequality* when applied to the pullback tensor induced by a *Weyl systems*.

#### 3.2.3 The Robertson-Schrödinger inequality

The Fubini Study metric related pullback structure induced by a Weyl systems has been identified in section 2.1.6 according to

$$Cov_{\rho_{\psi}}((R(X_j)R(X_k))dv^j \otimes dv^k)$$
(3.64)

By the virtue of (3.62) we shall find here the inequality

$$Cov_{\rho_{\psi}}((R(X_j)R(X_k)) \equiv \sigma(\rho_{\psi}) + \frac{i}{4a^2}\omega \ge 0, \qquad (3.65)$$

which is equivalent that all principal minors of the covariance matrix are positive. A 2n-form

$$df_{\tilde{R}(X_1)} \wedge df_{\tilde{R}(X_2)} \dots \wedge df_{\tilde{R}(X_{2n})}(\rho_{\psi}) \quad \text{with} \ f_{\tilde{R}(X_j)}(\rho_{\psi}) := \operatorname{Tr}(\rho_{\psi}\tilde{R}(X_j)) \quad (3.66)$$

implies in particular according to section 3.2.2

$$\|df_{\tilde{R}(X_1)} \wedge df_{\tilde{R}(X_2)} \dots \wedge df_{\tilde{R}(X_{2n})}(\rho_{\psi})\|^2 = \frac{1}{(2n)!} \det(\{\rho_{\psi}(\tilde{R}(X_j)\tilde{R}(X_k)\}_{j,k\in J}) \ge 0,$$
(3.67)

that is

$$\frac{1}{(2n)!} \det\left(\sigma(\rho_{\psi}) + \frac{i}{4a^2}\omega\right) \ge 0.$$
(3.68)

In the simplest case of one degree of freedom modeled on  $\mathcal{H} \cong L^2(\mathbb{R})$ , the set of the Weyl-map generating elements are given up to an imaginary unit by two Hermitian operators

$$R(X_1) := Q, \quad R(X_2) := P$$
 (3.69)

related to position and momentum respectively. The Weyl-map induced pullback tensor coefficient matrix defines in dependence of the fiducial state vector  $\psi \in L^2(\mathbb{R})$  the covariance matrix

$$\sigma(\rho_{\psi}) + i\Omega = \begin{pmatrix} (\Delta_{\rho_{\psi}}Q)^2 & \operatorname{Cov}_{\rho_{\psi}}^s(Q, P) + \frac{i}{4a^2} \\ \operatorname{Cov}_{\rho_{\psi}}^s(P, Q) - \frac{i}{4a^2} & (\Delta_{\rho_{\psi}}P)^2 \end{pmatrix}$$
(3.70)

with the variance  $(\Delta_{\rho_{\psi}} A)^2 := \operatorname{Cov}_{\rho_{\psi}}(A, A) = \operatorname{Var}_{\rho_{\psi}}(A)$  and the symmetrized covariances  $\operatorname{Cov}_{\rho_{\psi}}^s(A, B) = \operatorname{Cov}_{\rho_{\psi}}^s(B, A)$ . By taking into account the 2-form  $df_{\tilde{Q}} \wedge df_{\tilde{P}}(\rho_{\psi})$ , we find according to (3.67)

$$\|df_{\tilde{Q}} \wedge df_{\tilde{P}}(\rho_{\psi})\|^2 = \frac{1}{2} \det\left(\sigma(\rho_{\psi}) + \frac{i}{4a^2}\omega\right) \ge 0, \qquad (3.71)$$

yielding the Robertson-Schrödinger inequality

$$(\triangle_{\rho_{\psi}}Q)^{2}(\triangle_{\rho_{\psi}}P)^{2} \ge \frac{1}{16a^{4}} + \operatorname{Cov}_{\rho_{\psi}}^{s}(Q, P)^{2}.$$
 (3.72)

Due to  $\mathrm{Cov}^s_{\rho_\psi}(Q,P)^2 \geqslant 0$  it implies the Heisenberg inequality

$$\Delta_{\rho_{\psi}} Q \Delta_{\rho_{\psi}} P \geqslant \frac{1}{4a^2} = \begin{cases} \frac{1}{2} \text{ canonical convention} \\ \frac{1}{4} \text{ q-optical convention.} \end{cases}$$
(3.73)

In contrast to the Robertson-Schrödinger inequality, we shall emphasize that the Heisenberg inequality is *not* invariant under symplectic transformations [66]. The invariance of the Robertson-Schrödinger inequality under symplectic transformations follows directly from the invariance of the determinant of the general inequality (3.67),

$$\sigma(\rho_{\psi}) + \frac{i}{4a^2}\omega \ge 0,$$

under volume preserving transformations including symplectic transformations as special case.

To make the difference between the Heisenberg inequality and the Robertson-Schrödingier inequality explicit we compute according to (2.54) the symplectic eigenvalues of

$$i\omega \cdot \sigma = i \begin{pmatrix} 0 & -\frac{1}{4a^2} \\ \frac{1}{4a^2} & 0 \end{pmatrix} \cdot \begin{pmatrix} \sigma_{1,1} & \sigma_{1,2} \\ \sigma_{1,2} & \sigma_{2,2} \end{pmatrix} = \begin{pmatrix} -\frac{\sigma_{1,2}}{4a^2} & -\frac{\sigma_{2,2}}{4a^2} \\ \frac{\sigma_{1,1}}{4a^2} & \frac{\sigma_{1,2}}{4a^2} \end{pmatrix}$$
(3.74)

and find

spec
$$(i\omega \cdot \sigma) = \left\{ \pm \frac{1}{4} \sqrt{\frac{\sigma_{1,2}^2 - \sigma_{1,1}\sigma_{2,2}}{a^4}} \right\}.$$
 (3.75)

Hence, the Heisenberg-inequality fails to provide a symplectic invariant due to the lack of the symmetric covariances  $\sigma_{1,2}^2$ .

#### 3.2.4 Quantum Cramér Rao Inequality

Let  $\mathcal{M}$  be a subset in a Hilbert space  $\mathcal{H}$  provided by any smooth curve of state vectors

$$[0,1]_{\mathbb{CR}} \to \mathcal{H}, \quad \lambda \mapsto |\psi_{\lambda}\rangle.$$
 (3.76)

On this subset of Hilbert space vectors

$$\{\psi_{\lambda}\}_{\lambda\in[0,1]} = \mathcal{M} \subset \mathcal{H}$$
(3.77)

we may assume an operator  $\frac{\partial}{\partial \lambda}$  to be well defined. Moreover we may consider a further operator  $\hat{\lambda}$  on  $\mathcal{M}$ , such that

$$f_{\hat{\lambda}}(\psi_{\lambda}) = \frac{\langle \psi_{\lambda} | \lambda | \psi_{\lambda} \rangle}{\langle \psi_{\lambda} | \psi_{\lambda} \rangle} \equiv \lambda$$
(3.78)

holds. By applying on this setting the geometric inequalities as derived in section 3.2.2 according to (3.62) - (3.63) one finds

$$K_{\mathcal{H}_0}(df_{\frac{\partial}{\partial\lambda}}, df_{\frac{\partial}{\partial\lambda}})|_{\psi_{\lambda}}K_{\mathcal{H}_0}(df_{\hat{\lambda}}, df_{\hat{\lambda}})|_{\psi_{\lambda}} - |K_{\mathcal{H}_0}(df_{\frac{\partial}{\partial\lambda}}, df_{\hat{\lambda}})|_{\psi_{\lambda}}|^2 = (3.79)$$

$$Var_{\psi_{\lambda}}(\frac{\partial}{\partial\lambda})Var_{\psi_{\lambda}}(\hat{\lambda}) - |Cov_{\psi_{\lambda}}(\frac{\partial}{\partial\lambda},\hat{\lambda})|^{2} \ge 0 \quad (3.80)$$

which turns out to imply a quantum version of the Cramér Rao inequality [67]

$$Var_{\psi_{\lambda}}(\lambda) \ge \frac{|Cov_{\psi_{\lambda}}(\frac{\partial}{\partial\lambda},\lambda)|^{2}}{Var_{\psi_{\lambda}}(\frac{\partial}{\partial\lambda})} \ge \frac{1}{Var_{\psi_{\lambda}}(\frac{\partial}{\partial\lambda})}$$
(3.81)

as being used in quantum estimation theory [68] by associating  $\hat{\lambda}$  to an *un-biased estimator*<sup>12</sup> of the curve parameter  $\lambda$ . The implication of the Cramér Rao inequality in the above quantum version may be subsumed as follows. No matter which estimator one uses to approach a parameter in a quantum mechanical experimental setting, the variance of any estimator will always be bounded by the variance

$$Var_{\psi_{\lambda}}\left(\frac{\partial}{\partial\lambda}\right) = K_{\mathcal{H}_{0}}(df_{\frac{\partial}{\partial\lambda}}, df_{\frac{\partial}{\partial\lambda}})|_{\psi_{\lambda}}$$
(3.82)

$$= f_{\frac{\partial}{\partial\lambda}\frac{\partial}{\partial\lambda}}(\psi_{\lambda}) + f_{\frac{\partial}{\partial\lambda}}f_{\frac{\partial}{\partial\lambda}}(\psi_{\lambda})$$
(3.83)

$$= \langle \partial_{\lambda}\psi_{\lambda} | \partial_{\lambda}\psi_{\lambda} \rangle - \langle \psi_{\lambda} | \partial_{\lambda}\psi_{\lambda} \rangle^{2}. \qquad (3.84)$$

Actually, this turns out to be the pullback coefficient of the Fubini Study metric related pullback structure  $\kappa_{\mathcal{H}_0}$  as defined in (2.11). As matter of fact,

<sup>12</sup> The basic definition is assured here by (3.78). We will focus on this notion once again more in detail in section 5.

one finds

$$\kappa_{\mathcal{M}} \equiv \left\langle \partial_{\lambda} \psi_{\lambda} \left| \partial_{\lambda} \psi_{\lambda} \right\rangle d\lambda \otimes d\lambda - \left\langle \psi_{\lambda} \left| \partial_{\lambda} \psi_{\lambda} \right\rangle^{2} d\lambda \otimes d\lambda \right.$$
(3.85)

As being related to the pullback to the Fubini Study metric, it 'closes a circle' to the *quantum Fisher information* as described in section 1.4. Actually, the quantum Fisher information defined on pure states reduces to the classical Fisher information for Lagriangian submanifolds as we have seen in seen in section 2.1.3 (see also [7]). In conclusion, any quantum Cramér Rao inequality for estimating the 'position' on a curve of pure quantum states reduces to the classical Cramér Rao inequality. Note that the above quantum Cramér Rao inequality, even though being 'classical' may be seen as a generalization of the Heisenberg inequality as it allows to considers the pullback on both unitarily and non-unitarily generated curves.

# 4 Entanglement Monotones

Symmetries of Hamiltonian dynamical systems are related to constants of motion by virtue of Noether's theorem [54]. Constants of motion provide in turn a method to reduce the numbers of degrees of freedom. For instance, invariance under the symmetry group of time translations  $(\mathbb{R}, +)$  implies the identification of energy as constant of motion, and therefore the identification of time independent energy surfaces embedded in a phase space  $\mathbb{R}^{2n}$ , via the inverse  $H^{-1}(E) \subset \mathbb{R}^{2n}$  of the underlying Hamiltonian function. A Hamiltonian system generated by H will thus be constrained for all times on exactly one of these surfaces in dependence of the initial state  $v_0 \in \mathbb{R}^{2n}$ .

In the following section we shall propose a similar approach for the identification of what one could call *entanglement invariants* by virtue of functions being constant on 'entanglement surfaces' defining submanifolds of quantum states with equivalent amount of entanglement. In contrast to the foliation into constant energy surfaces defined via the inverse of a time invariant Hamilton function however, we shall tackle here an inverse problem:

*Given* a foliation of a Hilbert space into submanifolds of quantum states with same entanglement, how do we identify a function varying from leave to leave but being constant within each leave?

First of all we note that a general quantum evolution<sup>13</sup> will *not* keep the entanglement in a composite system  $\mathcal{H}_A \otimes \mathcal{H}_B$  unchanged. The invariance of entanglement is given only for non-interacting closed subsystems indeed.

From a kinematical perspective, this implies the identification of the *local* unitary subgroup

$$SU(\mathcal{H}_A) \times SU(\mathcal{H}_B) \subset SU(\mathcal{H}_A \otimes \mathcal{H}_B)$$
 (4.1)

as the fundamental symmetry group of entanglement. The identification of

 $<sup>^{13}</sup>$ According to the geometric formulation of quantum mechanics, a unitary quantum evolution is an integral curve of an Hamiltonian vector field being a Killing vector field.

'entanglement surfaces' is therefore provided by a stratification into orbits generated by this subgroup [69,70]. Such a stratification induces within the geometric formulation of quantum mechanics a pullback of the Fubini Study metric related structure (2.11) from  $\mathcal{H} := \mathcal{H}_A \otimes \mathcal{H}_B$  to each single orbit. In the simplest case of two qubits one may identify here local unitary orbits  $\Gamma_{\psi_{\lambda}} \subset \mathbb{C}^2 \otimes \mathbb{C}^2$  in dependence of a 1-parameter family of fiducial state vectors

$$[0,1]_{\mathbb{CR}} \to \mathbb{C}^2 \otimes \mathbb{C}^2, \quad \lambda \mapsto |\psi_\lambda\rangle$$
 (4.2)

intersecting each submanifold  $\Gamma_{\psi_{\lambda}}$  exactly once. The pullback on each single orbit  $\Gamma_{\psi_{\lambda}}$  may therefore considered complementary to the pullback on the curve (4.2) provided by a structure as defined in the previous section in (3.85). The following subsection 4.1 will give a short review how to identify pullback structures associated to unitarily generated orbits in dependence of a given general initial state vector of a composite bipartite Hilbert space in finite but arbitrary high dimensions as discussed in [1–3]. These structures will imply of a *qualitative* characterization of entanglement making available the identification of separable and 'maximal entangled' state vectors in purely geometrical terms. Thereafter, in the following subsections 4.2 - 4.5, we shall consider recent research results [4–6] based on the possibility to identify invariant functions under the local unitary group arising as inner products on invariant tensor fields. This will lead us to a *quantitative* characterization of entanglement in tensorial terms.

# 4.1 Separability and Lagrangian entanglement

By considering the reducible representation

$$\mathcal{G} \equiv SU(n) \times SU(n) \to SU(n^2)$$
$$g \equiv (g_A, g_B) \mapsto U(g) \equiv g_A \otimes g_B = (g_A \otimes \mathbb{1}_{\mathcal{H}_B})(\mathbb{1}_{\mathcal{H}_A} \otimes g_B) \quad (4.3)$$

infinitesimal generated by traceless orthonormal Hermitian matrices  $\{\sigma_j\}_{j\in J}$ tensored by the identity  $\mathbbm{1}_{\mathcal{H}_A}$  of a subsystem

$$\operatorname{Lie}(\mathcal{G}) \equiv su(n) \oplus su(n) \to su(n^2)$$
$$X_j \mapsto iR(X_j) \equiv \begin{cases} i\sigma_j \otimes \mathbb{1} & \text{for } 1 \leq j \leq n^2\\ \mathbb{1} \otimes i\sigma_{j-n^2} & \text{for } n^2 + 1 \leq j \leq 2n^2, \end{cases}$$
(4.4)

one finds according to theorem 2.1 a pull-back tensor field on the Lie group

$$\iota_{\mathcal{G}}^* \kappa_{\mathcal{H}_0} = Cov_{\rho_{\psi}}(R(X_j)R(X_k))\theta^j \otimes \theta^k$$
$$= \underbrace{(\rho_{\psi}([R(X_j)R(X_k)]_+) - \rho_{\psi}(R(X_j))\rho_{\psi}(R(X_k)))\theta^j \odot \theta^k}_{=\iota_{\mathcal{G}}^* \eta} + i\underbrace{\rho_{\psi}([R(X_j)R(X_k)]_-)\theta^j \wedge \theta^k}_{=\iota_{\mathcal{G}}^* \omega}$$

which decomposes for all  $\rho_{\psi} \in D^1(\mathbb{C}^n \otimes \mathbb{C}^n)$  into a symmetric and an antisymmetric coefficient matrix <sup>14</sup>

$$(\kappa_{jk}^{\rho_{\psi}}) = \begin{pmatrix} (\eta_{(jk)}^{\rho_A}) & (C_{jk}^{\rho_{\psi}}) \\ (C_{jk}^{\rho_{\psi}}) & (\eta_{(jk)}^{\rho_B}) \end{pmatrix} + i \begin{pmatrix} (\omega_{[jk]}^{\rho_A}) & 0 \\ 0 & (\omega_{[jk]}^{\rho_B}) \end{pmatrix},$$
(4.8)

$$\eta_{(jk)}^{\rho_A} = \frac{2}{n} \delta_{jk} + \rho_A(\sigma_r) d_{jkr} - \rho_A(\sigma_j) \rho_A(\sigma_k)$$
(4.9)

$$\omega_{[jk]}^{\rho_A} = \rho_A(\sigma_r) c_{jkr} \tag{4.10}$$

$$C_{jk}^{\rho_{\psi}} = \rho_{\psi}(\sigma_j \otimes \sigma_k) - \rho_{\psi}(\sigma_j)\rho_{\psi}(\sigma_k)$$
(4.11)

<sup>14</sup>Written explicitly one finds

$$\eta_{(jk)}^{\rho_A} = \rho_{\psi}([\sigma_j, \sigma_k]_+ \otimes \mathbb{1}) - \rho_{\psi}(\sigma_j \otimes \mathbb{1})\rho_{\psi}(\sigma_k \otimes \mathbb{1})$$

$$\tag{4.5}$$

$$=\rho_A([\sigma_j,\sigma_k]_+) - \rho_A(\sigma_j)\rho_A(\sigma_k) = \frac{2}{n}\delta_{jk} + \rho_A(\sigma_r)d_{jkr} - \rho_A(\sigma_j)\rho_A(\sigma_k)$$
(4.6)

$$\omega_{[jk]}^{\rho_A} = \rho_{\psi}([\sigma_j, \sigma_k]_- \otimes \mathbb{1}) = \rho_A([\sigma_j, \sigma_k]_-) = \rho_A(\sigma_r)c_{jkr}$$

$$(4.7)$$

by vitue of the symmetric and anti-symmetric structure constants  $d_{jkr}$  and  $c_{jkr}$  of the Lie algebra of SU(n).

#### 4.1.1 Segre embeddings seen from the Hilbert space

While the anti-symmetric part  $\omega^{\rho_{\psi}}$  admits a splitting for all entangled fiducial state vectors, we will find such a splitting in the symmetric part  $\eta^{\rho_{\psi}}$  according to the following necessary and sufficient condition [1,3].

**Theorem 4.1.**  $\rho_{\psi} \in D^1(\mathbb{C}^n \otimes \mathbb{C}^n)$  is separable iff

$$\eta_{SU(n)\times SU(n)}^{\rho_{\psi}} = \eta_{SU(n)\times SU(n)}^{\rho_A \otimes \rho_B} = \eta_{SU(n)}^{\rho_A} \oplus \eta_{SU(n)}^{\rho_B}$$
(4.12)

Hence, a Segre embedding [42, 43, 71]

$$\mathcal{R}(\mathcal{H}_A) \times \mathcal{R}(\mathcal{H}_B) \hookrightarrow \mathcal{R}(\mathcal{H}_A \otimes \mathcal{H}_B)$$
(4.13)

becomes detectable from the point of view of the Hilbert space iff

$$C_{jk}^{\rho_{\psi}} = 0. (4.14)$$

For general state vectors we shall remark the following geometric interpretation of the symmetric part  $\eta_{SU(n)\times SU(n)}$ . It encounters as pullback induced by the projection

$$SU(n) \times SU(n) \to SU(n) \times SU(n) / \mathcal{G}_{0,\psi}$$
 (4.15)

on the orbits associated with the isotropy group

$$\mathcal{G}_{0,\psi} := \{g \in SU(n) \times SU(n) | U(g)\rho_{\psi}U(g)^{\dagger} = \rho_{\psi}\}$$

$$(4.16)$$

the complete information of a family of *Riemannian* tensor fields on each given orbit.

#### 4.1.2 Symplectic orbits seen from the Hilbert space

The anti-symmetric part

$$\omega_{SU(n)\times SU(n)}^{\rho_{\psi}} = \rho_{\psi}([R(X_j), R(X_k)]_{-})\theta^j \wedge \theta^k$$
(4.17)

splits — as indicated above and in crucial contrast to the symmetric part — for all entangled state vectors into two families of tensor fields

$$\omega_{SU(n)\times SU(n)}^{\rho_{\psi}} = \omega_{SU(n)}^{\rho_A} \oplus \omega_{SU(n)}^{\rho_B}.$$
(4.18)

Each family is defined on a corresponding SU(n)-subgroup of  $SU(n) \times SU(n)$ by the *reduced density state dependent* anti-symmetric structures

$$\omega_{SU(n)}^{\rho_A} = \rho_{\psi}([\sigma_j, \sigma_k]_{-} \otimes \sigma_0)\theta^j \wedge \theta^k \tag{4.19}$$

$$= \rho_A([\sigma_j, \sigma_k]_-)\theta^j \wedge \theta^k, \qquad (4.20)$$

and  $\omega_{SU(n)}^{\rho_B} = \rho_B([\sigma_j, \sigma_k]_-)\theta^j \wedge \theta^k$  respectively.

Each of the latter anti-symmetric tensor fields may therefore identified with the quotient space projection-induced pullback of a symplectic structure  $\omega^{A}_{SU(n)/\mathcal{G}_{0}}$  which lives on a co-adjoint unitary orbit of (reduced) density states

$$g_A \rho_A g_A^{\dagger}, \tag{4.21}$$

with  $g_A \in SU(n)$  [72]. Lagrangian orbits generated by the local unitary group admit therefore a distinguished role, in particular, due to the following definition and associated implication [3,4]:

**Definition 4.2.**  $\rho_{\psi} \in D^1(\mathbb{C}^n \otimes \mathbb{C}^n)$  is called Lagragian entangled if

$$\omega_{SU(n)\times SU(n)}^{\rho_{\psi}} = 0.$$

**Theorem 4.3.**  $\rho_{\psi} \in D^1(\mathbb{C}^n \otimes \mathbb{C}^n)$  is Lagragian entangled iff the reduced

#### density state is maximal mixed.

This theorem links to the standard definition of 'maximally entangled' pure states as used in the literature [73]. Actually, the identification of maximal entangled states is *not* a priori obvious *before* having defined a measure of entanglement. As a matter of fact, the identification of the von Neumann entropy (1.8) as unique measure is established *after* rather then before the definition of maximal entanglement [19]. Such an approach may appear artificial from a general point of view indeed. In particular, why should the existence of an infinite amount of entanglement a priori be excluded? The notion of *Lagrangian entanglement*, as proposed here, evades this conceptual problem by being defined a priori without any quantitative association.

# 4.2 Inner products on tensor fields and intermediate entanglement

In the previous subsection we considered a qualitative description of entanglement by the identification of submanifolds containing either separable or Lagrangian entangled quantum state vectors. Actually, most state vectors in the composite Hilbert space turn out to be in neither of both of these two submanifolds. This can be illustrated in the simplest case of a two qubit Hilbert space

$$\mathcal{H} \equiv \mathcal{H}_A \otimes \mathcal{H}_B \cong \mathbb{C}^2 \otimes \mathbb{C}^2 \tag{4.22}$$

as follows. Any coordinate representation of a state vector in  $S(\mathcal{H})$  may be transformed here by means of a local unitary transformation to a Schmidt basis decomposition

$$|\psi_{\lambda}\rangle = \sqrt{\lambda} |00\rangle + \sqrt{1-\lambda} |11\rangle, \quad \lambda \in [0,1].$$
 (4.23)

In this way one encounters a topological distinction into three type of local unitary orbits

$$SU(2) \times SU(2)/\mathcal{G}_{0,\lambda}$$
 (4.24)

with the isotropy group

$$\mathcal{G}_{0,\lambda} := \{ g \in SU(2) \times SU(2) | U(g) \rho_{\psi_{\lambda}} U(g)^{\dagger} = \rho_{\psi_{\lambda}} \}$$
(4.25)

in dependence of fiducial states  $\rho_{\psi_{\lambda}} = |\psi_{\lambda}\rangle \langle \psi_{\lambda}|$  being either Lagrangian entangled ( $\lambda = 1/2$ ), separable ( $\lambda \in \{0, 1\}$ ) or neither Lagrangian nor separable ( $\lambda \in (0, 1/2) \cup (1/2, 1)$ ). They yield orbits of dimensions three, four and five respectively [69, 70]. The 'dense set' of state vectors is therefore made of an foliation into five dimensional orbits of what we could call *intermediate* entangled state vectors.

In the following we'll show how to distinguish the topologically equivalent orbits of intermediate states in geometrical terms by means of the corresponding Fubini Study metric related pullback procedure we discussed so far.

**Remark 4.4.** In higher dimensional bipartite systems there will be more then three topological inequivalent type of local unitary orbits [69, 70]. Indeed, different topological invariants like the dimension of a manifold will nevertheless be detected from the metrical pullback structures on the individual orbit. Such topological invariants may be detected in particular within our approach by the degeneracy of the pullback on the Lie group of local unitary transformation.

For the purpose to distinguish state vectors living in different but topological equivalent orbits we may consider invariant functions under local unitary transformations provided by a (super) Hermitian inner product

$$f(\psi) := \left\langle \kappa_{SU(n) \times SU(n)}^{\rho_{\psi}} \middle| \kappa_{SU(n) \times SU(n)}^{\rho_{\psi}} \right\rangle$$
(4.26)

on invariant tensor fields on  $SU(n) \times SU(n)$ . Invariant tensor fields on general

Lie groups may be defined in a constructive way either by *invariant operator* valued tensor fields (see section 4.5) or by the pullback of the Fubini Study metric seen from the Hilbert space as considered so far. The inner product on the latter class of invariant tensor fields will be defined in general terms as follows [4].

We define for any given Lie group  $\mathcal{G}$  associated with a unitary representation induced pullback structure

$$\kappa_{\mathcal{G}}^{
ho_{\psi}} = \kappa_{jk}^{
ho_{\psi}} heta^{j} \otimes heta^{k}$$

the inner product

$$\left\langle \kappa_{\mathcal{G}}^{\rho_{\psi}} \left| \kappa_{\mathcal{G}}^{\rho_{\psi}} \right\rangle := \bar{\kappa}_{jk}^{\rho_{\psi}} \kappa_{rl}^{\rho_{\psi}} \left\langle \theta^{j} \otimes \theta^{k} \left| \theta^{r} \otimes \theta^{l} \right\rangle.$$

To keep the formulas as readable as possible, we shall omit in the following the dependency on the fiducial state  $\rho_{\psi}$ . With

$$\left\langle \theta^{j} \otimes \theta^{k} \middle| \theta^{r} \otimes \theta^{l} \right\rangle = \left\langle \theta^{j} \middle| \theta^{r} \right\rangle \left\langle \theta^{k} \middle| \theta^{l} \right\rangle = \delta^{jr} \delta^{kl}$$
(4.27)

one finds then

$$\langle \kappa_{\mathcal{G}} | \kappa_{\mathcal{G}} \rangle = \bar{\kappa}_{jk} \kappa_{rl} \delta^{jr} \delta^{kl} = \bar{\kappa}_{jk} \kappa_{jk}.$$
(4.28)

## 4.2.1 Inner products on higher order tensor fields

In the following we may consider a class of  $\mathcal{G}$ -invariant functions arising from higher order tensor fields

$$\kappa_{\mathcal{G}}^{\otimes n} := \bigotimes_{k=1}^{n} \kappa_{\mathcal{G}} \tag{4.29}$$

by virtue of their corresponding inner product

$$\left\langle \kappa_{\mathcal{G}}^{\otimes n} \middle| \kappa_{\mathcal{G}}^{\otimes n} \right\rangle \tag{4.30}$$

$$= \langle \kappa_{\mathcal{G}} | \kappa_{\mathcal{G}} \rangle^{n} . \tag{4.31}$$

This can be be illustrated in the simplest case of n = 2 as follows. Consider a general covariant tensor of order four

$$T := \sum_{j_1 j_2 j_3 j_4} T_{j_1 j_2 j_3 j_4} \theta^{j_1} \otimes \theta^{j_2} \otimes \theta^{j_3} \otimes \theta^{j_4}.$$
(4.32)

The inner product  $\langle T | T \rangle$  on this tensor reads

$$\left\langle \sum_{j_1 j_2 j_3 j_4} T_{j_1 j_2 j_3 j_4} \theta^{j_1} \otimes \theta^{j_2} \otimes \theta^{j_3} \otimes \theta^{j_4} \right| \sum_{k_1 k_2 k_3 k_4} T_{k_1 k_2 k_3 k_4} \theta^{k_1} \otimes \theta^{k_2} \otimes \theta^{k_3} \otimes \theta^{k_4} \right\rangle$$

$$= \sum_{j_1 j_2 j_3 j_4} \sum_{k_1 k_2 k_3 k_4} \overline{T}_{j_1 j_2 j_3 j_4} T_{k_1 k_2 k_3 k_4} \left\langle \theta^{j_1} \otimes \theta^{j_2} \otimes \theta^{j_3} \otimes \theta^{j_4} \right| \theta^{k_1} \otimes \theta^{k_2} \otimes \theta^{k_3} \otimes \theta^{k_4} \right\rangle$$

$$= \sum_{j_1 j_2 j_3 j_4} \sum_{k_1 k_2 k_3 k_4} \overline{T}_{j_1 j_2 j_3 j_4} T_{k_1 k_2 k_3 k_4} \delta^{j_1 k_1} \delta^{j_2 k_2} \delta^{j_3 k_3} \delta^{j_4 k_4}$$

$$= \sum_{j_1 j_2 j_3 j_4} \overline{T}_{j_1 j_2 j_3 j_4} T_{j_1 j_2 j_3 j_4}. \quad (4.33)$$

A special tensor of order four may arise from the tensor product of two tensors of order two

$$T \equiv \left(\sum_{j_1 j_2} \kappa_{j_1 j_2} \theta^{j_1} \otimes \theta^{j_2}\right) \otimes \left(\sum_{j_3 j_4} \kappa_{j_3 j_4} \theta^{k_3} \otimes \theta^{j_4}\right)$$
$$= \sum_{j_1 j_2} \sum_{j_3 j_4} \kappa_{j_1 j_2} \kappa_{j_3 j_4} \theta^{j_1} \otimes \theta^{j_2} \otimes \theta^{j_3} \otimes \theta^{j_4}.$$
(4.34)

In this case, the tensor coefficients in (4.38) factorize into tensor coefficients of order two according to

$$T_{j_1 j_2 j_3 j_4} \equiv \kappa_{j_1 j_2} \kappa_{j_3 j_4}. \tag{4.35}$$

Note that this is not true for general tensors of order four.

Thus, by applying the inner product (4.33) on the special case given by

the tensor in (4.34), we find

$$\langle T | T \rangle = \sum_{j_1 j_2 j_3 j_4} \bar{T}_{j_1 j_2 j_3 j_4} T_{j_1 j_2 j_3 j_4} = \sum_{j_1 j_2 j_3 j_4} \bar{\kappa}_{j_1 j_2} \bar{\kappa}_{j_3 j_4} \kappa_{j_1 j_2} \kappa_{j_3 j_4}$$
$$= \sum_{j_1 j_2 j_3 j_4} \bar{\kappa}_{j_1 j_2} \kappa_{j_1 j_2} \bar{\kappa}_{j_3 j_4} \kappa_{j_3 j_4} = \sum_{j_1 j_2} \bar{\kappa}_{j_1 j_2} \kappa_{j_1 j_2} \sum_{j_3 j_4} \bar{\kappa}_{j_3 j_4} \kappa_{j_3 j_4}$$
(4.36)

$$= \langle \kappa_{\mathcal{G}} | \kappa_{\mathcal{G}} \rangle \langle \kappa_{\mathcal{G}} | \kappa_{\mathcal{G}} \rangle = \langle \kappa_{\mathcal{G}} | \kappa_{\mathcal{G}} \rangle^{2} .$$
(4.37)

The generalization is straight forward and goes as follows. The inner product  $\langle T | T \rangle$  on a general covariant tensor<sup>15</sup>

$$T := T_{j_1 j_2 \dots j_m} \theta^{j_1} \otimes \theta^{j_2} \dots \otimes \theta^{j_m}$$

$$(4.38)$$

of order m reads

$$\bar{T}_{j_1..j_m} T_{k_1..k_m} \left\langle \theta^{j_1} \otimes .. \otimes \theta^{j_m} \middle| \theta^{k_1} \otimes .. \otimes \theta^{k_m} \right\rangle$$
$$= \bar{T}_{j_1..j_m} T_{k_1..k_m} \delta_{j_1k_1} ... \delta_{j_mk_m} = \bar{T}_{j_1..j_m} T_{j_1..j_m}.$$
(4.39)

Now, we consider a tensor of even order m = 2n constructed from the n-th tensor product of order two tensors

$$T \equiv (T_{j_1 j_2} \theta^{j_1} \otimes \theta^{j_2})^{\otimes n}$$
  
=  $T_{j_1 j_2} T_{j_3 j_4} ... T_{j_{m-1} j_m} \theta^{j_1} \otimes \theta^{j_2} \otimes \theta^{j_3} \otimes \theta^{j_4} ... \theta^{j_{m-1}} \otimes \theta^{j_m}$   
=  $\prod_{r=1}^n T_{j_{2r-1} j_{2r}} \bigotimes_{r=1}^n \theta^{j_{2r-1}} \otimes \theta^{j_{2r}}.$  (4.40)

The tensor coefficients in (4.38) factorize in this special case (in each term

 $<sup>$^{15}$</sup>$  If not differently stated, we shall from now on use the Einstein convention by summing over same indices.

of the sum over same indices) into tensor coefficients of order two

$$T_{j_1 j_2 \dots j_m} = \prod_{r=1}^n T_{j_{2r-1} j_{2r}}.$$
(4.41)

Hence, with the inner product (4.39) one concludes

$$\langle T | T \rangle = \bar{T}_{j_1..j_m} T_{j_1..j_m} = \prod_{r=1}^n \bar{T}_{j_{2r-1}j_{2r}} \prod_{r=1}^n T_{j_{2r-1}j_{2r}}$$
$$= \prod_{r=1}^n \bar{T}_{j_{2r-1}j_{2r}} T_{j_{2r-1}j_{2r}} = \prod_{r=1}^n \langle T | T \rangle = \langle T | T \rangle^n .$$
(4.42)

This proofs the inner product relation (4.31). As a consequence, we may apply the inner product on the tensor product of the symmetric part and the tensor product of the anti-symmetric part separately and find

$$\left\langle \eta_{\mathcal{G}}^{\otimes n} \left| \eta_{\mathcal{G}}^{\otimes n} \right\rangle = \left\langle \eta_{\mathcal{G}} \left| \eta_{\mathcal{G}} \right\rangle^{n},$$

$$(4.43)$$

$$\left\langle \omega_{\mathcal{G}}^{\otimes n} \left| \omega_{\mathcal{G}}^{\otimes n} \right\rangle = \left\langle \omega_{\mathcal{G}} \left| \omega_{\mathcal{G}} \right\rangle^{n}.$$
 (4.44)

**Remark 4.5.** Actually, this will be not the case whenever one considers the inner product on the symmetrization of the tensor products on the symmetric part, and correspondently, the inner product on the anti-symmetrization of the tensor products on the anti-symmetric part respectively. Such an approach may be seen related to the notion of Poincaré invariants which are constructed from higher order anti-symmetric structures  $\omega^{\wedge n} := \omega \wedge \omega ... \wedge \omega$  [74].

# 4.3 Entanglement monotones on two qubits

In the following section we shall apply the inner product on higher order tensor fields to the case of the local unitary group

$$\mathcal{G} \equiv SU(n) \times SU(n) \tag{4.45}$$
with particular focus on  $n \equiv 2$ . In this case we may map a family of fiducial states

$$\rho_{\lambda} = \left|\psi_{\lambda}\right\rangle \left\langle\psi_{\lambda}\right| \tag{4.46}$$

associated to a family of state vectors in a Schmidt basis decomposition

$$|\psi_{\lambda}\rangle = \sqrt{\lambda} |00\rangle + \sqrt{1-\lambda} |11\rangle, \quad \lambda \in [0,1]$$
 (4.47)

to a corresponding family of induced pullback tensor fields

$$\kappa_{SU(2)\times SU(2)}^{\rho_{\lambda}} = \eta_{SU(2)\times SU(2)}^{\rho_{\lambda}} + i\omega_{SU(2)\times SU(2)}^{\rho_{\lambda}}$$
(4.48)

on the Lie group  $SU(2) \times SU(2)$ . The inner products (4.43) and (4.44) applied here to the higher order tensor products of the symmetric and the anti-symmetric part of (4.48) respectively, establish the main result of the present chapter: We find *entanglement monotones* for the inner product on the tensor products of the symmetric part and *purity monotones* for the inner product on the tensor products of the anti-symmetric part as illustrated within an appropriate normalization in figure 1 for the first five tensor power orders. As normalization we use in this regard the factors

$$\frac{1}{(2\text{Dim}(SU(2) \times SU(2))^n)} = \frac{1}{12^n},$$
(4.49)

for the inner products (4.43) and the factors

$$\frac{1}{\text{Dim}(S^2 \times S^2)^n} = \frac{1}{4^n}.$$
(4.50)



Figure 1: A class of entanglement and purity monotones constructed from  $SU(2) \times SU(2)$  invariant tensor fields  $\eta_{SU(2) \times SU(2)}^{\otimes n}$  and  $\omega_{SU(2) \times SU(2)}^{\otimes n}$ . The monotones arise here by considering an inner product yielding the invariant functions (4.51) and (4.52) respectively. The black curve corresponds in each case to n = 1 and the following colored curves corresponds to higher order tensor field inner products with  $n \in \{2, 3, 4, 5\}$ . The entanglement monotones admit in contrast to the purity monotones a normalization in dependence of the tensor order n.

for the inner products (4.44). The resulting functions are higher order polynomials on the Schmidt coefficient  $\lambda$ 

$$\frac{1}{12^{n}} \left\langle \eta_{SU(2)\times SU(2)}^{\rho_{\lambda}} \left| \eta_{SU(2)\times SU(2)}^{\rho_{\lambda}} \right\rangle^{n} = \left( \frac{16\lambda^{4}}{3} - \frac{32\lambda^{3}}{3} + 4\lambda^{2} + \frac{4\lambda}{3} + \frac{1}{3} \right)^{n} (4.51)$$

$$\frac{1}{4^{n}} \left\langle \omega_{SU(2)\times SU(2)}^{\rho_{\lambda}} \left| \omega_{SU(2)\times SU(2)}^{\rho_{\lambda}} \right\rangle^{n} = \left( 4\lambda^{2} - 4\lambda + 1 \right)^{n}. \quad (4.52)$$

establishing a quantitative justification for the identification between maximal entangled states and Lagrangian entangled states ( $\lambda = 1/2$ ) on the value '1' for the entanglement monotones and the value '0' for the purity monotones respectively. The purity monotone is normalized for separable states ( $\lambda \in \{0,1\}$ ) to '1'. In contrast, we find in the entanglement monotone a normalization

$$\frac{1}{12^n} \left\langle \eta_{SU(2)\times SU(2)}^{\rho_{\lambda}} \left| \eta_{SU(2)\times SU(2)}^{\rho_{\lambda}} \right\rangle^n \right|_{\lambda \in \{0,1\}} = \frac{1}{3^n}, \tag{4.53}$$

being dependent on the tensor field order n. We may therefore recover for  $n \to \infty$  the 'standard normalization' for separable states. This indicates that the inner products on *n*-order tensor products of the symmetric tensor field provide an *approximation* to a 'bona fide' entanglement measure with increasing n. This approximation will provide an advantage in contrast to standard entanglement measures when testing their quantum estimation efficiency as shown later on in section 5.

## 4.4 Towards a generalized algorithm

Before we proceed, let us outline a possible bigger picture into which we may set the considered approach so far by closing a *circle* to the introduction of the present chapter 4. For this purpose let us recall the Hamiltonian

$$H(q,p) = \frac{1}{2m}p^2 + kq^2.$$
 (4.54)

of a classical Harmonic oscillator. Here one finds associated energy ellipses

$$H^{-1}(E) := \Gamma_E \subset \mathbb{R}^2 \tag{4.55}$$

in dependence of the mass m and the constant k. If we set m = k = 1 we may identify the ellipses with circles  $S_{\sqrt{E}}^2$  each with a radius proportional to the energy  $H(q, p) = E \in \mathbb{R}_+$ . The set of energy circles provides a foliation

$$\mathbb{R}_0^2 \cong \bigcup_{E \in \mathbb{R}_+} \Gamma_E = \bigcup_{E \in \mathbb{R}_+} S_{\sqrt{E}}^1 \tag{4.56}$$

of the phase space into circles with different radius  $\sqrt{E}$ . The radius may be related in this regard to a parametrization of the set of orbits given by the quotient

$$\mathbb{R}_0^2/SO(2) \cong \mathbb{R}_+ \times S^1/SO(2) \cong \mathbb{R}_+$$
(4.57)

suggesting following commutative diagram.

This diagram appears very similar to what we found here in the case of the space of state vectors in a composite Hilbert space

$$S(\mathcal{H}) = S(\mathbb{C}^2 \otimes \mathbb{C}^2) = \bigcup_{\lambda} \Gamma_{\lambda}$$
(4.58)

being stratified into local unitary orbits of *entanglement surfaces*  $\Gamma_{\lambda}$ . The corresponding *entanglement function* relating to each orbit a constant, but distinguished value has been provided here by an inner product

$$f(\psi_{\lambda}) := \left\langle T_{SU(2)\times SU(2)}^{\otimes k} \middle| T_{SU(2)\times SU(2)}^{\otimes kn} \right\rangle$$
(4.59)

on higher order tensor products of invariant symmetric and anti-symmetric pullback tensor fields

$$T_{SU(2)\times SU(2)} := \begin{cases} \eta_{SU(2)\times SU(2)} \\ \omega_{SU(2)\times SU(2)}. \end{cases}$$
(4.60)

Any entanglement monotone on the set of Schmidt coefficients

$$\epsilon : \triangle_1 \to [0, 1]$$
  
$$\lambda \mapsto \epsilon(\lambda) \tag{4.61}$$

may therefore be (re-) constructed from such a function if

$$f\left(\Gamma_{\psi_{\lambda}}\right) =_{!} \epsilon(\lambda), \qquad (4.62)$$

that is, if the following diagram commutes

In the previous section we have shown that there exits a family of entanglement monotones where this diagram commutes. Interestingly, this family has not been discussed in the literature so far and provides therefore a new class of entanglement monotones.

At this point we outline possible generalizations to higher dimensions and mixed states. Indeed, it may become clear that the inner product of any tensor product on the symmetric part

$$\left\langle \eta_{SU(n)\times SU(n)}^{\otimes k}(\rho_{\psi}) \middle| \eta_{SU(n)\times SU(n)}^{\otimes k}(\rho_{\psi}) \right\rangle$$
 (4.63)

should in principle provide an entanglement monotone candidate for all finite dimensions  $n \in \mathbb{N}$  as being invariant under  $SU(n) \times SU(n)$ . This invariance will be discussed from a constructive point of view in more detail in the next section.

A generalization to convex combinations of pure states to the regime of mixed states finally, may be given by corresponding extensions to convex combination pure state entanglement monotones (see e.g. [42, 43] and references therein)

$$\mathcal{E}(\rho) := \sum_{\inf \rho = \sum p_j \rho_j} p_j \epsilon(\rho_j)$$
(4.64)

$$\equiv_{!} \sum_{\text{inf } \rho = \sum p_{j} \rho_{j}} p_{j} \left\langle \eta_{SU(n) \times SU(n)}^{\otimes k}(\rho_{j}) \left| \eta_{SU(n) \times SU(n)}^{\otimes k}(\rho_{j}) \right\rangle.$$
(4.65)

Actually, it appears not a trivial task to perform the computation here due to the infimum in the sum at the first glance. However, an exception where it is known how to compute the generalization from a pure to a mixed state entanglement measure is given by the *concurrence* [75, 76]. A digression on a link to the concurrence measure in geometrical terms will be discussed in the following section.

# 4.5 Invariant operator valued tensor fields (IOVTs) and mixed states entanglement

So far we outlined an entanglement characterization *algorithm* based on invariant tensor fields on the Lie group  $\mathcal{G} = U(n) \times U(n)$ , which 'replaces' functions

$$\epsilon : \triangle_{n-1} \to [0,1]$$
  
$$\lambda \mapsto \epsilon(\lambda) \tag{4.66}$$

on Schmidt-coefficients by functions on pullback tensor-coefficients:

$$(R(X_j)) \xrightarrow{\rho_{\psi}} f(\psi) := \sum_{j_1 j_2} |\kappa_{j_1 j_2}^{\rho_{\psi}}|^2$$

Similar to the case of pure states, we shall also identify within the generalized regime of mixed states entanglement monotone candidates by functions

$$f: D(\mathcal{H}_A \otimes \mathcal{H}_B) \to \mathbb{R}_+ \tag{4.67}$$

which are invariant under the local unitary group of transformations  $U(\mathcal{H}_A) \times U(\mathcal{H}_B)$  [77]. In this necessary strength, we propose in the following entanglement monotones candidates by taking into account constant functions on local unitary orbits of entangled quantum states, arising from *invariant oper*ator valued tensor fields (IOVTs) on  $U(\mathcal{H}_A) \times U(\mathcal{H}_B)$  as considered recently on general matrix Lie groups  $\mathcal{G}$  [4]. Let us review the basic construction.

#### 4.5.1 The basic construction

Given a unitary representation

$$U: \mathcal{G} \to U(\mathcal{H}), \tag{4.68}$$

we may identify an anti-Hermitian operator-valued left-invariant 1-form

$$-U(g)^{-1}dU(g) \equiv iR(X_j)\theta^j$$
(4.69)

on  $\mathcal{G}$ , where the operator  $iR(X_j)$  is associated with the representation of the Lie algebra Lie( $\mathcal{G}$ ). In this way, we may construct higher order invariant operator valued tensor fields

$$-U(g)^{-1}dU(g) \otimes U(g)^{-1}dU(g) = R(X_j)R(X_k)\theta^j \otimes \theta^j, \qquad (4.70)$$

on  $\mathcal{G}$  by taking into account the representation as being equivalently defined by means of the representation of the enveloping algebra of the Lie algebra in the operator algebra  $\mathcal{A} := \text{End}(\mathcal{H})$ . More specific, any element  $X_j \otimes X_k$  in the enveloping algebra becomes associated with a product

$$R(X_i)R(X_k) \in \mathcal{A} := \text{End}(\mathcal{H}), \tag{4.71}$$

where  $\mathcal{A}$ , may denote the vector space of a  $C^*$ -algebra. At this point, we may evaluate each one of these products by means of dual elements

$$\rho \in \mathcal{A}^*, \tag{4.72}$$

according to

$$\rho(R(X_j)R(X_k)) \equiv \operatorname{Tr}(\rho R(X_j)R(X_k)) \in \mathbb{C},$$
(4.73)

yielding a quantum state dependent tensor field

$$\rho(R(X_j)R(X_k))\theta^j \otimes \theta^k \tag{4.74}$$

on the group manifold. By taking the k-th product of invariant operatorvalued left-invariant 1-forms

$$-U(g)^{-1}dU(g) \otimes U(g)^{-1}dU(g) \otimes \dots \otimes U(g)^{-1}dU(g), \qquad (4.75)$$

we shall find a representation R-dependent IVOT of order k

$$\theta_R := \left(\prod_{a=1}^k R(X_{i_a})\right) \bigotimes_{a=1}^k \theta^{i_a} \tag{4.76}$$

on a Lie group  $\mathcal{G} = U(n) \times U(n)$ . After evaluating it with a *mixed* quantum state

$$\theta_R \mapsto \rho(\theta_R) := \theta_R^{\rho} = \rho\left(\prod_{a=1}^k R(X_{i_a})\right) \bigotimes_{a=1}^k \theta^{i_a}$$

one may identify  $\mathcal{G}$ -invariant functions via an inner product on tensor fields

$$f^R(\rho) := \langle \theta_R^{\rho} | \theta_R^{\rho} \rangle \tag{4.77}$$

as constructed in section 4.2.

#### 4.5.2 Purity, concurrence and covariance measures

In particular, for k = n = 2 [4], we recover in this way the purity and the concurrence related measures involving a spin-flip transformed state  $\tilde{\rho}$ by considering inner product combinations of symmetric and anti-symmetric tensor fields

$$\eta_R^{\rho} := \rho([R(X_j), R(X_k)]_+) \theta^j \odot \theta^k$$
(4.78)

$$\omega_R^{\rho} := \rho([R(X_j), R(X_k)]_{-})\theta^j \wedge \theta^k, \qquad (4.79)$$

according to

$$\frac{1}{8} \left( \left\langle \eta_R^{\rho} \mid \eta_R^{\rho} \right\rangle + (-1)^s \left\langle \omega_R^{\rho} \mid \omega_R^{\rho} \right\rangle \right) - \frac{1}{2} = \begin{cases} \operatorname{Tr}(\rho^2) & \text{for } s = 0\\ \operatorname{Tr}(\rho\tilde{\rho}) & \text{for } s = 1. \end{cases}$$
(4.80)

In more general terms, one may introduce R-classes of entanglement monotone candidates by taking into account polynomials

$$f_k^R(\rho) := \sum_n a_n \left\langle \theta_R^{\rho} \mid \theta_R^{\rho} \right\rangle^n, \quad \theta_R^{\rho} := \rho \left(\prod_{a=1}^k R(X_{i_a})\right) \bigotimes_{a=1}^k \theta^{i_a}.$$

The case

$$\tilde{R}(X_j) = R(X_j) - \rho(R(X_j))\mathbb{1}, \qquad (4.81)$$

recovers for IOVTs of order k = 2, a class of separability criteria associated with covariance matrices (CMs) (Gittsovich et al. 2008) by means of a *CM*tensor field

$$\theta_{\bar{R}}^{\rho} = (\rho(R(X_j)R(X_k)) - \rho(R(X_j))\rho(R(X_k))\theta^j \otimes \theta^k.$$
(4.82)

An open problem in the field of CM-ctiteria is provided by the question how to find an extension to quantitative statements [78]. A possible approach could be provided here by taking into account a  $\tilde{R}$ -class of entanglement monotone-candidates by considering

$$f_2^{\tilde{R}}(\rho) = \sum_n a_n \left\langle \theta_{\tilde{R}}^{\rho} \left| \theta_{\tilde{R}}^{\rho} \right\rangle^n. \right.$$

To give an example, we consider the function

$$f_2^{\tilde{R}}(\rho) \equiv \left\langle \theta_{\tilde{R}}^{\rho} \middle| \theta_{\tilde{R}}^{\rho} \right\rangle \tag{4.83}$$

applied to a family of 2-parameter states on a composite Hilbert space of two qubits given by

$$\rho_{x,\alpha_0} := x \left| \alpha_0 \right\rangle \left\langle \alpha_0 \right| + (1-x)\rho^*, \quad \left| \alpha_0 \right\rangle := \cos(\alpha_0) \left| 11 \right\rangle + \sin(\alpha_0) \left| 00 \right\rangle \quad (4.84)$$

and find a possible approximation to the concurrence measure

$$\max[\lambda_4 - \lambda_3 - \lambda_2 - \lambda_1, 0], \quad \lambda_j \in \operatorname{Spec}(\rho \widetilde{\rho}).$$
(4.85)

Both functions are plotted in figure 2.



Figure 2: The function  $f_2^{\tilde{R}}(\rho)$  gives rise to a possible approximation (left) to the concurrence measure (right) applied to a family of 2-parameter states on a composite Hilbert space of two qubits.

# 5 Entanglement Estimation

In the previous section we formulated an algorithm for a tensorial characterization of entanglement both from a qualitative and quantitative point of view. This algorithm had been modeled by a pairing between a quantum state  $\rho$  and a finite set of observables  $\{R(X_j)\}_{j\in J}$  giving rise to a function on 'classical' tensor coefficients

$$(R(X_j)) \xrightarrow{\rho} f(\rho) := \sum_{j_1 j_2} |\kappa_{j_1 j_2}^{\rho}|^2$$

on a given Lie group  $\mathcal{G}$  associated with a unitary representation. Interestingly, the tensor coefficients

$$\kappa_{j_1 j_2}^{\rho} = Cov_{\rho}(R(X_{j_1})R(X_{j_2}))$$
(5.1)

provide covariances and therefore empirically accessible quantities. With other words, these type of entanglement monotones, as suggested in particular by the geometric formulation of quantum mechanics here, may in principle be reconstructed empirically from an *incomplete* set of observables, without the need of a complete tomographic reconstruction of the quantum state.

In contrast, designing experimental settings able to achieve a quantitative description of entanglement is a current challenge, even with the most modern quantum experimental devices available today [8]. One fundamental reason may be related to the fact that standard entanglement monotones are in general non-linear functions on the convex body of quantum states [47, 77], rather than ordinary expectation values of 'quantum observables' associated with Hermitian operators in a Hilbert space. As indicated above, a possible approach to this problem could be provided, on the one hand, by a tomographic reconstruction of the quantum state involving a series of measurements associated with a complete set of observables (see, e.g., [79] for a review on quantum tomography). On the other hand, while a quantum state is supposed to contain the complete information about a physical system, a sufficient amount of information about entanglement may be extracted by means of an *incomplete* set of observables as in the case of the here considered pullback tensor coefficients using a reducible Lie algebra representation associated to  $SU(n) \times SU(n)$  rather than the irreducible representation of  $SU(n^2).$ 

Parallel to our approach we shall mention at this point similar remarks in the literature on standard entanglement measures like on the *concurrence* (4.85) [75, 76], which may be experimentally recovered from four — rather than fifteen — parameters associated with the Bloch representation of a bipartite qubit density matrix [80]. Moreover, an optimal experimental setting for quantifying pure state entanglement – based on the purity – which requires the reconstruction of only three parameters associated with local quantum observables, has been proposed in [81]. Actually, there exists a general framework on the optimization bounds of quantum experiments based on quantum state estimation theory [47, 82, 83] which originated in the seminal work of Helstrom in the 1960s [68]. In contrast, the specialization of this general framework to what could be called quantum entanglement estimation theory is a relatively young field of research (see [8] and references therein). Recent works have focused on the optimization bounds in terms of quantum information measures, like the Kullback mutual information [84], the fidelity [85] – and finally – the quantum Fisher information [8].

Crucially, in the simplest non-trivial case of a 1-parameter family of bipartite pure qubit states in the Schmidt decomposition, several standard entanglement measure fail to provide an efficient quantum Fisher information estimations – in particular – for *weakly* entangled states [8].

In this section, we propose an efficient estimation in the regime of weak entanglement by taking into account the family of alternative entanglement monotones as constructed in the previous section 4. This will allow us to merge the idea of a covariant tensorial characterization of entanglement with the contravariant tensorial identification of geometric inequalities of section 3 into one unified geometric framework.

This section is organized as follows. In subsection 5.1 we review the basic idea of quantum state estimation by using the *Cramèr-Rao inequality* as derived in section 3.2.4. Thereafter, in sections 5.1.1 and 5.1.2, we consider in particular the Schmidt coefficient estimation and the entanglement estimation of two entangled qubits, as recently discussed in [8]. Next, in subsection 5.1.3, we focus on the corresponding estimation of the purity. In subsection 5.2 we will address the estimation of the entanglement and alternative purity measures as constructed in the previous section. We give our conclusions and an outlook in section 5.3.

## 5.1 Quantum state estimation

The most elementary quantum estimation problem may be formulated as follows.

• Given an unknown quantum state vector  $|\psi_{\lambda}\rangle$  on a curve of state vectors parametrized by  $\lambda \in [0, 1]$ . Can we extract the parameter  $\lambda$  from  $|\psi_{\lambda}\rangle$ at small measurement cost?

To tackle this question one defines to any *finite* number of measurements, let's say a sample  $\{x_j\}$ , an *estimator* of  $\lambda$  by a map

$$\{x_j\} \mapsto \hat{\lambda}(\{x_j\}) \in [0, 1].$$
 (5.2)

The efficiency of the estimator is then given by the mean square error

$$\mathbb{E}_{\lambda}((\hat{\lambda}(\{x_j\}) - \lambda)^2). \tag{5.3}$$

The error coincides for *unbiased estimators* 

$$\mathbb{E}_{\lambda}(\hat{\lambda}) = \lambda \tag{5.4}$$

with the variance

$$\operatorname{Var}(\hat{\lambda}) = \mathbb{E}_{\lambda}(\hat{\lambda}^2) - \mathbb{E}_{\lambda}(\hat{\lambda})^2.$$
(5.5)

The efficiency of all unbiased estimators is then bounded by the *quantum-Cramér-Rao inequality* 

$$\operatorname{Var}(\hat{\lambda}) \ge \frac{1}{\kappa_{\lambda}},\tag{5.6}$$

where  $\kappa_{\lambda}$  denotes the coefficient of the pullback of the Fubini Study metric on a given curve of Hilbert space vectors  $|\psi_{\lambda}\rangle$  as seen from the Hilbert space. This inequality has been derived in (3.81) according to section 3.2.4 and closes therefore a circle from the geometric formulation of quantum mechanics to the concept of quantum Fisher information being identified here with the Fubini Study metric related pullback coefficient  $\kappa_{\lambda}$ . The physical interpretation of (5.6) becomes directly available when taking into account the *relative* error [47] depending on

- the number  $M \in \mathbb{N}$  of measurements and
- the signal to noise ratio  $\lambda^2/\text{Var}(\hat{\lambda})$

The relative error is defined by

$$\delta^2 := \frac{\operatorname{Var}(\hat{\lambda})}{M\lambda^2} \equiv \frac{1}{M} \frac{\operatorname{Noise}}{\operatorname{Signal}}.$$
(5.7)

The quantum Cramer-Rao inequality (5.6) implies thus a  $\kappa_{\lambda}$ -bounded *mini-mum* number of measurements

$$M_{\delta} \ge \frac{1}{\lambda^2 \delta^2} \frac{1}{\kappa_{\lambda}} \tag{5.8}$$

required for achieving an estimation of  $\lambda$  with a given fixed relative error  $\delta^2$ .

## 5.1.1 Schmidt coefficient estimation

The fundamental example we'll focus on in the following is the case where  $\lambda$  coincides with a Schmidt coefficient parametrizing a curve of entangled two qubit state vectors

$$|\psi_{\lambda}\rangle = \sqrt{\lambda} |00\rangle + \sqrt{1-\lambda} |11\rangle, \quad \lambda \in [0,1]$$
 (5.9)

in a composite Hilbert space  $\mathcal{H} \cong \mathbb{C}^2 \otimes \mathbb{C}^2$ . For this purpose we compute the pullback of the Fubini-Study metric

$$\left\langle \partial_{\lambda}\psi_{\lambda} \left| \partial_{\lambda}\psi_{\lambda} \right\rangle d\lambda \otimes d\lambda - \left\langle \psi_{\lambda} \left| \partial_{\lambda}\psi_{\lambda} \right\rangle^{2} d\lambda \otimes d\lambda \equiv \kappa_{\lambda}d\lambda \otimes d\lambda \right.$$
(5.10)

on  $\{|\psi_{\lambda}\rangle\}_{\lambda\in[0,1]}$  as seen from the Hilbert space. The coefficient yields in this case the quantum Fisher information

$$\kappa_{\lambda} = \frac{1}{\lambda - \lambda^2} \tag{5.11}$$

As one can see, the *minimum* number of measurements (5.8)

$$M_{\delta}(\lambda) = \frac{1}{\lambda^2 \delta^2} \frac{1}{\kappa_{\lambda}} = \frac{\lambda - \lambda^2}{\lambda^2 \delta^2} \sim \frac{1}{\lambda \delta^2}$$
(5.12)

diverges for  $\lambda \to 0$ . With other words, the estimation becomes inefficient for all states close to the separable state vector  $|11\rangle$ .

#### 5.1.2 Linear entropy estimation

At this point we may note that any entanglement measure on  $|\psi_\lambda\rangle$  relates to a measure

$$\epsilon : [0,1] \to [0,1], \quad \lambda \mapsto \epsilon(\lambda). \tag{5.13}$$

on the parameter space of Schmidt coefficients. Such a measure induces a non-linear parameter transformation on the pullback coefficient

$$\kappa_{\epsilon} \equiv \kappa_{\lambda(\epsilon)} (\partial_{\epsilon} \lambda(\epsilon))^2 \tag{5.14}$$

provided by the covariant transformation property of the pullback tensor

$$\kappa_{\lambda} d\lambda \otimes d\lambda = \kappa_{\lambda(\epsilon)} d\lambda(\epsilon) \otimes d\lambda(\epsilon)$$
(5.15)

$$= \kappa_{\lambda(\epsilon)} \partial_{\epsilon} \lambda(\epsilon) d\epsilon \otimes \partial_{\epsilon} \lambda(\epsilon) d\epsilon \qquad (5.16)$$

$$= \kappa_{\lambda(\epsilon)} (\partial_{\epsilon} \lambda(\epsilon))^2 d\epsilon \otimes d\epsilon \tag{5.17}$$

$$\equiv \kappa_{\epsilon} d\epsilon \otimes d\epsilon. \tag{5.18}$$

As a consequence, one finds a *distinguished* minimum number of measurements

$$M_{\delta}(\epsilon) = \frac{1}{\epsilon^2 \delta^2} \frac{1}{\kappa_{\epsilon}}$$
(5.19)

required for achieving an estimation of  $\epsilon(\lambda)$  rather than an estimation of  $\lambda$ . Indeed, similar to the Schmidt coefficient estimation, it implies again a divergence, in particular for vanishing values of  $\epsilon$ 

$$\lim_{\epsilon \to 0} M_{\delta}(\epsilon) \to \infty \tag{5.20}$$

whenever one identifies the measure  $\epsilon$  with the *linear entropy* 

$$\epsilon(\lambda) := 2\left(1 - \operatorname{Tr}\left((\rho_{\lambda}^{A})^{2}\right)\right) = 4\lambda(1-\lambda)$$
(5.21)

or the *negativity*  $\epsilon_N(\lambda) := \sqrt{\epsilon(\lambda)}$ . In conclusion, the estimation of standard entanglement measures related to the linear approximation of the von Neumann entropy becomes *inefficient* in the regime of *weak* entanglement [8].

#### 5.1.3 Purity estimation

At this point we may ask for the estimation efficiency of the purity

$$\operatorname{Tr}\left((\rho_{\lambda}^{A})^{2}\right)$$
 (5.22)

defining in contrast to the linear entropy

$$2(1 - \operatorname{Tr}((\rho_{\lambda}^{A})^{2}))$$

an entanglement *anti-monoton* admitting non-vanishing values close to one in the regime of weak entanglement. The purity measure (5.22) for the reduced density state

$$\rho_{\lambda}^{A} = \left(\begin{array}{cc} \lambda & 0\\ 0 & 1-\lambda \end{array}\right) \tag{5.23}$$

associated with the family of entangled state vectors (5.9) reads

$$\epsilon(\lambda) := \operatorname{Tr}((\rho_{\lambda}^{A})^{2}) = \lambda^{2} + (1-\lambda)^{2} = 1 - 2\lambda + 2\lambda^{2}.$$
 (5.24)

To compute the induced parameter transformation on the quantum Fisher information (5.14), we need to identify the inverse function solutions of the purity

$$\lambda(\epsilon) := \frac{1}{2} \left( 1 \pm \sqrt{2\epsilon - 1} \right). \tag{5.25}$$

Both solutions yield according to (5.14) the parameter-transformed quantum Fisher information

$$\kappa_{\epsilon} = -\frac{1}{4\epsilon^2 - 6\epsilon + 2}.\tag{5.26}$$

However, this implies a *negative* number of measurements

$$M_{\delta \equiv 1}(\epsilon) = -4 + \frac{6}{\epsilon} - \frac{2}{\epsilon^2}$$
(5.27)

within the estimation (5.19), appearing beyond a physical interpretation. Hence, the purity does not solve the problem of estimating weakly entangled qubits in an efficient way.

## 5.2 Estimation of inner products on tensor fields

A class of entanglement monotones may be constructed in terms of functionals on the linearization of the von Neumann entropy

$$S_{vN}(\lambda) = \operatorname{Tr}(\rho_{\lambda}^{A} \ln \rho_{\lambda}^{A}) = \sum_{j} \lambda_{j} \ln \lambda_{j}$$

$$\downarrow$$

$$S_{vN}^{linear}(\lambda) := 1 - \operatorname{Tr}((\rho_{\lambda}^{A})^{2}) = 1 - \sum_{j} \lambda_{j}^{2}$$

$$\downarrow$$

$$f\left(S_{vN}^{linear}(\lambda)\right) = \begin{cases} 2S_{vN}^{linear}(\lambda) \\ \sqrt{2S_{vN}^{linear}(\lambda)} \\ 1 - S_{vN}^{linear}(\lambda) \end{cases}$$
(5.28)

providing the linear entropy, the negativity and the purity as (anti-) monotones on Schmidt coefficients respectively. This class of entanglement monotones constructed in this way highlights a clear advantage compared to the von Neumann entropy as it does not involve the the diagonalization of the reduced density state (or equivalently, the singular value decomposition into Schmidt coefficients of the state vector associated to the product Hilbert space). However, as we have seen in the previous section, such monotones may not become accessible in experiments as soon we enter into the regime of weak entanglement.

At this point we observe the following. The linear entropy is directly related to the pullback coefficient  $\kappa_{\lambda}$  of the Fubini metric by

$$S_{vN}^{linear}(\lambda) = \frac{2}{\kappa_{\lambda}} \tag{5.29}$$

due to (5.11) in

$$2(1 - \text{Tr}((\rho_{\lambda}^{A})^{2}) = 4\lambda(1 - \lambda) =_{(5.11)} \frac{4}{\kappa_{\lambda}}.$$
 (5.30)

The quantum Cramér Rao inequality (5.6) may thus be rewritten for a family of two entangled qubits as entropy inequality

$$Var(\lambda) \ge \frac{1}{2} S_{vN}^{linear}(\lambda)$$
 (5.31)

and may therefore be linked to the class of quantum entropy inequalities as considered in section 3.1. Actually, the relation (5.29) between linear entropy and the pullback coefficient of the Fubini Study metric highlights a link to the beginning of section 1.4 considering any quantum Fisher information metric as the Hessian of some quantum relative entropy.

Our 'inverse trip' from geometric quantum mechanics to quantum information may therefore come to an end at this point. Actually, one of the most important implications of geometric quantum mechanics for quantum information theory has still to be discussed. Indeed,  $\kappa_{\lambda}$  is the pullback coefficient of the Fubini Study metric on a particular chosen 1-dimensional entangled state submanifold. In the following we ask: What happens if we estimate the entanglement monotones provided by the inner products

$$\epsilon_n := \frac{1}{12^n} \left\langle \eta_{SU(2) \times SU(2)}^{\otimes n} \middle| \eta_{SU(2) \times SU(2)}^{\otimes n} \right\rangle$$
(5.32)

$$\mu_n := \frac{(-1)^n}{4^n} \left\langle \omega_{SU(2) \times SU(2)}^{\otimes n} \middle| \omega_{SU(2) \times SU(2)}^{\otimes n} \right\rangle$$
(5.33)

on the tensor products of the symmetric and anti-symmetric part of the pullback tensor fields

$$\kappa_{SU(2)\times SU(2)} = \eta_{SU(2)\times SU(2)} + i\omega_{SU(2)\times SU(2)} \tag{5.34}$$

related to the orbits of the symmetry group of entanglement  $SU(2) \times SU(2)$ constructed in section 4.2?

For this purpose we apply the procedure of section 5.1.2 by first identifying the inverse functions  $\lambda(\epsilon_n)$  and  $\lambda(\mu_n)$  of the monotones (4.51) and (4.52),

$$\frac{1}{12^n} \operatorname{Tr}(\eta^2)^n = \left(\frac{16\lambda^4}{3} - \frac{32\lambda^3}{3} + 4\lambda^2 + \frac{4\lambda}{3} + \frac{1}{3}\right)^n$$
(5.35)

$$\frac{(-1)^n}{4^n} \operatorname{Tr}(\omega^2)^n = \left(4\lambda^2 - 4\lambda + 1\right)^n \tag{5.36}$$

as solutions of the equations

$$\left(\frac{16\lambda^4}{3} - \frac{32\lambda^3}{3} + 4\lambda^2 + \frac{4\lambda}{3} + \frac{1}{3}\right)^n - \epsilon_n(\lambda) = 0$$
 (5.37)

$$\left(4\lambda^2 - 4\lambda + 1\right)^n - \mu_n(\lambda) = 0 \tag{5.38}$$

associated with the inner products of the tensor products of the symmetric tensor fields and the tensor product of the antisymmetric tensor fields respectively. As a result we find both real and imaginary valued solutions. To provide a physical interpretation we consider the real-valued solutions and define the parameter transformation (5.14)

$$\kappa_{\epsilon_n} := \kappa_{\lambda(\epsilon_n)} (\partial_{\epsilon_n} \lambda(\epsilon_n))^2 \tag{5.39}$$

of the quantum Fisher information on the 1-parameter family of Schmidt coefficient decomposed quantum states according to formula (5.14). In this way we find the minumum number  $M_{\delta}(\epsilon_n)$  of measurements

$$M_{\delta}(\epsilon_n) = \frac{1}{\epsilon_n^2 \delta^2} \frac{1}{\kappa_{\epsilon_n}} \tag{5.40}$$

as defined in (5.19) for achieving an estimation with fixed relative error  $\delta$ . The result is illustrated with  $\delta \equiv 1$  for the first five powers in figure 3.

#### 5.2.1 Discussion for monotones from the symmetric part

Let us begin to analyze the estimation of the entanglement monotones associated with the inner products on the tensor products of the symmetric tensor



Figure 3: estimation of entanglement monotones  $\epsilon_n$  and purity monotones  $\mu_n$ for the first five orders (each order corresponds to the same color as used in the plot in figure 1). It shows the number  $M_{\delta}$  of measurements in dependence of the value  $\epsilon_n$  (and  $\mu_n$  respectively) of the monotones required for achieving an estimation in a 99,9% confidence interval with fixed relative error  $\delta \equiv 1$ . The dashed curves in the first plot correspond to the estimation of the linear entropy, and in the second plot to the negative-valued estimation of the purity as done in (5.27) when 'reflected' on the  $\mu$ -axis.

field in the first plot of figure 3. While the estimation of the linear entropy (dashed curve) diverges for weakly entangled states in the limit  $\epsilon_n \rightarrow 0$  (see section 5.1.2 and [8]), we find for the tensorial monotones an approximative improvement with a finite number of measurements towards the regime of weak entanglement in dependence of the tensor field order n. The inflection point into negative values indicates the boundary of the regime where the approximation looses its validity. The validity of the approximation into the regime of weakly entanglement may become enlarged by considering inner products on tensor fields of higher order. The green curve in the first plot of figure 3 corresponds to the highest order example n = 5 and clearly illustrates this enlargement when compared to the lower tensor field orders.

#### 5.2.2 Discussion for monotones from the anti-symmetric part

The estimation of the purity monotones associated with inner products on the tensor products of the anti-symmetric tensor fields is illustrated in the second plot of figure 3. All curves clearly show here an efficient estimation for weakly entangled states, i.e. for all states close to  $\mu_n = 1$ . The dashed curve corresponds here to the estimation of the standard purity (section 5.1.3) when reflected on the  $\mu$ -axis into positive values. Indeed, only the curves associated with the inner products on the tensor fields may admit a physical interpretation.

## 5.3 Conclusions and outlook

In the geometric formulation of quantum mechanics one considers the Fubini-Study metric at the first place. Any action of the symmetry group of entanglement on a family of entangled quantum state vectors induces a family of degenerate pull back tensor fields each defined as pull back of the Fubini-Study metric from the Hilbert space to the Lie group  $SU(2) \times SU(2)$ . Along the decomposition of the Fubini-Study metric into a Riemannian and a symplectic tensor field one finds a decomposition

$$\kappa_{SU(2)\times SU(2)} = \eta_{SU(2)\times SU(2)} + i\omega_{SU(2)\times SU(2)}$$

into degenerate symmetric and anti-symmetric pullback structures. Via an inner product on higher order tensor fields it is possible to identify two classes of monotonic functions characterizing the entanglement and purity of a bipartite quantum system. These geometrically constructed classes of entanglement and purity monotones provide advantages in the estimation of entangled qubits when compared to standard entanglement and purity monotones. The basic picture emerging here may be subsumed as follows. While the inner product

$$\left\langle \eta_{SU(2)\times SU(2)}^{\otimes n} \left| \eta_{SU(2)\times SU(2)}^{\otimes n} \right\rangle \right.$$

yields an *approximative* efficient entanglement estimation for *all* state vectors, one finds for the inner product

$$\left\langle \omega_{SU(2)\times SU(2)}^{\otimes n} \left| \omega_{SU(2)\times SU(2)}^{\otimes n} \right\rangle \right.$$

an *exact* efficient purity estimation for *weakly* entangled state vectors.

It would be interesting to investigate whether this approach admits also advantages in the entanglement estimation of more general composite quantum systems involving multi-partite systems, mixed quantum states and infinite dimensional Hilbert spaces. As a matter of fact, such a generalization becomes directly testable by the tensor field-valued pairing

 $\downarrow$ 

$$\{\rho_{\vec{\lambda}}\}_{\vec{\lambda}\in\mathcal{M}}\times\{U^{-1}(g)dU(g)^{\otimes k}\}_{g\in\mathcal{G}}$$
(5.41)

$$\rho_{\vec{\lambda}}\left(U^{-1}(g)dU(g)^{\otimes k}\right) \equiv \kappa_{\mathcal{G}}(\vec{\lambda})$$
(5.43)

as described here in section 4.5 between general manifolds  $\{\rho_{\vec{\lambda}}\}_{\vec{\lambda}\in\mathcal{M}}$  of quantum states and *invariant operator valued tensor fields*  $U^{-1}(g)dU(g)^{\otimes k}$  on general Lie groups  $\mathcal{G}$  associated with a unitary representation  $U: \mathcal{G} \to \mathcal{U}(\mathcal{H})$ . A generalization in several directions could therefore be tackled by focusing on the corresponding inner products on tensor fields

$$\left\langle \kappa_{\mathcal{G}}^{\otimes n}(\vec{\lambda}) \middle| \kappa_{\mathcal{G}}^{\otimes n}(\vec{\lambda}) \right\rangle.$$

A deeper understanding on the relation with the quantum Fisher information and all its possible generalized variants (see [21] and references therein) finally, may come along here in terms of a geometrization of the  $C^*$ -algebraic approach of quantum mechanics [41,61] (see appendix B) including star products of quantum tomograms [79] which may close a circle to the empirical bounds on precision of quantum measurements in terms of generalized uncertainty relations [86].

Indeed, some fundamental aspects in this direction have been found in section 3 by providing a general framework for the translation of any given quantum statistical measure

$$S: D(\mathcal{H}) \to \mathbb{R}_+,\tag{5.44}$$

including the von Neumann entropy, in terms of a functional

$$\tilde{S}: \mathcal{F}_K(\Gamma) \to \mathbb{R}_+$$
 (5.45)

on the  $C^*$ -algebra of Kähler functions

$$f_A(g) := \frac{\langle g | A | g \rangle}{\langle g | g \rangle} = \frac{\langle \psi_0 | U(g)^{\dagger} A U(g) | \psi_0 \rangle}{\langle \psi_0 | \psi_0 \rangle}$$
(5.46)

on a unitary orbit  $\Gamma \subset \mathcal{H}_0$ . These Kähler functions are deeply connected to Husimi functions

$$f_{\rho}(g) = \frac{\langle g | \rho | g \rangle}{\langle g | g \rangle} = \frac{\langle \psi_0 | U(g)^{\dagger} \rho U(g) | \psi_0 \rangle}{\langle \psi_0 | \psi_0 \rangle}$$
$$= \operatorname{Tr} \left( \rho U(g) \frac{|\psi_0\rangle \langle \psi_0|}{\langle \psi_0 | \psi_0 \rangle} U(g)^{\dagger} \right) \equiv \operatorname{Tr} (\rho U(g) E_0 U(g)^{\dagger})$$
(5.47)

- and therefore to a special subclass of quantum tomograms [79] – and Wehrl type entropy inequalities as shown in section 3.1. A possible foundation for generalized uncertainty relations finally, may become available in terms of geometric inequalities as worked out in section 3.2 for the Robertson Schrödinger inequality and for a version of a quantum Cramér Rao inequality. Actually, the latter has been sufficient for taking into account some of the current developments of quantum estimation theory – in particular – when applied to the empirical bounds of entanglement quantification as considered here in this last section of the underlying work.

At this point we shall remark: The challenge on an efficient empirical quantification of quantum entanglement can be seen 'embedded' according to Pawel Horodecki into a more general set of questions of quantum information theory [80]. Horodecki indicated this class of questions in 2002 as follows:

# "What kind of information can be extracted from an unknown quantum state at small measurement cost?"

At least from the physicist point of view, this is perhaps one of the most elementary question to begin with, when dealing with the notion of quantum information. What we know from basic quantum mechanics is that we can extract Born type probability measures

$$\operatorname{Tr}(\rho E_x) = p(x) \tag{5.48}$$

or more general, quantum tomograms (compare also formula (5.47))

$$\operatorname{Tr}(\rho U_s E_x U_s^{\dagger}) \equiv \mathcal{P}(x_s) \tag{5.49}$$

at 'low cost' as they require only one POVM  $\{E_x\}_{x\in Q}$  related correspondently to only one single observable. This is in contrast to general quantum statistical measures like the purity

$$Tr(\rho\rho) \tag{5.50}$$

or the von Neumann entropy

$$Tr(\rho \ln \rho) \tag{5.51}$$

requiring a tomographic reconstruction of the state  $\rho$  involving a possibly continuous set of POVMs related to a complete set of observables.

The content of this work has been focused on a third class of measures being extractable from quantum states in terms of what we could call *geometric measures* 

$$\operatorname{Tr}(d\rho \otimes d\rho) \tag{5.52}$$

$$\operatorname{Tr}(\rho U^{\dagger} dU^{\otimes k}) \tag{5.53}$$

provided by the geometric formulation of quantum mechanics.

In fact, the geometric formulation of quantum mechanics provides perhaps one of the most appropriated frameworks to tackle Horodecki's question as stated above. To be specific, the Fubini Study metric provides a conceptual unification of the following both concepts, that is

• information of a quantum state, and

• measurement costs

on the one hand by quantifying entanglement – as an extractable information from a quantum state – in terms of

• the pullback on the symmetry group of entanglement,

and on the other hand, by quantifying the minimum of measurement costs with

• the pullback on a 1-parameter family of quantum state vectors.

Of course it may remain an open problem to identify all properties of a general quantum state (and therefore the 'complete' information from an unknown quantum state) at small measurement cost. Geometry indeed, will always provide the most fundamental mathematical language for describing the state attributes of our world – no matter if classical or quantum.

# A Weyl Systems

Consider a symplectic vector space  $(V, \omega)$  with an Abelian non-compact Lie group of translations  $V \cong \mathbb{R}^{2n}$  endowed with a symplectic structure  $\omega$ . Any generating vector field of translations in the associated Lie algebra may therefore be identified with an element  $v \in V$  according to the isomorphism  $T_v V \cong V$ .

We consider an irreducible unitary (projective) representation of V on a Hilbert space  $\mathcal{H}$  defined by a *Weyl-system*, that is, a map  $W: V \to \mathcal{U}(\mathcal{H})$ , satisfying the following conditions [66]:

1. W is strongly continuous as a function on V;

2. 
$$W(v+v') = e^{-\frac{i}{2}\omega(v,v')}W(v)W(v')$$

According to the Stone-von Neumann theorem it is possible to write

$$W(v) = e^{iR(v)},\tag{A.1}$$

in terms of a self-adjoint realization R(v) of a generating vector field of translations v. It implies

$$[R(v), R(v')] = -i\omega(v, v').$$
(A.2)

Hence, all self-adjoint operators R(v) amounted with elements of a Lagrangian subspace of V commute.

In this regard we recall some basic notions of symplectic geometry (see e.g. [87] p. 403) as follows. To any subspace

$$U \subset V, \tag{A.3}$$

there is a  $\omega$ -orthogonal complement

$$U^{\perp} := \{ v \in V | \omega(v, u) = 0, u \in U \}.$$
(A.4)

The subspace U may define either a *isotropic* or a *coisotropic* subspace if  $U \subset U^{\perp}$  or  $U^{\perp} \subset U$  respectively. An isotropic subspace U which induces together with another isotropic subspace U' a splitting

$$V = U \oplus U',\tag{A.5}$$

is called Lagrangian subspace. In turns out that U is Lagrangian iff U is isotropic and has  $\text{Dim}(U) = \frac{1}{2}\text{Dim}(V)$ , which is equivalent with  $U = U^{\perp}$ . A given Lagrangian subspace U and its associated splitting into Lagrangian subspaces

$$v := (u, f) \in U \oplus U' \tag{A.6}$$

induces a decomposition on the Weyl system into

$$W(v) = W(u, f) = e^{-\frac{i}{2}\omega((u,0),(0,f))}U(u)V(f)$$
(A.7)

with

$$U \equiv W|_U : U \to \mathcal{U}(\mathcal{H}) = \mathcal{U}(L^2(U))$$
(A.8)

and

$$V \equiv W|_{U'} : U' \to \mathcal{U}(\mathcal{H}) = \mathcal{U}(L^2(U)).$$
(A.9)

For  $\psi \in L^2(U)$  the Weyl system action reads [66]

$$U(U')\psi(u) = \psi(u+U') \tag{A.10}$$

$$V(f)\psi(u) = e^{-i\omega((u,0),(0,f))}\psi(u).$$
(A.11)

# A.1 Introducing canonical coordinates

## A.1.1 Real coordinates

We have seen that given a symplectic structure  $\omega$  on V, we may consider a splitting into Lagrangian subspaces. In this way we may introduce canonical

coordinates

$$v = (q, p) \in \mathbb{R}^n \oplus (\mathbb{R}^n)^* \tag{A.12}$$

according to the splitting associated to the symplectic structure

$$\omega = \delta_{jk} dp^j \wedge dq^k. \tag{A.13}$$

Actually, a vector space of translations V may admit different symplectic structures and associated splittings into Lagrangian subspaces in dependence of the physical and experimental setting. For instance, we may deal either with a composite phase space of n particles in one dimension or with one particle in n dimensions:

$$V \cong \mathbb{R}^{2n} \cong \begin{cases} (T^* \mathbb{R})^n \\ T^* \mathbb{R}^n \end{cases}$$
(A.14)

Correspondently, these cases may endowed with distinguished symplectic structures

$$\omega := \begin{cases} \bigoplus_{j=1}^{n} \omega_j \text{ with } \omega_j = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}. \\ \begin{pmatrix} 0 & \mathbb{1}_n \\ -\mathbb{1}_n & 0 \end{pmatrix}. \end{cases}$$
(A.15)

Let  $\{X_j\}_{j\in J}$  be a basis on  $V \cong \mathbb{R}^{2n}$  (or a basis of generating vector fields in the Lie algebra of V respectively). Such a basis becomes therefore represented within a Weyl system by a set of Hermitian operators  $\{R(X_j)\}_{j\in J}$  on  $\mathcal{H}$  with different position and momentum operator realizations

$$\{R(X_1), R(X_2), ..., R(X_{n+1}), R(X_{n+2}), ..., R(X_{2n})\} := \begin{cases} \{Q_1, P_1, ..., Q_n, P_n\} \\ \{Q_1, Q_2, ..., P_{n-1}, P_n\} \end{cases}$$
(A.16)

in dependence of the symplectic structure.

**Remark A.1.** Both settings may be related in terms of a permutation matrix [59].

In any of the above two splittings in real canonical coordinates, we find a Weyl system

$$W(q,p) = e^{i(q^j P_j - p^j Q_j)}$$
 (A.17)

### A.1.2 Complex coordinates

The latter expression becomes identified with the *displacement operator* when rewritten in complex coordinates

$$W(z) \equiv e^{z^j a_j^{\top} - \bar{z}^j a_j},\tag{A.18}$$

as being used in quantum optics for n bosons in one dimension (see e.g. [59]). The relation to real coordinates is given by taking into the decomposition

$$z^{j} = \frac{1}{\sqrt{2}}(q^{j} + ip^{j})$$
 (A.19)

in real and imaginary coordinates. Indeed, by setting  $\kappa_1 = \frac{1}{\sqrt{2}}$ , both descriptions become equivalent, i.e.

$$W(z) = W(q, p). \tag{A.20}$$

The Weyl system generators  $iR(X_j)$  are related to the basis

$$\{Z_j\}_{j\in J} := \{X_j + iX_{n+j}\}_{j\in J}$$
(A.21)

on  $V \cong \mathbb{C}^n \cong \mathbb{R}^{2n}$  yielding the mode operators

$$R(Z_j) = \kappa_1(Q_j + iP_j) := a_j, \quad R(\bar{Z}_j) = \kappa_1(Q_j - iP_j) := a_j^{\dagger}, \quad (A.22)$$

with  $\kappa_1 \in \mathbb{R}$ .

# A.2 Symplectic transformation

Let us focus on those transformations which leave the underlying symplectic structure of a given Weyl-system invariant. These transformation define the Lie group  $Sp(2n, \mathbb{R})$  of symplectic transformation

$$S: V \to V, \quad \omega(Sv, Sv') = \omega(v, v').$$
 (A.23)

On the level of the Weyl-system we find, that the following diagram

commutes, if we identify  $\Phi_{U_S}$  with an automorphism

$$\Phi_{U_S}(W(v)) = U_S^{-1}W(v)U_S = W(Sv) := W_S(v)$$
(A.25)

with  $U_S \in U(\mathcal{H})$ . This induces a transformation on the infinitesimal level of the Hermitian operator representation given by

$$R_S(v) \equiv R(Sv) = U_S^{-1} R(v) U_S. \tag{A.26}$$

The application of a symplectic transformation on a translation is taken into account in the affine symplectic group

$$ISp(2n,\mathbb{R}) := Sp(2n,\mathbb{R}) \times \mathbb{R}^{2n}.$$
(A.27)

Many quantum dynamical systems, as those being considered in quantum optics and condensate matter, are based on unitary representations of the latter group, defining in this way the so-called metaplectic representations [59]. Such representations may be generated by Hamiltonians

$$H = \underbrace{\sum_{k=1}^{n} g_{k}^{(1)} a_{k}^{\dagger}}_{\text{translations}} + \underbrace{\sum_{k,l=1}^{n} g_{kl}^{(3)} a_{k}^{\dagger} a_{l}^{\dagger}}_{\text{squeezing}} + \underbrace{\sum_{k>l=1}^{n} g_{kl}^{(2)} a_{k}^{\dagger} a_{l}}_{\text{mixing}}$$
(A.28)

that are linear and bilinear in n field modes. They induce translations, squeezing and mixing operations associated to unitary representations of noncompact and compact subgroups of  $ISp(2n, \mathbb{R})$ . In the case n = 2, for instance, we encounter unitary representations of the subgroups

$$\mathbb{R}^4$$
,  $SU(1,1)$  and  $SU(2)$  (A.29)

respectively.

# A.3 The Wigner-Weyl correspondence

Given a Weyl system  $W: V \to \mathcal{U}(\mathcal{H})$  we may define the so-called *character*istic function of a generic operator A on  $\mathcal{H}$  by

$$\chi[A](v) := \operatorname{Tr}(AW(v)). \tag{A.30}$$

Vice versa, to a function in  $L^1(\mathbb{R}^{2n}, d^n \alpha d^n x)$  one may associate an Hilbert-Schmidt operator on  $L^2(\mathbb{R}^n, d^n x)$  as follows [79]: Consider the Fourier transform  $\tilde{f}$  within the inverse Fourier transform formula

$$f(q,p) = \int d^{n}\alpha d^{n}x \tilde{f}(\alpha,x) e^{i(\alpha q - xp)}$$
(A.31)

and replace the 1-dimensional irreducible unitary representation

$$e^{i(\alpha q - xp)} \tag{A.32}$$

of  $V \cong \mathbb{R}^{2n}$  with the *projective* irreducible unitary representation of V, defined by the corresponding Weyl-system

$$W(x,\alpha) := e^{i(\alpha Q - xP)},\tag{A.33}$$

yielding the operator

$$W(f) := \int d^n \alpha \, d^n x \, \tilde{f}(\alpha, x) W(x, \alpha). \tag{A.34}$$

This formula may now be compared with the reconstruction formula [59]

$$A = \int_{V} \frac{d^{2n}v}{\pi^n} \chi[A](v) W^{\dagger}(v)$$
 (A.35)

of a generic operator A from its characteristic function. At this point we may introduce the so-called *Wigner function* of an operator A as the Fourier transform of the characteristic function according to

$$\mathcal{W}[A](q,p) := \int d^n \alpha \, d^n x \, \chi[A](\alpha, x) e^{i(\alpha q - xp)}. \tag{A.36}$$

In conclusion, the Weyl system framework provides a fundamental mathematic tool to relate operators to functions and vice versa. In particular, this allows to consider a Quantization-Dequantization algorithm scheme involving both Weyl systems W and Wigner functions W according to the following table:

	QUANTIZATION	DEQUANTIZATION
'Input'	f	A
'Output'	W(f)	$\mathcal{W}[A]$

The outlined dequantization correspondence should be taken with great care to avoid physical misinterpretations. In particular, the Wigner function of quantum state may also take negative values providing thus not a probability but a quasi-probability distribution on the symplectic space V. Probability

distributions may be defined on a Lagrangian subspace in terms of *quantum* tomograms in terms of a inverse Radon transform of the Wigner function [79]. Vice versa, the Wigner function of a quantum state may be reconstructed from a given a family of quantum tomograms in terms of a Radon transform. Quantum tomograms are therefore of fundamental importance in quantum estimation theory [83].
## **B** Tensor fields from geometrized $C^*$ -algebras

By evaluating a quantum state on invariant operator valued tensor field on a Lie group  $\mathcal{G}$  as done in section 4.5, we may identify the resulting covariant classical tensor field as a pull-back induced by the projection of the Lie group  $\mathcal{G}$  on a homogenous space  $\mathcal{G}/\mathcal{G}_0$  generated by the associated action of  $\mathcal{G}$  in the vector space  $\mathcal{A}$  of a  $C^*$ -algebra. This suggests consider a more general class of tensor fields on  $\mathcal{G}$  by taking into account tensor fields from geometrized  $C^*$ -algebras and their pull-back to the group.

Let us restart for this purpose by considering the definition of a finite dimensional  $C^*$ -algebra at the first place:

**Definition B.1** (Finite dimensional  $C^*$  algebra). A Banach space  $\mathcal{A}$  which is endowed with an associative product

$$\mathcal{A} \times \mathcal{A} \to \mathcal{A} \tag{B.1}$$

$$(A,B) \mapsto A \cdot_K B, \tag{B.2}$$

and an involution operation

$$\mathcal{A} \to \mathcal{A}$$
 (B.3)

$$A \mapsto A^*,$$
 (B.4)

which is compatible with the Banach norm according to

$$||AA^*|| = ||A||^2, \tag{B.5}$$

is called a finite dimensional  $C^*$ -algebra.

Any finite dimensional  $C^*$ -algebra is isomorphic to an algebra  $M_n(\mathbb{C})$  of complex  $n \times n$ -matrices. Here we note that the usual row-by-column matrix product

$$(A,B) \mapsto AB \tag{B.6}$$

is a special example of an associative product. In general we may deal in this regard also with alternative products not necessarily coinciding with the usual matrix product. For instance we may use

$$A \cdot_K B \equiv AKB \tag{B.7}$$

with a fixed matrix  $K \in M_n(\mathbb{C})$ , which satisfies

$$(A \cdot_K B) \cdot_K C = A \cdot_K (B \cdot_K C), \tag{B.8}$$

for all  $A, B, C \in \mathcal{A}$ .

By considering a realification of the the C\*-algebra  $M_n(\mathbb{C})$  from  $\mathcal{A} \cong \mathbb{R}^{n^2} \oplus i\mathbb{R}^{n^2}$  to

$$\mathcal{A}_{\mathbb{R}} \cong \mathbb{R}^{2n^2},\tag{B.9}$$

we may deal with the question whether the later stands in an analogy with a 'classical phase space' manifold, admitting classical tensors in a similar way it has been constructed on the realification of a Hilbert space in [1]. In contrast to a Hilbert space, we have to take new structures into account which come along the definition of a  $C^*$ -algebra. They essentially consist of

- an associative bi-linear product, which is non-commutative and
- a Banach Norm, compatible with the product.

As we will see in the following, both points provide implications for the construction of tensor fields, in particular defined in contra-variant form on the dual vector space  $\mathcal{A}^*$ , resp. in covariant form on the vector space of the  $C^*$ -algebra  $\mathcal{A}$ . Let us start with the construction of contra-variant tensors by following the argumentation line as proposed in [61].

# B.1 Contra-variant tensors on the dual vector space $\mathcal{A}^*$

Consider the dual space

$$\mathcal{A}^* := \operatorname{Lin}(\mathcal{A}) \tag{B.10}$$

of complex valued linear functionals

$$\alpha: \mathcal{A} \to \mathbb{C} \tag{B.11}$$

on  $\mathcal{A}$ . Note that the duality isomorphism is considered here in the category of (finite dimensional) vector spaces and not in the category of abstract  $C^*$ algebras. Here we observe that for all elements  $A \in \mathcal{A}$  there exits an element  $\widehat{A}$  in the bi-dual  $(\mathcal{A}^*)^* := \operatorname{Lin}(\mathcal{A}^*)$ , i.e. the space of linear functionals

$$\widehat{A}: \mathcal{A}^* \to \mathbb{C} \tag{B.12}$$

on the dual space  $\mathcal{A}^*$ , provided by

$$\widehat{A}(\alpha) := \alpha(A) \tag{B.13}$$

for all  $\alpha \in \mathcal{A}^*$ . The relation (B.13) defines an embedding

$$\mathcal{A} \hookrightarrow \operatorname{Lin}(\mathcal{A}^*) \subset \mathcal{F}(\mathcal{A}^*) \tag{B.14}$$

$$A \mapsto \widehat{A},$$
 (B.15)

of  $\mathcal{A}$  into to the bi-dual  $(\mathcal{A}^*)^*$  and will be of fundamental importance for the following discussion. In particular, within the bi-dual space of linear functionals on  $\mathcal{A}^*$  we are allowed to identify a commutative product among linear 'functions' which provides a quadratic function

$$(\widehat{A} \cdot \widehat{B})(\alpha) := \widehat{A}(\alpha) \cdot \widehat{B}(\alpha), \tag{B.16}$$

and therefore caries the usual differential calculus. In particular we will have an exterior derivative on elements of  $\text{Lin}(\mathcal{A}^*)$  where the usual Leibniz-rule is available according to

$$d(\widehat{A} \cdot \widehat{B}) = d\widehat{A} \cdot \widehat{B} + \widehat{A} \cdot d\widehat{B}.$$
(B.17)

In this way we may identify  $d\hat{A}$  as a 1-form on  $\mathcal{A}^*$ , resp. as a section in the global trivialized co-tangent bundle over the dual space according to

$$d\widehat{A}: \mathcal{A}^* \to T^*\mathcal{A}^* = \mathcal{A}^* \times (\mathcal{A}^*)^* \equiv \mathcal{A}^* \times \mathcal{A}.$$
 (B.18)

$$\alpha \mapsto (\alpha, \widehat{A}) \equiv (\alpha, A),$$
 (B.19)

i.e. a section in a vector bundle  $\mathcal{A}^* \times \mathcal{A}$ .

On the other hand we may translate the non-commutative product

$$A \cdot_K B$$
 (B.20)

on  $\mathcal{A}$  into a corresponding non-commutative product on  $\operatorname{Lin}(\mathcal{A}^*)$ , whenever we consider the embedding (B.13), as a way to translate the product between operators to a product between functions according to

$$\alpha(A \cdot_K B) =_{(B.13)} \widehat{A \cdot_K B}(\alpha) := \widehat{A}(\alpha) \star_K \widehat{B}(\alpha).$$
(B.21)

Such a product appears in the literature of non-commutative geometry and provides an instance of a so called *star-product* (see e.g. [88]). It is nonlocal product since it is associated with row-by-column product which is the analog of the *convolution product*<sup>16</sup>. The non-commutative structure of  $\mathcal{A}^*$ is now completely encoded in the product  $\star_K$  between these functions in the bi-dual space  $\operatorname{Lin}(\mathcal{A}^*) \subset \mathcal{F}(\mathcal{A}^*)$ .

<sup>&</sup>lt;sup>16</sup>For instance the usual Moyal product is the convolution product on the Heisenberg-Weyl group.

This essentially allows us now to identify a contra-variant rank-2 tensor field on  $\mathcal{A}^*$  according to the construction

$$\tau_K(d\widehat{A}, d\widehat{B})(\alpha) := \alpha(A \cdot_K B) = \widehat{A}(\alpha) \star_K \widehat{B}(\alpha).$$
(B.22)

Hence, we arrived to a fundamental relation which provides us a *geometriza*tion of the dual space of any given finite dimensional  $C^*$ -algebra [61]. Moreover, we may consider the decomposition of the associative product on the  $C^*$ -algebra according to

$$A \cdot_{K} B = \frac{1}{2} (A \cdot_{K} B + B \cdot_{K} A) + \frac{1}{2} (A \cdot_{K} B - B \cdot_{K} A)$$
(B.23)

and identify a corresponding induced decomposition of the tensor field by

$$\tau_K(d\widehat{A}, d\widehat{B})(\alpha) = \alpha(A \circ_K B) + i\alpha([A, B]_K),$$
(B.24)

which geometrizes the symmetric and the anti-symmetric products

$$A \circ_K B := \frac{1}{2} (A \cdot_K B + B \cdot_K A) \tag{B.25}$$

$$[A,B]_K := \frac{1}{2i} (A \cdot_K B - B \cdot_K A) \tag{B.26}$$

by means of a real symmetric

$$R_K(d\widehat{A}, d\widehat{B})(\alpha) := \alpha(A \circ_K B), \tag{B.27}$$

and an imaginary anti-symmetric

$$\Lambda_K(d\widehat{A}, d\widehat{B})(\alpha) := \alpha([A, B]_K), \tag{B.28}$$

(2,0)-tensor field on  $\mathcal{A}^*$ . Both structures are equivalently defined via symmetrized and anti-symmetrized star-products on the bi-dual of linear func-

tionals on  $\mathcal{A}^*$  according to

$$R_K(d\widehat{A}, d\widehat{B})(\alpha) = \frac{1}{2} (\widehat{A} \star_K \widehat{B} + \widehat{B} \star_K \widehat{A})|_\alpha := (\widehat{A}, \widehat{B})_k|_\alpha$$
(B.29)

$$\Lambda_K(d\widehat{A}, d\widehat{B})(\alpha) = \frac{1}{2i} (\widehat{A} \star_K \widehat{B} - \widehat{B} \star_K \widehat{A})|_\alpha := \{\widehat{A}, \widehat{B}\}_k|_\alpha.$$
(B.30)

It is clear that distinguished bi-linear products  $A \circ_K B$  and  $[A, B]_K$  and associated contra-variant tensors may be realized in this regard in dependence of the choice of the real matrix  $K \in M_n(\mathbb{C})$  coming along the product

$$A \cdot_K B \equiv AKB \tag{B.31}$$

of the  $C^*$ -algebra. In particular, if we choose the identity

$$K \equiv \mathbb{1} \in M_n(\mathbb{C}) \tag{B.32}$$

we recover the standard non-commutative associative matrix product, which allows to identify

$$R_{\mathbb{1}}(d\widehat{A}, d\widehat{B})(\alpha) := \alpha([A, B]_{+}) = (\widehat{A}, \widehat{B})|_{\alpha}, \tag{B.33}$$

$$\Lambda_1(d\widehat{A}, d\widehat{B})(\alpha) := \alpha([A, B]_-) = \{\widehat{A}, \widehat{B}\}|_\alpha, \tag{B.34}$$

a Riemanian and a Poisson tensor field, which are induced by a Jordan and a Lie product respectively. Correspondently, theses products provide a Jordan and a Poisson bracket on the bi-dual  $\operatorname{Lin}(\mathcal{A}^*) \subset \mathcal{F}(\mathcal{A}^*)$  of linear functionals on  $\mathcal{A}^*$ . Moreover, by restricting to the real elements of a complex matrix algebra endowed with the standard matrix product corresponding to K = 1, we may find that both products, the symmetric Jordan product and and Lie product are closed in respect to a Lie-Jordan algebra [41]. Hence, we may also directly start on the subspace of real elements provided by the Hermitian matrices and find an equivalent geometrization on the space of observables in terms of a Riemanian and a Poisson structure

$$R_{\mathbb{1}}(d\widehat{A}, d\widehat{B})(\xi) := \xi([A, B]_{+}) = (\widehat{A}, \widehat{B})|_{\xi}, \qquad (B.35)$$

$$\Lambda_{1}(d\widehat{A}, d\widehat{B})(\xi) := \xi([A, B]_{-}) = \{\widehat{A}, \widehat{B}\}|_{\xi},$$
(B.36)

for all

$$\xi \in u^*(n) := \operatorname{Lin}(u(n)) \tag{B.37}$$

and

$$A, B \in u(n), \quad \widehat{A}, \widehat{B} \in \operatorname{Lin}(u^*(n))$$
 (B.38)

on the space  $u^*(n)$  of real valued linear functionals defined on u(n).

**Remark B.2.** The geometric structures in (B.35), (B.36) can be recovered by a momentum-map push-forward of a contravariant Hermitian tensor field from a Hilbert space  $\mathcal{H} \cong \mathbb{C}^n$  to the corresponding space of observables  $u^*(n)$ [43]. In particular, one encounters in this way a Hermitian realization, which provides a generalization of symplectic realizations from symplectic to Poisson manifolds [61].

We may now 'close a circle', namely by defining a *covariant* defined tensor field on the vector space of a  $C^*$ -algebra  $\mathcal{A}$ . For a general state  $\omega \in \mathcal{A}^*$  this covariant tensor field will be a degenerate pull-back tensor field from a GNSconstructed Hilbert space  $\mathcal{H}_{\omega} := \mathcal{A}/J_{\omega}$ . The identification of such a covariant tensor field can be made explicit as follows.

## B.2 Covariant tensors on the vector space $\mathcal{A}$

As we remarked at the beginning, a  $C^*$ -algebra may admit a Banach Norm, which is not necessarily induced by an (Hermitian) inner product. However, from physical point of view, we are interested in the Hilbert spaces identified via the Gelfand-Naimark-Segal-construction (GNS). In this regard we shall restrict our attention to inner products defined by

$$\langle A | A \rangle_{\omega} := \omega(A^{\dagger}A) \tag{B.39}$$

on  $\mathcal{A}$ , with

$$\omega \in \mathcal{A}^*, \tag{B.40}$$

a fixed linear functional on  $\mathcal{A}$ . The latter is identified with a *state*, whenever we associate to it the two additional properties

Positivity :
$$\Leftrightarrow \omega(A^{\dagger}A) > 0$$
 (B.41)

Normalization :
$$\Leftrightarrow \omega(1) = \operatorname{Tr}(\rho_{\omega}) = 1.$$
 (B.42)

We will denote the space of states associated to a given  $C^*$ -algebra by  $D(\mathcal{A})$ . At this point we may consider a *covariant* tensor field on  $\mathcal{A}$ 

$$\tau_{\omega}(X_A, X_B)(a) := \omega(X_A(a) \cdot X_B(a)), \tag{B.43}$$

defined by a contraction with vector fields

$$X_C: \mathcal{A} \to T\mathcal{A} = \mathcal{A} \times \mathcal{A} \tag{B.44}$$

$$a \mapsto (a, C).$$
 (B.45)

Hence, we arrive to an invariant (0, 2)-tensor field

$$\tau_{\omega}(X_A, X_B)(a) = \omega(A^{\dagger}B). \tag{B.46}$$

A coordinate description is provided as follows. To a given element  $A \in \mathcal{A} = M_n(\mathbb{C})$ , we may either find a coordinate matrix description

$$A = \sum_{j,k} A_{jk} \left| j \right\rangle \left\langle k \right|. \tag{B.47}$$

or, by considering a basis  $\{|e_j\rangle\}_{j\in J}$  on the vector space  $\mathcal{A}$ , we may expand any given element  $A \in \mathcal{A}$  as a vector

$$A \equiv |A\rangle = \sum_{j=1}^{n} \langle e_j |A\rangle |e_j\rangle = \sum_{j=1}^{n} \alpha^j |e_j\rangle, \qquad (B.48)$$

where we identified coordinate functions

$$\langle e_j | A \rangle := \alpha^j(A), \quad \alpha^j \in \mathcal{A}^*.$$
 (B.49)

with linear functionals on  $\mathcal{A}$ . The identification

$$A \equiv |A\rangle \tag{B.50}$$

between (matrix) column-row objects A and (vector) column objects  $|A\rangle$  can be made clear due to the 1-to-1 correspondence between the linear functionals and matrix coefficients  $A_{jk}$ 

$$\{\alpha_l\}_{l\in I} \leftrightarrow \{A_{jk}\}_{j,k\in J},\tag{B.51}$$

resp. basis vectors

$$\{|e_l\rangle\}_{l\in I} \leftrightarrow \{|j\rangle\langle k|\}_{j,k\in J},$$
 (B.52)

Here we may introduce the notation

$$d|A\rangle \equiv |dA\rangle = dA,\tag{B.53}$$

defined by the exterior derivative on the coordinate functions according to

$$|dA\rangle := \sum_{j=1}^{n} d\alpha^{j} |e_{j}\rangle \tag{B.54}$$

and find

Proposition B.3.

$$\tau_{\omega} \equiv \langle dA \otimes dA \rangle_{\omega} := \omega (dA^{\dagger} \otimes dA). \tag{B.55}$$

*Proof.* One has to show that

$$\omega(dA^{\dagger} \otimes dA)(X_B, X_C)(a) = \omega(B^{\dagger}C).$$
(B.56)

By writing the left hand side in coordinates we find

$$\omega(dA^{\dagger} \otimes dA)(X_B, X_C)(a) = \sum_{j,k=1}^n \omega(|e_j\rangle \langle e_k|) d\alpha^j \otimes d\alpha^k(X_B, X_C)(a) \quad (B.57)$$

with  $X_B : a \mapsto (a, B)$  and  $X_C : a \mapsto (a, C)$ , and their expansion

$$X_B = \sum_{j=1} \langle e_j | B \rangle \frac{\partial}{\partial \alpha^j} = \sum_{j=1} \alpha_j(B) \frac{\partial}{\partial \alpha^j}$$
(B.58)

$$X_C = \sum_{j=1} \langle e_j | c \rangle \frac{\partial}{\partial \alpha^j} = \sum_{j=1} \alpha_j(C) \frac{\partial}{\partial \alpha^j}$$
(B.59)

in a holonomic basis of  $\mathcal{A}$ .

To this point, we may focus on some direct implications of the geometrization

$$(\mathcal{A}, \langle A | A \rangle_{\omega}) \to (\mathcal{A}, \langle dA \otimes dA \rangle_{\omega}) \tag{B.60}$$

of (degenerate) inner products on a given  $C^*$ -algebra. As indicated before, such a construction stands in a neat relation to the geometrization

$$(\mathcal{H}_{\omega}, \langle \psi | \psi \rangle) \to (\mathcal{H}_{\omega}, \langle d\psi \otimes d\psi \rangle)$$
(B.61)

of the Hermitian inner product  $\langle \psi | \psi \rangle$  on a Hilbert space by means of a Hermitian tensor field  $\langle d\psi \otimes d\psi \rangle$ , which we have been considered in the ge-

ometric formulation of quantum mechanics as presented here in this work in section 2. In fact, the later Hermitian tensor field may now be recovered from the GNS-construction associated to the above considered geometrized  $C^*$ -algebra and goes along the following lines.

**Remark B.4.** In general, the inner product of a vector space V admits a geometrization by means of a covariant tensor field, while the dual  $V^*$  takes into account a contra-variant tensor field. The case where V coincides with a Hilbert manifold is described in section 2.2.

## **B.3** Kählerian manifolds from states

We may now consider real coordinate description of the pair  $(\mathcal{A}, \langle dA \otimes dA \rangle_{\omega})$ . For this purpose we find on each element  $A \in \mathcal{A}$  the decomposition into

$$A = \sum_{j,k} A_{jk}(A) \left| j \right\rangle \left\langle k \right| \equiv \sum_{j,k} (Q_{jk}(A) + iP_{jk}(A)) \left| j \right\rangle \left\langle k \right| \tag{B.62}$$

where  $Q_{jk}, P_{jk} \in \mathcal{A}^*$  denote real scalar valued coordinate functions

$$Q_{jk}(A) = \frac{1}{2} \operatorname{Tr}(\widehat{A}(\alpha) + (\widehat{A}(\alpha))^{\dagger}) |j\rangle \langle k|) = \frac{1}{2} \operatorname{Tr}(\alpha(A) + (\alpha(A))^{\dagger} |j\rangle \langle k|) \quad (B.63)$$
$$P_{jk}(A) = \frac{1}{2i} \operatorname{Tr}(\widehat{A}(\alpha) - (\widehat{A}(\alpha))^{\dagger}) |j\rangle \langle k|) = \frac{1}{2i} \operatorname{Tr}(\alpha(A) - (\alpha(A))^{\dagger} |j\rangle \langle k|) \quad (B.64)$$

on  $\mathcal{A}$ . In this coordinates one finds the 1-forms  $dQ_{jk}, dP_{jk} \in \mathcal{T}_1^0(\mathcal{A})$  and an operator valued tensor

$$dA^{\dagger} \otimes dA = (dQ_{kj} - idP_{kj}) |j\rangle \langle k| \otimes (dQ_{ls} + idP_{ls}) |l\rangle \langle s|$$
$$= |j\rangle \langle k|l\rangle \langle s| (dQ_{kj} - idP_{kj}) \otimes (dQ_{ls} + idP_{ls})$$
$$= |j\rangle \langle s| (dQ_{kj} - idP_{kj}) \otimes (dQ_{ks} + idP_{ks})$$
$$= |j\rangle \langle s| (dQ_{kj} \otimes dQ_{ks} + dP_{kj} \otimes dP_{ks} + idQ_{kj} \otimes dP_{ks} - idP_{kj} \otimes dQ_{ks})$$
(B.65)

with the tensor product  $\otimes$  defined on  $\mathcal{T}_1^0(\mathcal{A})$ . Hence, the tensor field (B.55),  $\langle dA \otimes dA \rangle_{\omega}$  on  $\mathcal{A}$  will read in these coordinates as

$$\operatorname{Tr}(\omega |j\rangle \langle s|)(dQ_{kj} \otimes dQ_{ks} + dP_{kj} \otimes dP_{ks} + idQ_{kj} \otimes dP_{ks} - idP_{kj} \otimes dQ_{ks}).$$
 (B.66)

### **B.3.1** Pure states

In this setting we identify a GNS-construction map

$$\pi_{\omega}: \mathcal{A} \to \mathcal{H}_{\omega} := \mathcal{A}/J_{\omega} \tag{B.67}$$

defined by the Gelfand ideal  $J_{\omega}$ . If we consider the case of a pure state  $\omega := |1\rangle \langle 1|$  one gets due to the coefficients in (B.66),

$$\operatorname{Tr}(|1\rangle \langle 1 | j \rangle \langle s |) = \delta_{s1} \delta_{j1}, \qquad (B.68)$$

a *degenerate* covariant tensor field on  $\mathcal{A}$ 

$$dQ_{k1} \otimes dQ_{k1} + dP_{k1} \otimes dP_{k1} + idQ_{k1} \otimes dP_{k1} - idP_{k1} \otimes dQ_{k1}, \qquad (B.69)$$

which is a  $\pi_{\omega}$ -induced pull-back from a Kählerian tensor field living on the 'quotient'  $\mathcal{H}_{\omega}$ . This can be seen by setting  $Q_{k1} := q_k, P_{k1} := p_k$  which relates the degenerate pull-back tensor (B.69) to a Riemannian and symplectic structure

$$dq_k \otimes dq_k + dp_k \otimes dp_k + idq_k \otimes dp_k - idp_k \otimes dq_k$$
$$= dq_k \odot dq_k + dp_k \odot dp_k + idq_k \wedge dp_k$$
(B.70)

on the GNS-constructed Hilbert space  $\mathcal{H}_{\omega} = \mathbb{C}^n$ .

### B.3.2 Mixed states

Let us consider next the generalized case of a mixed state  $\omega := \sum_{i=1}^{m} \lambda_i |i\rangle \langle i|$ in terms of a rank-m projector with  $\lambda_i > 1$  and  $\sum_{i=1}^{m} \lambda_i = 1$ . The coefficients in (B.66) become then

$$\sum_{i=1}^{m} \operatorname{Tr}(\lambda_{i} |i\rangle \langle i |j\rangle \langle s|) = \sum_{i=1}^{m} \lambda_{i} \delta_{is} \delta_{ij}.$$
 (B.71)

In this way one finds the degenerate covariant tensor field on  $\mathcal{A}$ 

$$\sum_{i=1}^{m} \lambda_{i} (dQ_{ki} \otimes dQ_{ki} + dP_{ki} \otimes dP_{ki} + idQ_{ki} \otimes dP_{ki} - idP_{ki} \otimes dQ_{ki})$$

$$= \sum_{i=1}^{m} dq_{k}^{(i)} \otimes dq_{k}^{(i)} + dp_{k}^{(i)} \otimes dp_{k}^{(i)} + idq_{k}^{(i)} \otimes dp_{k}^{(i)} - idp_{k}^{(i)} \otimes dq_{k}^{(i)}$$

$$= \sum_{i=1}^{m} dq_{k}^{(i)} \otimes dq_{k}^{(i)} + dp_{k}^{(i)} \otimes dp_{k}^{(i)} + idq_{k}^{(i)} \wedge dp_{k}^{(i)}, \quad (B.72)$$

with  $\lambda_i Q_{ki} := q_k^{(i)}, \lambda_i P_{ki} := p_k^{(i)}$ . Again, we may identify this structure with a pull back tensor field from the corresponding GNS-constructed Hilbert space  $\mathcal{H}_{\omega} = \bigoplus_{i=1}^m \mathbb{C}^n$ .

#### B.3.3 Maximal rank states

It becomes clear that if we consider a (normalizing) multiple of the identity

$$\omega = \frac{1}{n}\mathbb{1}, \quad c \in \mathbb{R}_0 \tag{B.73}$$

we may identify an Hermitian inner product

$$\langle A | A \rangle_{\frac{1}{n}\mathbb{1}} := \frac{1}{n} \operatorname{Tr}(A^{\dagger}A),$$
 (B.74)

which makes out of any vector space of a given  $C^*$ -algebra  $\mathcal{A} \cong M_n(\mathbb{C})$  a Hilbert space. This inner product may admit two possible interpretations. First it recovers the inner product which induces the Hilbert-Schmidt norm on a finite dimensional Hilbert space of 'Hilbert Schmidt operators', whenever we encounter the isomorphism

$$\mathcal{A} \cong \mathcal{H}^* \otimes \mathcal{H} \tag{B.75}$$

in the category of Banach vector spaces.

As second interpretation we identify it with the Hermitian inner product on a GNS-constructed Hilbert space

$$\mathcal{H}_{\omega} := \mathcal{A}/\mathcal{J}_{\omega} \cong \mathcal{A} \text{ for } \omega = \frac{1}{n}\mathbb{1},$$
 (B.76)

which becomes isomorphic to the matrix Banach space itself, once the Gelfand ideal  $\mathcal{J}_{\omega}$  turns out to be trivial in the case for maximal mixed state  $\omega = \frac{1}{n}\mathbb{1}$ . This is also the case in the more general situations for states with maximal rank (m=n),

$$\omega := \sum_{i=1}^{n} \lambda_i \left| i \right\rangle \left\langle i \right|. \tag{B.77}$$

Hence, we end up with a GNS-induced Hermitian tensor field, which is nondegenerate on the whole space  $\mathcal{A}$ , which in this way becomes a Kählerian manifold.

**Remark B.5.** These observations may lead to consider the a family vector spaces

$$\mathcal{A}_{\alpha} := (\mathcal{A}, \langle A | A \rangle_{\alpha}) \tag{B.78}$$

with distinguished, in general degenerate inner products  $\langle A | A \rangle_{\alpha}$  parametrized by the space of states  $D(\mathcal{A})$  according to

$$\bigcup_{\alpha \in D(\mathcal{A})} \mathcal{A}_{\alpha}.$$
 (B.79)

Via the GNS-construction over each state one finds

$$\bigcup_{\alpha \in D(\mathcal{A})} \mathcal{A}_{\alpha} / \mathcal{J}_{\alpha} = \bigcup_{\alpha \in D(\mathcal{A})} \mathcal{H}_{\alpha} \cong_{loc} D(\mathcal{A}) \times \mathcal{H}_{\alpha},$$
(B.80)

and therefore a Kähler bundle over  $D(\mathcal{A})$  (see also [61] and references therein). Note that a Kähler bundle is not a fiber bundle in the usual sense, since its 'fibers'  $\mathcal{H}_{\alpha}$  are not isomorphic. The structure

$$\langle dA \otimes dA \rangle_{\alpha}$$
 (B.81)

provides therefore a family of tensor fields on  $\mathcal{A}$  being parametrized by  $\alpha \in D(\mathcal{A})$ .

# **B.4** Covariant tensors on $u^*(n)$ - Construction in the Bloch representation

Consider a decomposition of the vector space  $\mathcal{A} \cong M_n(\mathbb{C})$  of complex matrices according to

$$\mathcal{A} = \operatorname{Re}\mathcal{A} + \operatorname{Im}\mathcal{A}, \tag{B.82}$$

with

$$M_n(\mathbb{C}) \cong \mathbb{R}^{n^2} \oplus i\mathbb{R}^{n^2} \cong u^*(n) \oplus iu^*(n), \qquad (B.83)$$

into two subspaces of Hermitian and anti-Hermitian matrices by decomposing each element into real and imaginary part

$$A \equiv A_1 + iA_2, \quad A_k \in u^*(n), \tag{B.84}$$

with

$$A_1 := \frac{1}{2}(A + A^{\dagger}) \tag{B.85}$$

$$A_2 := \frac{1}{2i}(A - A^{\dagger}). \tag{B.86}$$

By introducing a basis  $\{\sigma_j\}_{j\in J}$  of Hermitian matrices on the real subspaces  $u^*(n)$  we identify coordinate functions

$$q^{j}(A_{1}) := \sigma_{j}(A_{1}) \equiv \operatorname{Tr}(\sigma_{j}A_{1})$$
(B.87)

$$p^{j}(A_{2}) := \sigma_{j}(A_{2}) \equiv \operatorname{Tr}(\sigma_{j}A_{2})$$
(B.88)

as particular instances of real-valued linear functionals on  $u^*(n)$  defined by the introduced basis elements. Hence, we have for all  $A_k \in u^*(n)$ 

$$q^j, p^j \in \operatorname{Lin}(u^*(n), \mathbb{R}),$$
 (B.89)

where we shall note, that  $q_1$  and  $p_2$  are functionals on distinguished spaces appearing as two (direct summed) copies of  $u^*(n)$ .

By means of this functionals we recover the Bloch representation expansions

$$A_1 = q^0(A_1)\mathbb{1} + \sum_{j=1} q^j(A_1)\sigma_j \equiv \sum_{j=0} q^j(A_1)\sigma_j$$
(B.90)

$$A_2 = p^0(A_2)\mathbb{1} + \sum_{k=1} p^k(A_2)\sigma_k \equiv \sum_{j=0} p^j(A_2)\sigma_j,$$
 (B.91)

where we set  $\sigma_0 \equiv 1$ . Within this coordinates, we identify the Hermitian matrix valued 1-forms

$$dA_1 = \sum_{j=0} dq^j (A_1) \sigma_j \tag{B.92}$$

$$dA_2 = \sum_{k=0} dp^k (A_2) \sigma_k \tag{B.93}$$

on the subspace  $u^*(n)$  of Hermitian matrices, which gives rise to an operator valued 1-form

$$dA = dA_1 + idA_2 \tag{B.94}$$

on  $\mathcal{A}$ . In this way we may identify an operator valued covariant tensor field

$$dA^{\dagger} \otimes dA.$$
 (B.95)

It decomposes into

$$dA_1 \odot dA_1 + dA_2 \odot dA_2 + idA_1 \wedge dA_2, \tag{B.96}$$

a symmetric and an anti-symmetric operator valued tensor (0,2)-tensor. Here we may consider the pull-back of the tensor field (B.96) from  $\mathcal{A}$  to  $\operatorname{Re}\mathcal{A} = u^*(n)$ , whenever we set  $dA_2 \equiv 0$ . In particular for n = 2 we find

$$\sigma_j dq^0 \odot dq^j + \sum_{j=0}^3 \mathbb{1} dq^j \odot dq^j + i\sigma_l \epsilon_{jkl} dq^j \wedge dq^k.$$
(B.97)

An evaluation with an dual element  $\omega \in \mathcal{A}^* \cong M_2(\mathbb{C})$  gives

$$\operatorname{Tr}(\omega\sigma_j)dq^0 \odot dq^j + \sum_{j=0}^3 \operatorname{Tr}(\omega)dq^j \odot dq^j + i\operatorname{Tr}(\omega\sigma_l)\epsilon_{jkl}dq^j \wedge dq^k.$$
(B.98)

If we set

 $\omega \equiv A_1 \tag{B.99}$ 

we find

$$q^{j}dq^{0} \odot dq^{j} + 2q^{0}dq^{j} \odot dq^{j} + iq^{l}\epsilon_{jkl}dq^{j} \wedge dq^{k}.$$
 (B.100)

On the other hand, if  $\omega$  is a multiple of the identity

$$\omega \equiv c\mathbb{1},\tag{B.101}$$

we end up with an Euclidean tensor field. In particular for  $c \equiv 1/n$  we have

$$\sum_{j=0}^{3} dq^j \odot dq^j. \tag{B.102}$$

## B.5 Tensors from *G*-orbits - A relation to IOVTs

For a general submanifold N being embedded in the vector space  $\mathcal{A}$  of a  $C^*$ -algebra of finite dimensional operators, we will find for a given embedding

$$\iota: N \hookrightarrow \mathcal{A},\tag{B.103}$$

a pull-back tensor field

$$\iota_N^* \left\langle dA \otimes dA \right\rangle_\omega \tag{B.104}$$

on N. In particular we will be interested in the following to consider cases where N admits the structure of a homogenous space by means of orbits associated to a unitary representation of a Lie group  $\mathcal{G}$  in  $\mathcal{A}$ . By considering for this purpose a family of actions of  $\mathcal{G}$  on a given fiducial operator  $A_0 \in$  $\mathcal{A}$ , we encounter several possibilities for realizing the embedding of these orbits in  $\mathcal{A}$ . There will be indeed different representations of the group  $\mathcal{G}$  generating the corresponding orbit  $\mathcal{G}/\mathcal{G}_{A_0}$  in  $\mathcal{A}$  by acting on this fiducial point transitively (all invariant actions to be identified are denoted here by the isotropy group  $\mathcal{G}_{A_0}$ ). Hence, we have to distinguish different embedding of the orbit in dependence of the chosen representation. As a first fundamental difference we shall distinguish two classes provided by vector representation actions on the one hand, and adjoint representation actions on the other hand.

#### B.5.1 Vector representation induced orbits

By taking into account the  $n^2$ -dimensional complex vector space structure of  $\mathcal{A}$  we may identify vector representations

$$Q \to \operatorname{Aut}(\mathcal{A}) = GL(n^2, \mathbb{C}),$$
 (B.105)

in particular by focusing on those given by unitary vector representations of a unitary subgroup  $Q \subset U(n)$ ,

$$Q \to U(n^2) \subset GL(n^2, \mathbb{C}),$$
 (B.106)

$$U(q)U(q') = U(q q'),$$
 (B.107)

with  $q, q' \in Q$ .

**Remark B.6.** The following consideration may become valid also for unitary

representations of general Lie groups  $\mathcal{G}$ . For sake of simplicity however, we shall restrict in this and the following subsection to case of a unitary representation of a unitary subgroup  $Q \subset U(n^2)$ .

The action on a fiducial point  $A_0 \in \mathcal{A}$  reads

$$U(q)A_0 \equiv A_q \in \mathcal{A}, \quad q \in Q \tag{B.108}$$

and defines an embedding of the group manifold Q in the dual algebra  $\mathcal{A}$ by means of an orbit  $Q/Q_{A_0}$ . A particular simple embedding of Q may be achieved, whenever we have the special setting given by  $Q = U(n^2)$ ,  $A_0 = \mathbb{1}$ , and U(q) = q, the defining representation. The embedding action (B.108) reduces to

$$U(q)\mathbb{1} \equiv q \in \mathcal{A}, \ q \in Q, \tag{B.109}$$

It is important to underline this special case, since we will encounter much more possibilities of realizing the embedding, once we have to deal with more generic situations given by  $Q \subset U(n)$ .

A homogeneous space  $Q/Q_{A_0}$  which becomes embedded according to (B.108) in  $\mathcal{A}$  induces a pulled back tensor

$$\iota_Q^* \left\langle dA \otimes dA \right\rangle_\omega = \operatorname{Tr}(\omega \, dA_q^\dagger \otimes dA_q) \tag{B.110}$$

on the Lie group Q.

To give an explicit expression of the differentials involved in the pulled back tensor, we need to specify a Lie algebra representation

$$R: T_e Q \to T_e U(n), \tag{B.111}$$

$$[R(e_j), R(e_k)] = R([e_j, e_k]),$$
(B.112)

associated to (B.106) where  $\{e_j\}_{j \in J}$  denotes a basis of Hermitian matrices spanning the Lie algebra  $T_eQ$ . By the use of anti-Hermitian matrix valued left invariant 1-forms

$$U^{\dagger}(q)dU(q) \equiv iR(e_j)\theta^j(q) \tag{B.113}$$

composed out of a basis  $\{\theta_j\}_{j\in J}$  of left invariant 1-forms on the group manifold and a basis  $\{e_j\}_{j\in J}$  of Hermitian matrices spanning the tangent space  $T_e U(n)$ , one finds then

$$dU(q) = iU(q)R(e_j)\theta^j, \tag{B.114}$$

and therefore

$$dA_q = dU(q)A_0 = iU(q)R(e_j)\theta^j A_0.$$
 (B.115)

In this way the pulled back tensor field reads

$$\operatorname{Tr}(\omega \, dA_q^{\dagger} \otimes dA_q) = \operatorname{Tr}(Ad_{A_0}(\omega)R(e_j)R(e_k))\theta^j \otimes \theta^k, \tag{B.116}$$

admitting a decomposition into a symmetric

$$\operatorname{Tr}(Ad_{A_0}(\omega) [R(e_j), R(e_k)]_+) \theta^j \odot \theta^k, \qquad (B.117)$$

and an anti-symmetric tensor

$$i \operatorname{Tr}(Ad_{A_0}(\omega) [R(e_j), R(e_k)]_{-}) \theta^j \wedge \theta^k.$$
 (B.118)

Here we shall identify the coefficients of the tensor in the above decomposition by means of

$$L_{jk} := \operatorname{Tr}(Ad_{A_0}(\omega)R(e_j)R(e_k)) \tag{B.119}$$

$$L_{(jk)} := \text{Tr}(Ad_{A_0}(\omega) [R(e_j), R(e_k)]_+))$$
(B.120)

$$L_{[jk]} := \text{Tr}(Ad_{A_0}(\omega) [R(e_j), R(e_k)]_{-}).$$
(B.121)

Remark B.7. The symmetric part of the tensor field may become related to a

Riemannian structure if  $Ad_{A_0}(\omega)$  is a Hermitian operator which is positive. In general however, we may encounter also pseudo-Riemanian structures, whenever  $Ad_{A_0}(\omega)$  turns out to be an element of the sub-algebra  $u_{k_+,k_-}^*(n) \subset \mathcal{A}$  of Hermitian matrices with  $k_+$  positive and  $k_-$  negative eigenvalues. The relation of the of the sub-algebra  $u_{k_+,k_-}^*(n)$  with semi-Hermitian structures has been described in [43].

**Remark B.8.** For  $A_0 = 1$  we will have a trivial isotropy group  $Q_1 = \{1\}$ . In this case the anti-symmetric tensor

$$Tr(\omega[R(e_j), R(e_k)]_{-})\theta^j \wedge \theta^k$$
 (B.122)

identifies a pull-back structure on the orbit  $Q/Q_1 = Q$ . It will vanish in dependence of the state  $\omega$  in all directions  $R(e_i)$  with

$$[R(e_j), \omega]_{-} = 0 \tag{B.123}$$

due to

$$Tr(\omega[R(e_j), R(e_k)]_{-}) = Tr(R(e_k)[\omega, R(e_j)]_{-}).$$
 (B.124)

With other words the pulled back tensor on the orbit  $Q/Q_1 = Q$  will be degenerate in these directions. On the corresponding quotient space  $Q/Q_{\omega}$ however, with  $Q_{\omega}$ , an isotropy subgroup of Q being generated by all Hermitian matrices  $R(e_j)$  which commute with the state  $\omega$ , this tensor becomes related to a non-degenerate and therefore symplectic structure. Moreover, it coincides with the symplectic tensor

$$Tr(\omega \, dU^{\dagger} \wedge dU),$$
 (B.125)

which has been related to the analysis on degeneracy structures of geometric phases for multi-level quantum systems discussed in [72].

In particular we may distinguish between different types of pulled back

tensors according to the three special cases

$$Ad_{A_0}(\omega) = \begin{cases} \omega & \text{for } A_0 = \mathbb{1} \\ A_0 A_0^{\dagger} & \text{for } \omega = \mathbb{1} \\ A_0^2 A_0^{\dagger} & \text{for } \omega = A_0. \end{cases}$$
(B.126)

Crucially, the first two cases provide a relation to tensor fields from quantum state-evaluated IOVTs on Lie groups  $\mathcal{G}$  as discussed in section 4.5.1. Indeed, in the second case,  $A_0A_0^{\dagger}$  is a Hermitian operator, which is positive. Hence, by taking into account a normalizing factor  $(\text{Tr}(A_0A_0^{\dagger}))^{-1}$  we shall be able to close a circle to the IOVT construction as done in section 4.5.1. This will be discussed for general Lie groups  $\mathcal{G}$  in detail in subsection B.5.3.

### B.5.2 Adjoint representation induced orbits

The unitary representation  $U(q) \in U(n)$  in (B.106) may induce a nonequivalent class of embeddings of orbits in  $\mathcal{A}$  by means of the adjoint action

$$A_q \equiv U(q)A_0 U^{\dagger}(q), \qquad (B.127)$$

on a fiducial point  $A_0 \in A$ . The corresponding isotropy group  $Q_{A_0}$  becomes in this case defined by all  $q \in Q$  for which

$$[U(q), A_0] = 0 \tag{B.128}$$

holds. The induced pulled back tensor on  $Q \subset \mathcal{A}$  may read then again formally

$$\iota_Q^* \left\langle dA \otimes dA \right\rangle_\omega = \operatorname{Tr}(\omega \, dA_q^\dagger \otimes dA_q). \tag{B.129}$$

The evaluation of the coefficients however, will differ crucially from those in (B.119) - (B.121) associated to the vector representation induced embedding

(B.108), as we will show in the following. With the differential  $dA_q$  given by

$$dA_q = Ad_{U(q)}[U^{\dagger}(q)dU(q), A_0]$$
 (B.130)

and the operator-valued left invariant 1-forms

$$U^{\dagger}(q)dU(q) = iR(e_j)\theta_j(q)$$
(B.131)

with  $\{\theta_j\}_{j\in J}$ , as defined in section 4.5.1 as a basis of left-invariant 1-forms on the Lie group manifold Q (with a Lie-algebra admitting a basis  $\{e_j\}_{j\in J}$ ) , one finds

$$\operatorname{Tr}(\omega(U[iR(e_j), A_0]U^{\dagger}U[iR(e_k), A_0]U^{\dagger}))\theta^j \otimes \theta^k$$
$$= -\operatorname{Tr}(Ad_{U(q)}(\omega)[R(e_j), A_0][R(e_k), A_0])\theta^j \otimes \theta^k.$$
(B.132)

This tensor field shows clearly that it is degenerate along the commutant of  $A_0$ . Hence, it provides the pull-back tensor field from a non-degenerate structure on the homogeneous space  $Q/Q_{A_0}$ .

By focusing on the coefficients of the tensor, evaluated on the identity  $q_0 \equiv 1 \in Q$  with  $U(q_0) = 1$  we find

$$K_{jk} := -\text{Tr}(\omega[R(e_j), A_0][R(e_k), A_0]), \qquad (B.133)$$

admitting via (anti)-symmetrization a decomposition

$$-\mathrm{Tr}(\omega[[R(e_j), A_0], [R(e_k), A_0]]_+) - i\mathrm{Tr}(\omega[[R(e_j), A_0], [R(e_k), A_0]]_-).$$
(B.134)

Hence, the tensor coefficients may denoted in the following form

$$K_{jk} = K_{(jk)} + iK_{[jk]},$$
 (B.135)

decomposed into a symmetric part

$$K_{(jk)} := -\text{Tr}(\omega[[R(e_j), A_0], [R(e_k), A_0]]_+)$$
(B.136)

and an anti-symmetric part

$$K_{[jk]} := -\text{Tr}(\omega[[R(e_j), A_0], [R(e_k), A_0]]_{-}).$$
(B.137)

Remark B.9. The anti-symmetric part

$$K_{[jk]} = -Tr([R(e_j), A_0] [\omega, [R(e_k), A_0]]_{-})$$
(B.138)

will vanish along all directions  $[R(e_k), A_0]$  for which

$$[\omega, [R(e_k), A_0]]_{-} = 0 \tag{B.139}$$

holds. It defines therefore a symplectic structure on the quotient space  $(Q/Q_{A_0})/Q_{\omega}$ with  $Q_{\omega}$ , the isotropy group generated by the corresponding invariant directions  $[R(e_k), A_0]$ . For the special case  $\omega = 1$  one finds therefore only a symmetric part

$$K_{jk} = K_{(jk)} = -Tr([R(e_j), A_0][R(e_k), A_0]),$$
(B.140)

which defines a left-invariant metric [72].

The tensor coefficients described in (B.133)-(B.137) may extract local properties of the orbits. Indeed, by considering variational aspects of these tensors in the neighborhood of  $q_0 = 1$ , we may find also global properties of the orbits encoded by higher rank tensors, like for instance the Riemanian curvature associated to a Riemannian metric (B.140). Anti-symmetric tensors which are symplectic on the other hand, may be used for the constructions of volume-forms. More general, by applying k times the tensor product on the differential (B.130)

$$dA_q = Ad_{U(q)}[iR(e_j), A_0]\theta_j(q)$$
(B.141)

of the matrix valued function  $A_q$  on the orbit, we may consider in this regard a (0,k)-tensor construction

$$\bigotimes_{l=l_1}^{l_k} dA_q = (i)^k U(q) \prod_{l_1}^{l_k} [R(e_l), A_0] U(q)^{\dagger} \bigotimes_{l=l_1}^{l_k} \theta^l$$
(B.142)

and its evaluation on the tangent spaces according to

$$(i)^k \operatorname{Tr}\left(Ad_{U(q)}(\omega) \prod_{l=l_1}^{l_k} [R(e_l), A_0]\right) \bigotimes_{l=l_1}^{l_k} \theta^l.$$
(B.143)

In this way we may identify for instance the coefficients of a scalar valued (0,4)-tensor field defined globally on the orbit by

$$Tr(Ad_{U(q)}(\omega)[R(e_l), A_0][R(e_r), A_0][R(e_j), A_0][R(e_k), A_0]).$$
(B.144)

The later reduces to

$$K_{lrjk} := \operatorname{Tr}(\omega[R(e_l), A_0][R(e_r), A_0][R(e_j), A_0][R(e_k), A_0]), \qquad (B.145)$$

when evaluated over  $q_0 = 1$ .

### **B.5.3** A relation to IOVTs on $\mathcal{G}$ : The case $\omega = \mathbb{1}$

Let us restart by considering an inner product

$$c \operatorname{Tr}(\omega A^{\dagger} A)$$
 (B.146)

on a  $C^*$ -matrix algebra  $\mathcal{A} \cong M_n(\mathbb{C})$  which becomes identified with a Hilbert manifold of Hilbert Schmidt operators in finite dimensions for  $\omega = 1$ , where  $c \in \mathbb{R}_0$  denotes a normalization constant. With (B.55), we may promote this inner product to a tensor field

$$c \operatorname{Tr}(\omega dA^{\dagger} \otimes dA),$$
 (B.147)

on the vector space  $\mathcal{A}$ . For  $\omega = 1$  we find:

**Theorem B.10.** Let  $\{\theta_j\}_{j\in J}$  a basis of left-invariant 1-forms on  $\mathcal{G}$ , and  $\{X_j\}_{j\in J}$ , a Lie algebra basis of  $\mathcal{G}$  with  $\{iR(X_j)\}_{j\in J}$ , its representation in the Lie-Algebra of  $U(\mathcal{H})$  associated to a unitary representation  $U : \mathcal{G} \to U(\mathcal{H})$ . The submersion

$$\iota: \mathcal{G} \to \mathcal{A} \tag{B.148}$$

defined by the action

$$A_g \equiv \begin{cases} U(g)A_0U(g)^{-1}, \ (adjoint \ represention) \\ \\ U(g)A_0, \qquad (vector \ representation) \end{cases}$$
(B.149)

induces then a pull-back of the tensor field

$$\beta := c \operatorname{Tr}(dA^{\dagger} \otimes dA) \tag{B.150}$$

from  $\mathcal{A}$  to  $\mathcal{G}$ , according to

$$\iota_{\mathcal{G}}^*\beta = \begin{cases} c \operatorname{Tr}([R(X_j), A_0^{\dagger}][R(X_k), A_0])\theta^j \otimes \theta^k, \\ \\ c \operatorname{Tr}(A_0 A_0^{\dagger} R(X_k) R(X_k))\theta^j \otimes \theta^k \end{cases}$$
(B.151)

associated to the adjoint and the vector representation acting on  $A_0 \in \mathcal{A} \cong$ 

 $M_n(\mathbb{C})$  respectively.

*Proof.* The pull-back of the tensor field

$$\iota_{\mathcal{G}}^*\beta = c\mathrm{Tr}(d\iota_{\mathcal{G}}^*(A^{\dagger}) \otimes d\iota_{\mathcal{G}}^*(A)) \tag{B.152}$$

is provided by the pull-back of a function  $A \in \text{Lin}(\mathcal{A})$  on the dual algebra  $\mathcal{A}$  to a function

$$\iota_{\mathcal{G}}^*(A) \equiv A_{(\bullet)} : g \mapsto A_g \tag{B.153}$$

on the Lie group manifold  $\mathcal{G}$ . The pull-back of the function is induced by the immersion of the Lie-group in dependence of the chosen action in (B.149). Hence, the pull-back tensor is given by evaluating

$$c \operatorname{Tr}(dA_g^{\dagger} \otimes dA_g).$$
 (B.154)

Let us begin with the adjoint representation, where we consider the relation  $d(U(g)U(g)^{-1}) = 0$ , resp.

$$dU(g)^{-1} = -U(g)^{-1}dU(g)U(g)^{-1},$$
(B.155)

within the exterior derivative

$$d\rho_g = dU(g)A_0U(g)^{-1} + U(g)A_0dU(g)^{-1}$$
$$= dU(g)A_0U(g)^{-1} - U(g)A_0U(g)^{-1}dU(g)U(g)^{-1}.$$
(B.156)

By multiplication with the identity on the first summand we get

$$U(g)U(g)^{-1}dU(g)A_0U(g)^{-1} - U(g)A_0U(g)^{-1}dU(g)U(g)^{-1}$$
$$= U(g)([iR(X_j)\theta^j, A_0])U(g)^{-1},$$

where we identify the left-invariant operator valued-1-forms  $U(g)^{-1}dU(g)\equiv$ 

 $iR(X_j)\theta^j$ . Furthermore we find in the last expression the adjoint representation

$$\operatorname{Ad}_{U(g)}[iR(X_j)\theta^j, A_0]. \tag{B.157}$$

Since  $\theta^j$  is scalar-valued, we may reorganize the operator valued 1-form  $d\rho_g$  into

$$dA_g = \operatorname{Ad}_{U(g)}[iR(X_j), A_0]\theta^j.$$
(B.158)

The pull-back tensor  $c \operatorname{Tr}(dA_g^{\dagger} \otimes dA_g)$  reads then

$$c \operatorname{Tr}(\operatorname{Ad}_{U(g)}[-iR(X_j), A_0^{\dagger}]\theta^j \otimes \operatorname{Ad}_{U(g)}[iR(X_k), A_0]\theta^k).$$
(B.159)

By defining  $\otimes$  on the  $\theta^k$ -spanned module of 1-forms  $\mathcal{T}_1^0(\mathcal{G})$  being related via the pull-back to the tensor product on  $F := \mathcal{T}_1^0(u^*(\mathcal{H}))$ , we find

$$c \operatorname{Tr}(\operatorname{Ad}_{U(g)}[R(X_j), A_0^{\dagger}] \operatorname{Ad}_{U(g)}[R(X_k), A_0]) \theta^j \otimes \theta^k$$
$$= c \operatorname{Tr}(\operatorname{Ad}_{U(g)}([R(X_j), A_0^{\dagger}][R(X_k), A_0])) \theta^j \otimes \theta^k.$$
(B.160)

By using

$$\operatorname{Tr}(\operatorname{Ad}_U C) = \operatorname{Tr}(U(CU^{-1})) = \operatorname{Tr}((CU^{-1})U) = \operatorname{Tr}(C),$$

for any complex matrix C, we end up with the statement for the case of an adjoint representation induced immersion.

In the case of a vector representation induced immersion we have to consider the exterior derivative

$$dA_g = dU(g)A_0 = U(g)U^{-1}(g)dU(g)A_0 = iU(g)R(X_j)A_0\theta^j$$
(B.161)

and its tensor product yielding

$$dA_g^{\dagger} \otimes dA_g = A_0^{\dagger} R(X_j) R(X_k) A_0 \theta^j \otimes \theta^k$$
(B.162)

and therefore the evaluation

$$c \operatorname{Tr}(A_0^{\dagger} R(X_j) R(X_k) A_0) \theta^j \otimes \theta^k.$$
(B.163)

**Corollary B.11.** Let  $T_{jk}^{\alpha}$  and  $T_{jk}^{\beta}$  denote the coefficients of the pull-back tensor on a Lie group  $\mathcal{G}$  obtained by a corresponding unitary action on  $(\mathcal{H}, \alpha)$ according to [3, 55] and into  $(\mathcal{A}, \beta)$  according to Theorem B.10 respectively. Up to a normalization constant  $c \in \mathbb{R}$ , one finds:

(a) If  $A_0$  is Hermitian in the adjoint representation then

$$T_{jk}^{\beta} = T_{kj}^{\beta}.\tag{B.164}$$

(b) If  $A_0$  is a normalized rank-1 projector in the adjoint representation then

$$T^{\beta}_{(jk)} = T^{\alpha}_{(jk)}$$
 (B.165)

(c) If  $A_0$  is a normalized rank-1 projector in the vector representation then

$$T^{\beta}_{[jk]} = T^{\alpha}_{[jk]} \tag{B.166}$$

*Proof.* (a) It is clear that the coefficients

$$c \operatorname{Tr}([R(X_j), A_0^{\dagger}][R(X_k), A_0])$$
 (B.167)

are symmetric in the indices j and k due to the permutation invariance within the trace if  $A_0^{\dagger} = A_0$ .

(b) The expansion shows that these coefficients may be seen furthermore

decomposed into a sum of two terms<sup>17</sup>

$$Tr(A_0^2[R(X_j), R(X_k)]_+) - Tr(A_0R(X_j)A_0R(X_k)),$$
(B.168)

where we set the normalization constant c equal -1/2. At this point we chose the fiducial matrix to be pure, i.e.  $A_0^2 = A_0$ , say

$$A_0 := |0\rangle \langle 0| \in u^*(\mathcal{H}), \tag{B.169}$$

and find

 $Tr(A_0[R(X_j), R(X_k)]_+) - Tr(A_0R(X_j))Tr(A_0R(X_k)).$ (B.170)

(c) For the anti-symmetrized pull-back coefficients in the vector representation we have

$$c \operatorname{Tr}(A_0^2[R(X_j), R(X_k)]_{-})$$
 (B.171)

Hence, the second statement is immediate.

In conclusion, we have found a generalization of the symmetric and (antisymmetric) part of the pull-back tensor obtained from the Hilbert space. This tensor field may become non-degenarate and therefore Riemannian (symplectic), whenever it descends from the Lie group  $\mathcal{G}$  to a unitarily generated orbit  $\mathcal{G}/\mathcal{G}_0$  in  $\mathcal{A}$  associated to a fiducial Hermitian matrix and an unitary adjoint (vector) representation action  $\phi : \mathcal{G} \times \mathcal{A} \to \mathcal{A}$  with isotropy group  $\mathcal{G}_0 := \{g \in \mathcal{G} | \phi(g, A_0) = \rho_g = A_0 \in \mathcal{A}\}$ . For a possible application finally, we will need:

**Corollary B.12** (Relation to IOVTs). The fiducial operator  $A_0$ -dependent pull-back of

$$\beta = c \operatorname{Tr}(dA^{\dagger} \otimes dA), \qquad (B.172)$$

<sup>&</sup>lt;sup>17</sup>We distinguish the conventions  $[A, B]_+ = \frac{1}{2}(AB + BA), [A, B]_- = \frac{1}{2i}(AB - BA)$ , and [A, B] = AB - BA.

to a unitarily represented Lie group  $\mathcal{G}$ , coincides in the vector representation with an operator valued tensor field (IOVT) evaluated on a linear functional

$$\rho := cA_0 A_0^{\dagger} \in \mathcal{A}^* \tag{B.173}$$

which is

$$\begin{cases} positive for c = 1 \\ positive and normalized for c = \frac{1}{Tr(A_0^{\dagger}A_0)} \end{cases}$$
(B.174)

*Proof.* The vector representation action induces on  $\text{Tr}(A_0^{\dagger}A_0)$  the transformation

$$Tr((U(g)A_0)^{\dagger}U(g)A_0) = Tr(A_0^{\dagger}U(g)^{\dagger}U(g)A_0) = Tr(A_0^{\dagger}A_0).$$
(B.175)

By comparing the operator valued tensor in (4.76) with

$$\iota_{\mathcal{G}}^*\beta = c \operatorname{Tr}(A_0 A_0^{\dagger} R(X_k) R(X_k)) \theta^j \otimes \theta^k, \qquad (B.176)$$

we may set  $\rho := cA_0 A_0^{\dagger}$  for any  $A_0 \in \mathcal{A} \cong M_n(\mathbb{C})$ .

# C Tensors from generalized momentum maps

While the Fubini-Study metric identifies a quantum information metric for pure states [7], we may try to approach possible quantum information metrics also for the generalized regime of mixed states as seen form the Hilbert space. Let us for this purpose generalize constructively the notion of a pure state to a mixed state  $\rho$  by means of a convex combination of pure states  $\rho_{\psi_l} := \rho_l$ , according to

$$\rho := \sum_{l} p_l \rho_l, \text{ and } \sum_{l} p_l = 1.$$
 (C.1)

This suggests to consider a generalized momentum map

$$\widetilde{\mu} : \bigoplus_{j} \mathcal{H}_{0} \longrightarrow u^{*}(\mathcal{H}),$$
(C.2)

$$\tilde{\mu}(|\psi_1\rangle \oplus |\psi_2\rangle \oplus ...) = \sum_{l} p_l \mu(|\psi_l\rangle) = \sum_{l} p_l \frac{|\psi_l\rangle \langle \psi_l|}{\langle \psi_l |\psi_l\rangle} := \rho, \quad (C.3)$$

which is defined on a direct sum of Hilbert spaces. After considering the pull-back tensor

$$\frac{1}{2} \operatorname{Tr}(d\rho \otimes d\rho) = \frac{1}{2} \sum_{l,s} p_l p_s \operatorname{Tr}(d \frac{|\psi_l\rangle \langle \psi_l|}{\langle \psi_l | \psi_l \rangle} \otimes d \frac{|\psi_l\rangle \langle \psi_l|}{\langle \psi_l | \psi_l \rangle})$$
(C.4)

by means of the differentials

$$d\frac{|\psi_l\rangle\langle\psi_l|}{\langle\psi_l|\psi_l\rangle} = \frac{|d\psi_l\rangle\langle\psi_l| + |\psi_l\rangle\langle d\psi_l|}{\langle\psi_l|\psi_l\rangle} - \frac{|\psi_l\rangle\langle\psi_l|d\langle\psi_l|\psi_l\rangle}{\langle\psi_l|\psi_l\rangle^2}, \quad (C.5)$$

in the expanded expression

$$\sum_{l,s} p_{l} p_{s} \left( \frac{\langle d\psi_{s} \otimes d\psi_{l} \rangle}{\langle \psi_{l} | \psi_{l} \rangle \langle \psi_{s} | \psi_{s} \rangle} \langle \psi_{l} | \psi_{s} \rangle + \frac{\langle d\psi_{l} \otimes d\psi_{s} \rangle}{\langle \psi_{l} | \psi_{l} \rangle \langle \psi_{s} | \psi_{s} \rangle} \langle \psi_{s} | \psi_{l} \rangle \right) \\
- \frac{\langle \psi_{s} | d\psi_{l} \rangle \otimes d \langle \psi_{s} | \psi_{s} \rangle}{2 \langle \psi_{l} | \psi_{l} \rangle \langle \psi_{s} | \psi_{s} \rangle^{2}} \langle \psi_{l} | \psi_{s} \rangle - \frac{\langle d\psi_{l} | \psi_{s} \rangle \otimes d \langle \psi_{s} | \psi_{s} \rangle}{2 \langle \psi_{l} | \psi_{l} \rangle \langle \psi_{s} | \psi_{s} \rangle^{2}} \langle \psi_{s} | \psi_{l} \rangle \\
- \frac{d \langle \psi_{l} | \psi_{l} \rangle \otimes \langle \psi_{l} | d\psi_{s} \rangle}{2 \langle \psi_{s} | \psi_{s} \rangle \langle \psi_{l} | \psi_{l} \rangle^{2}} \langle \psi_{s} | \psi_{l} \rangle - \frac{d \langle \psi_{l} | \psi_{l} \rangle \otimes \langle d\psi_{s} | \psi_{l} \rangle}{2 \langle \psi_{s} | \psi_{s} \rangle \langle \psi_{l} | \psi_{l} \rangle^{2}} \langle \psi_{l} | \psi_{s} \rangle \\
+ \frac{d \langle \psi_{l} | \psi_{l} \rangle \otimes d \langle \psi_{l} | d\psi_{s} \rangle}{2 \langle \psi_{s} | \psi_{s} \rangle^{2} \langle \psi_{l} | \psi_{l} \rangle^{2}} \langle \psi_{l} | \psi_{s} \rangle \langle \psi_{s} | \psi_{s} \rangle \\
+ \frac{\langle \psi_{s} | d\psi_{l} \rangle \otimes \langle \psi_{l} | d\psi_{s} \rangle}{2 \langle \psi_{l} | \psi_{l} \rangle \langle \psi_{s} | \psi_{s} \rangle} + \frac{\langle d\psi_{l} | \psi_{s} \rangle \otimes \langle d\psi_{s} | \psi_{l} \rangle}{2 \langle \psi_{l} | \psi_{l} \rangle \langle \psi_{s} | \psi_{s} \rangle} \right), \quad (C.6)$$

one may impose the constraint

$$\langle \psi_l | \psi_s \rangle - \delta_{ls} = 0, \tag{C.7}$$

yielding  $\langle d\psi_l | \psi_s \rangle = - \langle \psi_l | d\psi_s \rangle$ , and find

$$\sum_{l} p_{l}^{2} \left( \frac{\langle d\psi_{l} \otimes d\psi_{l} \rangle}{\langle \psi_{l} | \psi_{l} \rangle} - \frac{\langle \psi_{l} | d\psi_{l} \rangle \otimes \langle d\psi_{l} | \psi_{l} \rangle}{2 \langle \psi_{l} | \psi_{l} \rangle^{2}} - \frac{\langle d\psi_{l} | \psi_{l} \rangle \otimes \langle \psi_{l} | d\psi_{l} \rangle}{2 \langle \psi_{l} | \psi_{l} \rangle^{2}} \right) + \sum_{l \neq s} p_{l} p_{s} \left( \frac{\langle \psi_{s} | d\psi_{l} \rangle \otimes \langle \psi_{l} | d\psi_{s} \rangle}{\langle \psi_{l} | \psi_{l} \rangle \langle \psi_{s} | \psi_{s} \rangle} \right).$$
(C.8)

For pure states, this tensor field 'collapses' to the Fubini-Study metric as seen from the Hilbert space in (2.11).

## References

- P. Aniello, J. Clemente-Gallardo, G. Marmo, and G. F. Volkert. Classical Tensors and Quantum Entanglement I: Pure States. *Int. J. Geom. Meth. Mod. Phys.*, 7:485, 2010.
- G. Marmo and G. F. Volkert. Geometrical description of quantum mechanics – transformations and dynamics. *Physica Scripta*, 82(3):038117, September 2010.
- [3] G. F. Volkert. Tensor fields on orbits of quantum states and applications. Dissertation, LMU München: Fakultät für Mathematik, Informatik und Statistik, Aug 2010.
- [4] P. Aniello, J. Clemente-Gallardo, G. Marmo, and G. F. Volkert. Classical Tensors and Quantum Entanglement II: Mixed States. Int. J. Geom. Meth. Mod. Phys., 8(3):1–32, 2011.

- [5] P. Aniello, J. Clemente-Gallardo, G. Marmo, and G. F. Volkert. From Geometric Quantum Mechanics to Quantum Information. ArXiv eprints, http://arxiv.org/abs/1101.0625, January 2011.
- [6] P. Aniello, J. Clemente-Gallardo, G. Marmo, and G. F. Volkert. Tensorial characterization and quantum estimation of weakly entangled qubits. *preprint*, October 2011.
- [7] P. Facchi, R. Kulkarni, V. I. Man'ko, G. Marmo, E. C. G. Sudarshan, and F. Ventriglia. Classical and quantum Fisher information in the geometrical formulation of quantum mechanics. *Physics Letters A*, 374:4801–4803, November 2010.
- [8] M. G. Genoni, P. Giorda, and M. G. A. Paris. Optimal estimation of entanglement. *Physical Review A*, 78(3):032303-+, September 2008.
- [9] R. P. Feynman. Simulating Physics with Computers. International Journal of Theoretical Physics, 21:467–488, June 1982.
- [10] M. A. Nielsen and I. L. Chuang. Quantum Computation and Quantum Information. *Camebridge University Press*, 2001.
- [11] Peter W. Shor. Polynomial time algorithms for prime factorization and discrete logarithms on a quantum computer. SIAM J. Sci. Statist. Comput., 26:1484, 1997.
- [12] P. W. Shor. Quantum computing. Doc. Math. J. DMV, Extra Volume ICM, 23:467–486, 1998.
- [13] J. S. Bell. Speakable and Unspeakable in Quatum Mechanics. Cambridge Univ. Press, page 212, 1987.
- [14] C. H. Bennett, D. P. DiVincenzo, J. A. Smolin, and W. K. Wootters. Mixed-state entanglement and quantum error correction. *Phys. Rev. A*, 54:3824–3851, Nov 1996.

- [15] Choi. M.D. Completely positive linear maps on complex matrices. Linear Algebra and its Applications, 10(3):285 – 290, 1975.
- [16] I. Bengtsson and K. Zyczkowski. Geometry of Quantum States. Cambridge University Press, New York, 2006.
- [17] D. P. DiVincenzo. Topics in Quantum Computers. eprint arXiv:condmat/9612126, December 1996.
- [18] B. Schumacher. Quantum coding. Phys. Rev. A, 51(4):2738–2747, Apr 1995.
- [19] M. J. Donald, M. Horodecki, and O. Rudolph. The uniqueness theorem for entanglement measures. *Journal of Mathematical Physics*, 43:4252– 4272, September 2002.
- [20] A. Lesniewski and M. B. Ruskai. Monotone Riemannian metrics and relative entropy on noncommutative probability spaces. *Journal of Mathematical Physics*, 40:5702–5724, November 1999.
- [21] P. Gibilisco, D. Imparato, and T. Isola. Uncertainty principle and quantum Fisher information. II. Journal of Mathematical Physics, 48(7):072109, July 2007.
- [22] A. Heslot. Quantum mechanics as a classical theory. Phys. Rev. D, 31(6):1341–1348, Mar 1985.
- [23] D. J. Rowe, A. Ryman, and G. Rosensteel. Many-body quantum mechanics as a symplectic dynamical system. *Phys. Rev. A*, 22(6):2362– 2373, Dec 1980.
- [24] V. Cantoni. Generalized 'transition probability'. Communications in Mathematical Physics, 44:125–128, 1975.
- [25] V. Cantoni. The Riemannian structure on the states of quantum-like systems. Communications in Mathematical Physics, 56:189–193, 1977.

- [26] V. Cantoni. Intrinsic geometry of the quantum-mechanical phase space, Hamiltonian systems and Correspondence Principle. *Rend. Accad. Naz. Lincei*, 62:628–636, 1977.
- [27] V. Cantoni. Geometric aspects of Quantum Systems. Rend. sem. Mat. Fis. Milano, 48:35–42, 1980.
- [28] V. Cantoni. Superposition of physical states: a metric viewpoint. Helv. Phys. Acta, 58:956–968, 1985.
- [29] R. Cirelli, P. Lanzavecchia, and A. Mania. Normal pure states of the von Neumann algebra of bounded operators as Kähler manifold. *Journal of Physics A Mathematical General*, 16:3829–3835, November 1983.
- [30] R. Cirelli and P. Lanzavecchia. Hamiltonian vector fields in quantum mechanics. *Nuovo Cimento B Serie*, 79:271–283, February 1984.
- [31] M. C. Abbati, R. Cirelli, P. Lanzavecchia, and A. Maniá. Pure states of general quantum-mechanical systems as Kähler bundles. *Nuovo Cimento B Serie*, 83:43–60, September 1984.
- [32] J. P. Provost and G. Vallee. Riemannian structure on manifolds of quantum states. *Communications in Mathematical Physics*, 76:289–301, September 1980.
- [33] A. Ashtekar and T. A. Schilling. Geometrical Formulation of Quantum Mechanics. In A. Harvey, editor, On Einstein's Path: Essays in honor of Engelbert Schucking, page 23, (1999).
- [34] G. Gibbons. Typical states and density matrices. Journal of Geometry and Physics, 8:147–162, March 1992.
- [35] D. C. Brody and L. P. Hughston. Geometric quantum mechanics. Journal of Geometry and Physics, 38:19–53, April 2001.
- [36] M. De Gosson. The principles of Newtonian and quantum mechanics. Imperial College Press, London, 2001.
- [37] J. F. Cariñena, J. Grabowski, and G. Marmo. Quantum Bi-Hamiltonian Systems. International Journal of Modern Physics A, 15:4797–4810, 2000.
- [38] G. Marmo, G. Morandi, A. Simoni, and F. Ventriglia. Alternative structures and bi-Hamiltonian systems. *Journal of Physics A Mathematical General*, 35:8393–8406, October 2002.
- [39] G. Marmo, A. Simoni, and F. Ventriglia. Geometrical Structures Emerging from Quantum Mechanics. in J. C. Gallardo, E. Martinez (Eds.), Groups, Geometry and Physics, Monografias de la Real Academia de Ciencias, Zaragoza, 2006.
- [40] J. Clemente-Gallardo and G. Marmo. The space of density states in geometrical quantum mechanics. In F. Cantrijn, M. Crampin and B. Langerock, editor, 'Differential Geometric Methods in Mechanics and Field Theory', Volume in Honour of Willy Sarlet, Gent Academia Press, pages 35–56, (2007).
- [41] J. F. Cariñena, J. Clemente-Gallardo, and G. Marmo. Geometrization of quantum mechanics. *Theoretical and Mathematical Physics*, 152:894– 903, July 2007.
- [42] J. Grabowski, M. Kuś, and G. Marmo. Symmetries, group actions, and entanglement. Open Sys. Information Dyn., 13:343–362, 2006.
- [43] J. Grabowski, M. Kuś, and G. Marmo. Geometry of quantum systems: density states and entanglement. *Journal of Physics A Mathematical General*, 38:10217–10244, December 2005.

- [44] J. Clemente-Gallardo and G. Marmo. Basics of Quantum Mechanics, Geometrization and Some Applications to Quantum Information. International Journal of Geometric Methods in Modern Physics, 5:989, 2008.
- [45] E. Ercolessi, G. Marmo, and G. Morandi. From the Equations of Motion to the Canonical Commutation Relations. to be published in "La Rivista del Nuovo Cimento", May 2010.
- [46] N. M. J. Woodhouse. Geometric quantization; 1st ed. Oxford Mathematical Monographs. Clarendon Press, Oxford, 1980.
- [47] M. G. A. Paris. Quantum estimation for quantum technology. Int. J. Quant. Inf., 7(125), 2009.
- [48] L. De Broglie. La mécanique ondulatoire et la structure atomique de la matiére et du rayonnement. J. Phys. Radium, 8(5):225-241, May 1927.
- [49] David Bohm. A suggested interpretation of the quantum theory in terms of "hidden" variables. i. *Phys. Rev.*, 85(2):166–179, Jan 1952.
- [50] D. Dürr, S. Goldstein, and N. Zanghì. Quantum equilibrium and the origin of absolute uncertainty. *Journal of Statistical Physics*, 67:843– 907, June 1992.
- [51] P. Chernoff and J. E. Marsden. Properties of Infinite Dimensional Hamiltonian Systems. Springer, New York, 1974.
- [52] R. Schmid. Infinite dimensional Hamiltonian Systems. Bibliopolis, Naples, 1987.
- [53] S. Lang. Differential and Riemannian Manifolds. Springer, New York, 1995.

- [54] G. Marmo, E. J. Saletan, A. Simoni, and B. Vitale. Dynamical Systems

   A Differential Geometric Approach to Symmetry and Reduction. John Wiley & Sons, Chichester, 1985.
- [55] P. Aniello, G. Marmo, and G. F. Volkert. Classical tensors from quantum states. Int. J. Geom. Meth. Mod. Phys., 06(3):369–383, 2009.
- [56] A. M. Perelomov. Coherent states for arbitrary Lie group. Communications in Mathematical Physics, 26:222, 1972.
- [57] F. T. Arecchi, E. Courtens, R. Gilmore, and H. Thomas. Atomic coherent states in quantum optics. *Phys. Rev. A*, 6(6):2211–2237, Dec 1972.
- [58] E. Onofri. A note on coherent state representations of Lie groups. Journal of Mathematical Physics, 16:1087–1089, May 1975.
- [59] A. Ferraro, S. Olivares, and M. G. A. Paris. Gaussian states in continuous variable quantum information. Napoli Series on physics and Astrophysics, Bibliopolis Naples, March 2005.
- [60] J. Williamson. On the Algebraic Problem Concerning the Normal Forms of Linear Dynamical Systems. American Journal of Mathematics, 58:23, January 1936.
- [61] D. Chruscinski and G. Marmo. Remarks on the GNS Representation and the Geometry of Quantum States. Open Syst. Info. Dyn., 16:157–177, 2009.
- [62] R. Cirelli, A. Manià, and L. Pizzocchero. Quantum Phase Space Formulation of Schrödinger Mechanics. International Journal of Modern Physics A, 6:2133–2146, 1991.
- [63] A. Wehrl. On the relation between classical and quantum-mechanical entropy. *Reports on Mathematical Physics*, 16:353–358, December 1979.

- [64] E. H. Lieb. Proof of an entropy conjecture of Wehrl. Communications in Mathematical Physics, 62:35–41, August 1978.
- [65] P. Facchi, G. Marmo, and F. Ventriglia. Private communication sessions, Zaragoza, January 2011.
- [66] E.C.G. Sudarshan G. Esposito, G. Marmo. From Classical to Quantum Mechanics. *Cambridge University Press, Cambridge*, 2004.
- [67] C. Radhakrishna Rao. Information and the accuracy attainable in the estimation of statistical parameters. Bull. Calcutta Math. Soc., 37:8191, 1945.
- [68] C. W. Helstrom. Quantum detection and estimation theory. Journal of Statistical Physics, 1:231–252, June 1969.
- [69] M. Kuś and K. Zyczkowski. Geometry of entangled states. Phys. Rev. A, 63(3):032307, Feb 2001.
- [70] M. M. Sinolecka, K. Zyczkowski, and M. Kuś. Manifolds of Equal Entanglement for Composite Quantum Systems. Acta Physica Polonica B, 33:2081, August 2002.
- [71] R. Hartshorne. Algebraic geometry. Graduate texts in mathematicals. Springer, Berlin, 1977.
- [72] E. Ercolessi, G. Marmo, G. Morandi, and N. Mukunda. Geometry of Mixed States and Degeneracy Structure of Geometric Phases for Multi-Level Quantum Systems. *International Journal of Modern Physics A*, 16:5007–5032, 2001.
- [73] I. Bengtsson. A Curious Geometrical Fact about Entanglement. In G. Adenier, A. Y. Khrennikov, P. Lahti, & V. I. Man'ko, editor, *Quan*tum Theory: Reconsideration of Foundations, volume 962 of American Institute of Physics Conference Series, pages 34–38, (2007).

- [74] V. I. Arnold. Les methodes mathematiques de la Mecanique Classique. Editions Mir, Moscow, 1976.
- [75] William K. Wootters. Entanglement of Formation of an Arbitrary State of Two Qubits. *Phys. Rev. Lett.*, 80(10):2245–2248, Mar 1998.
- [76] V. Coffman, J. Kundu, and W. K. Wootters. Distributed entanglement. Phys. Rev. A, 61(5):052306, Apr 2000.
- [77] G. Vidal. Entanglement monotones. Journal of Modern Optics, 47:355– 376, February 2000.
- [78] O. Gittsovich, O. Gühne, P. Hyllus, and J. Eisert. Unifying several separability conditions using the covariance matrix criterion. *Phys. Rev.* A, 78(5):052319, Nov 2008.
- [79] A. Ibort, V. I. Man'ko, G. Marmo, A. Simoni, and F. Ventriglia. An introduction to the tomographic picture of quantum mechanics. *Physica Scripta*, 79(6):065013, June 2009.
- [80] P. Horodecki. Measuring quantum entanglement without prior state reconstruction. *Phys. Rev. Lett.*, 90:167901, Apr 2003.
- [81] J. M. Sancho and S. F. Huelga. Measuring the entanglement of bipartite pure states. *Phys. Rev. A*, 61(4):042303-+, April 2000.
- [82] V. Bužek, R. Derka, G. Adam, and P. L. Knight. Reconstruction of Quantum States of Spin Systems: From Quantum Bayesian Inference to Quantum Tomography. Annals of Physics, 266:454–496, July 1998.
- [83] M. G. A. Paris and J. Řeháček, editors. Quantum State Estimation, volume 649 of Lecture Notes in Physics, Berlin Springer Verlag, 2004.
- [84] A. Acín, R. Tarrach, and G. Vidal. Optimal estimation of two-qubit pure-state entanglement. *Phys. Rev. A*, 61:062307, May 2000.

- [85] E. Bagan, M. A. Ballester, R. Muñoz Tapia, and O. Romero-Isart. Purity estimation with separable measurements. *Phys. Rev. Lett.*, 95:110504, Sep 2005.
- [86] O. V. Man'ko, V. I. Man'ko, and G. Marmo. Alternative commutation relations, star products and tomography. *Journal of Physics A Mathematical General*, 35:699–719, January 2002.
- [87] J. E. Marsden R. Abraham. Foundations of Mechanics. Benjamin-Cummings, New York, Reading, MA, 1978.
- [88] P. Aniello. Star products: a group-theoretical point of view. J. Phys. A: Math. Theor., 42:475210, November 2009.